Memorandum And Articles of Association Of VODAFONE IDEA LIMITED



Office of the Registrar of Companies RoC Bhavan, Opp Rupal Park Society Behind Ankur Bus Stop, Ahmedabad, Gujarat, India, 380013

Certificate of Incorporation pursuant to change of name

[Pursuant to rule 29 of the Companies (Incorporation) Rules, 2014]

Corporate Identification Number (CIN): L32100GJ1996PLC030976

I hereby certify that the name of the company has been changed from IDEA CELLULAR LIMITED to VODAFONE IDEA LIMITED with effect from the date of this certificate and that the company is limited by shares.

Company was originally incorporated with the name Idea Cellular Limited.

Given under my hand at Ahmedabad this Thirty first day of August two thousand eighteen.

DS MINISTRY
OF CORPORATE
AFFAIRS 05

L R MEENA

Registrar of Companies
RoC - Ahmedabad

Mailing Address as per record available in Registrar of Companies office:

VODAFONE IDEA LIMITED

SUMAN TOWER, PLOT NO.18,, SECTOR-11,, GANDHINAGAR, Gujarat, India, 382011



Co. No. 04 - 30976

Fresh certificate of Incorporation Consequent on

CHANGE OF NAME

The the OFFICE OF
THE REGISTRAR OF COMPANIES
GUJARAT,
DADRA AND NAGAR HAVELIT.
[Under the Companies Act. 1956 (1 of 1956)]

IN THE MATTER OF

BIRLA TATA AT & T LIMITED

I hereby certify that

BIRLA TATA AT & T LIMITED

which was originally incorporated on 14/03/1995 under the Companies Act. 1956 and under the name

BIRLA COMMUNICATIONS LIMITED

having duly passed the necessary resolution in terms of Section 21/31/44 of the Companies Act. 1956, on 29/04/2002 and the approval of the Central Government signifies in writing having been accorded thereto by the Registrar of Companies, Gujarat, vide his letter dated 01/05/2002 in terms of Government of India, Ministry of Law, Justice, & Company Affairs. (Department of Company Affairs) Notification No. GSR 507(E) dated 24-06-1985 the name of the said Company is this day changed to

IDEA CELLULAR LIMITED

and this certificate is issued pursuant to section 23(1) of the said Act.

Given under my hand at AHMEDABAD.

Dated this 01/05/2002

Dated this U1/U5/2002

(S. K. MANDAL)
REGISTRAR OF COMPANIES, GUJARAT

EGISTRAR OF COMPANIES, GUJARAT DADRA & NAGAR HAVELL. Co No.: 04 - 30976

Fresh Certificate of Incorporation consequent on

CHANGE OF NAME

IN THE OFFICE OF THE REGISTRAR OF COMPANIES GUJARAT, DADRA AND NAGAR HAVELI

[Under the Companies Act, 1956 (1 of 1956)]

IN THE MATTER OF

BIRLA AT & T COMMUNICATIONS LIMITED

I certify that

BIRLA AT & T COMMUNICATIONS LIMITED

which was originally incorporated on 14/03/1995 under the Companies Act, 1956 and under the name

BIRLA COMMUNICATIONS LIMITED

having duly passed the necessary resolution in terms of Section 21 of the Companies Act, 1956, on 24/09/2001 and the approval of the Central Government signifies in writing having been accorded thereto by the Registrar of Companies, Gujarat, vide his letter dated 6/11/2001 in terms of Government of India, Ministry of Law, Justice & Company Affairs. (Department of Company Affairs) Notification No. GSR 507(E) dated 24/06/1985, the name of the said company is this day changed to

BIRLA TATA AT & T LIMITED

and this certificate is issued pursuant to Section 23(1) of the said Act.

Given under my hand at Ahmedabad.

Dated this 6TH November, 2001

(S K MANDAL)
REGISTRAR OF COMPANIES
GUJARAT, DADRA & NAGAR HAVELI



CO. NO. 04- 30976

[कम्पनी अधिनियम, 1956 की घारा 18 (3)]
[Section 18(3) of The Companies Act, 1956]
एक राज्य से दूसरे राज्य में रिजस्ट्रोक्टल कार्याक्ष्य के अन्तरण की पुष्ठि
करनेवाके न्यायालय के आदेश की रिजस्ट्रीकरण का प्रमाणपत्र

CERTIFICATE OF REGISTRATION

OF

THE ORDER OF COMPANY LAW BOARD,
CONFIRMING TRANSFER OF THE REGISTERED OFFICE
FROM ONE STATE TO ANOTHER

(_ने विशेष संकर	प्रदारा
रजिस्ट्रीकृत कार्यालय का	राज्य से		राज्य में अन्त	करण करके स	धान की
बाबत संगम-ज्ञापन के उपबंधों में परिव	र्तन कर दिया है और ऐसे परिव	र्तन केा		_तारीख	
के आदेश द्वारा पुष्टि कर दी गई है TheBIRLA AT & '	r communications	LIMITE	0		-
		having	by Special	Resolution a	altered
the provisions of its Memora	ndum of Association w	ith respect	to the place	of the Regi	stered
Office by changing it from					
LAW BOARD WESTERNRE					FANT
vide C P No. 504/17/ मैं एतद्द्वारा प्रमाणित करता ह	CLB/WR/1996 कि उक्त आदेश की प्रमाणि	त प्रति इस वि	रन रजिस्ट्रीकृत व	र दी गई है।	
I hereby certify that a c	ertified copy of the sai	d Order ha	s this day b	een registere	d.
मेरे इस्ताक्षर से यह तारीख_		का दिया गया	1		
Given under my hand a	t AHMEDABAD this_	22nd	_day of OCT	OBER	199 6
One Thousand Nine Hundre	d and NINETY		_•		



(S.K. SAHA)
कम्पनियों का रजिस्ट्रार
Registrar of Companies,
GUJARAT,
DADRA & NAGAR HAVELI

**** ****

FRESH CERTIFICATE OF INCORPORATION CONSEQUENT ON CHANGE OF NAME

IN THE OFFICE OF THE REGISTRAR OF COMPANIES, MAHARASHTRA. **BOXIBAXX** MUMBAI.

In the matter of BIRLA COMMUNICATIONS

I hereby approve and signify in writing under Section 21 of the Companies Act, 1956 (Act of 1956) read with the Government of India, Department of Company Affairs, Notification No. G. S. R. 507E dated the 24th June 1985 the change of name of the Company:

from BIRLA COMMUNICATIONS LIMITED

BIRLA T&TA COMMUNICATIONS LIMITED to

and I hereby certify that BIRLA COMMUNICATIONS LIMITED which was originally incorporated on FOURTEENTH

under the Companies Act, 1956 and under the name day of MARCH. 1995 BIRLA COMMUNICATIONS LIMITED having

duly passed the necessary resolution in terms of section 21 /226 kk (a) (2) (b) of the Companies Act, 1956 the name of the said Company is this day changed to BIRLA AT&T COMMUNICATIONS and this

certificate is issued pursuant to Section 23(1) of the said Act.

MUMBA I

Given under my hand at BKMBAK this THIRTIETH

200

day of

one thousand nine hundred ninety _SIX

(R. VASUDEVAN) & Registrar of Companies. Maharashtra, Bombayx

MUMBAI.



प्राह्मप॰ साई॰ सार॰ Form I. R.

निगमन का प्रमाण-पत्र

CERTIFICATE OF INCORPORATION

No. 11-86465 of 19 .95.	
	त करता हूं कि आंख····· 6 (1956 का 1) के अधीन निगमित भी गई है और यह
***************************************	· · · · · · · · · · · · · · · · · · ·
I hereby certify that .BIRLA COMMU	NICATIONS LIMITED
is this day incorporated under the Companies and that the Company is limited.	
मेरे हरताबार से आज ता॰	भे दिया गया। Ythis FOURTEENTH
day of 35 Marsh One thousand nine h	nundred and NINETYFIVE
(S. I	२.V.V.SA TYANARAYAN A कग्पनियों का रजिस्ट्रार
Addl.	Registrar of Companies

No.11-86465



कारबार प्रारम्भ करने के लिए प्रमाण-पत्र Certificate for Commencement of Business

कम्पनी अधिनियम, 1956 की धारा 149(3) के अनुसरण Pursuant of Section 149(3) of the Companies Act, 1956

मैं एतद्झरा प्रमाणि त करता हूं कि · · · · · · · · · · · · · · · · · ·
•••••
•••••
जो कम्पनी अधिनियम, 1956 के अधीन तारीख ' ' ' ' ' ' ' को नियमित की गई
थी और जिसने आज विहित प्ररूप में सम्यक रूप से सत्यापित घोषणा फाइल कर दी है कि
उक्त अधिनियम की धारा 149(1) (क) से लेकर (घ)तक/149(2) (क) से लेकर (ग)
तक की शर्तों का अनुपालन किया गया है, कारबार प्रारम्भ करने की हकदार है।
LIMITED.
which was incorporated under the Companies Act, 1956, on the FOURTEENTHday of
मेरे हस्ताक्षर सेः यह तारीख ' ' ' को ' ' को '
में दिया गया ।
Given under my hand at . BOMBAY.
this. ELEVENTHday of AUGUST one thousand nine hundred
NINETYFIVE THE HEADY (Y.M.DEOLIKAR) THE HEADY THE WEST OF THE
कम्पनियों का रजिस्टार
Registrar of Companies
॰ एस॰ सी॰-10 S.C-10
3. 1. 2

प्रभा । नुटेक-230 सिषिल/ 85-86-भासमुटेक-(सी-//)-14-7-88-5,000. MGIPTC-230 Civil/8-86-GIPTC-(C-71)--[4-7-89-5,000.

UNDER THE COMPANIES ACT, 1956

PUBLIC COMPANY LIMITED BY SHARES

MEMORANDUM OF ASSOCIATION

OF

VODAFONE IDEA LIMITED*

- I. The name of the Company is VODAFONE IDEA LIMITED*.
- II. The registered office of the Company will be situated in the state of Gujarat.
- III. The objects for which the Company is established are:
- (A) THE MAIN OBJECTS OF THE COMPANY TO BE PURSUED BY THE COMPANY ON ITS INCORPORATION ARE:
- 1. To provide all or any of the following services namely: basic telephone services, cellular telephone services, unified access services (basic and cellular services), international long distance calling services, national long distance calling services, public mobile radio trunked services (PMRTS), global mobile personal communications services (GMPCS), V-SAT, electronic mail services, video text, voice mail services, data communication services, paging services, private switching network services, transmission network of all types, computer networks i.e. local area network, wide area network, multimedia services, intelligent network and other value added services and do all activities for providing such services like excavation, construction, infrastructure fabrication, installation, commissioning and testing of equipment, marketing and selling.
- 2. To carry on the business of manufacture, assemble, buy, sell, import, export, service, repair or otherwise deal in all types of electronics equipment viz, electronic communication, teletext, televideo, microwave and facsimile equipment, telecommunication and telematics equipment, network switching equipment, network communication equipment, all sorts of electrical and electronic wireless sets, high frequency apparatus, radar equipment, sonars, osciloscopes of all kinds and description, electronic and electrical products, industrial electronics, software procedures, peripheral products, modules, instruments, hardware and software system, all kinds of solid state devices, control system and allied equipment, aerospace and defence electronics, entertainment electronics, household electronics and such other electronic equipment gadget items which may be developed and introduced in India and elsewhere.
- 3. To carry on the business of manufacture, improve, assemble, prepare, design, develop and install equipment, fabrication repair, anything and everything in electronics, telephone networks, cellular mobile networks systems, paging systems, electronic mail, voice mail, data communications, electric gadgets and appliances, measuring and testing instruments, components, accessories and spares for control engineering, communication, defence and computer data processing application that may be developed by invention, experiment and research.

^{*} Name changed from 'Idea Cellular Limited' to "Vodafone Idea Limited" at the Extra-ordinary General Meeting held on June 26, 2018 and confirmed by the Registrar of Companies, Ahmedabad (Gujarat) vide Certificate of Incorporation dated 31st August, 2018.

3.(a) To carry on the business of internet service and broadband service provider and to provide, render or make available and operate, sell, export, import, trade, maintain, improve, repair, service, research, develop all kinds of internet services including internet telephony, voice over internet protocol, fax over internet protocol, voice messaging applications and services in respect of and relating to bandwidth, hosting of websites, information technology and telecommunications or wireless communications through internet or any other electronic media and deal or trade in accessories, assemblies, apparatus, spares, hardware and software for internet services.

(B) THE OBJECTS INCIDENTAL OR ANCILLIARY TO THE ATTAINMENT OF THE MAIN OBJECTS:-

- 4. To purchase, take in exchange or on lease, rent, hire, occupy, allow to be occupied otherwise and use any freehold, lease-hold or other immovable property and any lands, estates, shops, warehouses, show-rooms, offices, workshops, stock-in-trade, waterways, easements or other rights or interests in any land, buildings and premises or any other immovable or movable, real or personal property or right which the Company may think necessary or convenient for the purposes of its business, and as to any real property, either in consideration of a gross sum or of a rent charge in cash, services or kind or on perpetual lease rent in the manner aforesaid or partly in one way and partly in another or others.
- 5. On any land or waterways purchased, leased or otherwise acquired to erect, build, construct, improve, maintain, develop, alter, enlarge, pull down, replace, work, manage, any buildings, works, houses, shops, workshops, offices, waterhouses, showrooms, refreshment rooms, lavatories, and other conveniences, cottages and other buildings with light and power generating and other fixtures and fittings and apparatus for working and for the comfort and accommodation of working people, and roads, ways, branches or sidings and other works and conveniences which may seem calculated directly or indirectly to advance the Company's interest, and to contribute, to subsidize or otherwise assist or take part in the construction, improvement, maintenance, working management, carrying out or control thereof.
- 6. To borrow, raise and secure the payment and repayment of money for any of the purposes of the Company's business or otherwise, in such manner as the Company shall think fit, and in particular, by the issue of redeemable preference shares, mortgage debentures or debenture stock, perpetual or otherwise and issuable or payable at par or at a premium or discount and repayable by periodical drawings or otherwise, to bearer or otherwise, charged upon all or any of the Company's undertaking and/or property (both present and future or movable or immovable) or by other obligations or securities of the Company or by mortgage or charge on all or any part of the property of the Company, present and

future, movable or immovable including its uncalled capital or without any charge, and to purchase, redeem or pay off, cancel and discharge any such securities, subject however to the provisions of Section 58 A of the Companies Act, 1956 and the directives of the Reserve Bank of India.

- 7. To draw, make, accept, endorse, discount, execute, retire, discharge, negotiate, issue and honor bills of exchange, cheques, promissory notes, bills of lading, dock and warehouse warrants, rail receipts and other negotiable, semi-negotiable or transferable instruments or securities.
- 8. To open current, overdraft, loan, cash, credit or deposit account or accounts with any bank, company, firm or person.
- 9. To receive money on deposit from and to lend moneys to any person, firm, association, society, company or corporation at interest or otherwise and on such terms and on such security as may seem expedient or without any security and in particular, to members or customers and others having or likely to have dealings with the Company, provided that the Company shall not carry on any banking business as defined by the Banking Regulation Act, 1949, subject however to the provisions of Section 58 A of the Companies Act, 1956 and the directives of the Reserve Bank of India.
- 10. To incur debts and obligations for the conduct of any business of the Company, and to purchase or hire goods, material or machinery on credit or otherwise for any business or purpose of the Company.
- 11. To establish, maintain and conduct or discontinue or close agencies and branches and appoint representatives in any part of the world for the conduct of the business of the Company or for the purchase, sale or exchange either for ready delivery or future.
- 12. To employ or otherwise engage technical experts, engineers, mechanics, foremen and skilled and unskilled labor for any of the purposes of the business of the Company.
- 13. To develop, apply for, purchase or otherwise acquire and promote, protect prolong and renew whether in India or elsewhere, any technology patents, patent rights, brevets d'invention, licenses, concessions, trade marks, designs and the like, conferring any exclusive or non-exclusive or limited right of use or any secret of other information as to any invention, process or privileges which may seem capable of being used for any of the purposes of the Company, and to use, exercise, develop, manufacture under, or grant licenses or privileges in respect of the above or otherwise to turn to account the property, rights or information, use or license so acquired, and to subsidize, take part in or assist in any experiments, investigation and/or research likely to prove beneficial to the Company.
- 14. To adopt such means of making known the activities of the Company, as may seem expedient, and in particular, by advertising in the press, by circulars, by purchase and exhibitions of works of art or interest and by registering and establishing and protecting

- trade marks, publication of books and periodicals and by granting prizes, rewards and donations.
- 15. To lend out, deposit, invest and deal with the moneys of the company not immediately required in such manner and upon such terms as may from time to time be determined by the Directors.
- 16. To guarantee the performance of contracts and obligations or debts of any other company, corporations, firm or person, including (without prejudice to the said generally) bank over drafts, bills of exchange and promissory notes and generally to give securities, guarantees, warranties and indemnities, with or without consideration.
- 17. To acquire and undertake the whole sale or any party of the goodwill, business, concern, undertaking, property, rights, assets and liabilities of any person, firm, association, society, company or corporation carrying on any business which the Company is authorized to carry on or possessed of property suitable for the purpose of this Company and to pay for the same by shares or debentures of this Company, or by cash or otherwise, or partly in one way and partly in another or others, and to conduct, expand and develop or wind-up and liquidate such business and to purchase and take steps for the acquisition of existing and new licenses in connection with any such business.
- 18. To form, establish, promote, subsidize, aid, acquire, organize or be interested in any other company or companies, including subsidiaries, or partnerships for the purpose of acquiring all or any of the undertaking, property and liabilities of this Company or of any share therein by way of exchange of its shares or otherwise.
- 19. To enter into partnerships or into any arrangements for sharing profits, union of interests, co-operation, joint ventures, reciprocal concessions, licenses or otherwise, with any person, firm, association, society, company or corporation carrying on or engaged in or about to carry on or engage in any business or transaction which this Company is authorized to carry on, and to give any person, firm or company, special rights, licenses and privileges in connection with the above.
- 20. To take, purchase, subscribe for or acquire by exchange or otherwise and to hold or deal in any shares (whether fully or partly paid), stock, debenture stock, obligations or other securities in or of any other company, or which are issued by any authority, whether sovereign, governmental, corporate, municipal, local or otherwise in India or elsewhere, and to cause the same or any of them to be vested in or held by a nominee or nominees for and on behalf of the Company and upon a distribution of assets or division of profits to distribute any such shares, stock, obligations or other securities amongst the members of the Company in specie.
- 21. To improve, manage, develop, mortgage, charge, sell, transfer, exchange, lease, underlease, surrender or otherwise deal with; dispose of or turn to account, all or any part of the business, immovable or movable property, rights, and effects for the time being of the Company in such manner, on such terms and for such purposes as the Company may

think fit and as to any sale of real property either in consideration of a gross sum or of rent or others and to sell, transfer or dispose of the whole of the undertaking of the Company or any part thereof, for cash or such other consideration as the Company may think fit, and in particular for shares, debentures or securities of any other company having objects altogether or in part similar to those of this Company.

- 22. To amalgamate with any other company having objects altogether or in part similar to those of this Company or otherwise.
- 23. To continue, establish and support or aid in the establishment or support of co-operative societies, associations and other institutions, funds, trusts, amenities and conveniences and at its discretion to grant bonuses, pensions, allowances and to make payment towards insurance and to subscribe or guarantee money for charitable or benevolent objects, also to remunerate or make donations by cash or other assets or by the allotment of shares credited as fully or partly paid up or in any other manner (so far as by law allowed) to any party for services rendered or to be rendered in placing or assisting to place any shares in the Company's capital or any debentures, debenture stock or other securities of the Company or in or about the formation or promotion of the Company or the conduct of any of its business.
- 24. To donate, contribute, subscribe, promote, support or aid or otherwise assist or guarantee money to charitable, benevolent, religious, scientific, national, public or other institutions, funds or objects or for any exhibition or for any public, general or other objects.
- 25. To create any depreciation fund, reserve fund, sinking fund, insurance fund or any other special fund, whether for depreciation or for repairing, improving, extending or maintaining any of the property of the Company or any other purpose conducive to the interest of the Company.
- 26. To place, to reserve or to distribute bonus shares in respect of forfeited shares and moneys arising from the sale by the Company of forfeited shares in conformity with the provisions of law.
- 27. Subject to the provisions of the Companies Act, 1956, to distribute among the members in specie or kind, any property of the Company or any proceeds of sale or disposal of any property of the company, or to gift in favor of any person, firm, body corporate or institution, so that no distribution amounting to reduction of capital or of dividend be made except in conformity with the requirements of law for the time being in force.
- 28. To refer any claims, demands, disputes or any other question by or against the Company or in which the Company is interested or concerned and whether between the Company and the member or members or his or their representatives or between the Company and third party to arbitration in India or at any place outside India and to observe, perform and to do all acts, deeds, matters and things to carry out or enforce the awards.

- 29. To apply all or any expenses incurred in connection with the formation, promotion or incorporation of this Company or any other company or of and incidental to the winding up of any company the whole or part of the property whereof is acquired by this Company.
- 30. To procure the recognition of the Company in any country, state or place and to establish and regulate agencies for the purpose of the Company's business and to apply or join in applying to any parliament, local government, municipal or other authority or body for any rights or privileges that may seem conducive to the Company's objects or any of them and to oppose any proceedings or applications which may seem calculated directly or indirectly to prejudice the Company's interests.
- 31. To do all or any of the above things in all or any of the States of India and/or in any part of the world and either as principals, agents, contractors, trustees or otherwise and by or through trustees, attorneys, agents or otherwise and either alone or in conjunction with others, and to do all such things as are incidental or conducive to the attainment of the above objects or any of them.
- 32. To provide for the welfare of the Directors or ex-Directors or the employees or exemployees of the Company and the wives, widows and families, of such persons by building or by contributing to the building of houses, dwelling houses, chawls or by grants of money, pensions, allowances, bonus or other payments or by creating and from time to time subscribing to provident and other funds and by providing or subscribing towards schools, places of instruction and recreation, hospitals, dispensaries, medical and other attendance and other assistance as the Company shall think fit, and to form, subscribe to or otherwise aid benevolent, religious, scientific, national, public, political or other institutions or objects or purposes.
- 33. To apply for, purchase or otherwise acquire and protect, prolong and renew whether in India or elsewhere any patents, brevets d'invention, licenses, concessions, and the like conferring any exclusive or non-exclusive or limited right to use of any secret or other information as to any invention which may seem capable of being used for any of the purposes of the Company.
- 34. To enter into arrangements with any government or authority, municipal, local or otherwise, that may seem conducive to the Company's objects or any of them; and to obtain from any such government or authority any rights, privileges and concessions which the Company may think fit desirable to obtain; and to carry out, exercise, and comply with any such arrangements, rights, privileges and concessions.
- 35. To open accounts with any banks or financial institutions and to draw, make, accept, endorse, negotiate, discount, execute and issue promissory notes, bills of exchange, hundies, bills of lading, warrants, debentures and other negotiable or transferable instruments and to buy, sell and deal in the same.

- 36. To sell or otherwise dispose of the whole or any part of the business and/or undertaking of the Company, either together or in portions for such consideration and on such terms as may be considered expedient.
- 37. To purchase or otherwise acquire and undertake, and to supervise and manage, all or any part of the business, property, assets and liabilities of any person or company.
- 37a. To invest and deal with the moneys of the Company not immediately required for the purpose of its business in or on such investments or securities and in such manner as may be considered expedient and to dispose of or vary any such investments or securities.
- 38. To enter into any partnership or into any arrangement for sharing profits or to amalgamate with any person or company carrying on or proposing to carry on any business.
- 39. To lend or advance money or give credit to such persons or companies and on such terms as may be considered expedient, and to receive money on deposit or loan from any person or company.
- 40. To borrow and raise money and to obtain and utilize banking facilities of any nature on any terms and for any purposes whatsoever, including but not limited to facilities for the issue by any bank or financial institution of bonds, guarantees, indemnities, documentary and other credits to any person in respect of the obligations or purported obligations of the Company, and to give counter-indemnities on any terms to any banks or financial institutions issuing such bonds, guarantees, indemnities, documentary or other credits and to secure any debt or liability by mortgages of or charges upon all or any part of the undertaking, real and personal property, assets, rights and revenues (present or future) and uncalled capital of the Company and by the creation and issue on any terms of debentures, debenture stock or other securities of any description.
- 41. To enter into any guarantee, bond, indemnity or counter indemnity and otherwise give security or become responsible for the performance of any obligations or the discharge of any liabilities of or by any person or company in any manner on any terms and for any purpose any consideration or advantage and whether or not in furtherance of the attainment of any other objects of any Company and in particular a subsidiary of holding company of the company (without derogation from the generality of the foregoing) to guarantee, support or secure, by personal covenant or by mortgaging or charging all or any part of the undertaking, real and personal property, assets and revenues (present and future) and uncalled capital of the Company, or by both such methods, or in any other manner whatsoever, the payment or repayment of any moneys secured by, or payable under or in respect of, any debts, obligations or securities whatsoever and the discharge of any liabilities whatsoever, including but not limited to those of any company which is for the time being a "subsidiary or holding company of the company" or a subsidiary of any such holding company or is otherwise associated with the Company in business.
- 42. To pay for any property, assets or rights acquired by the company, and to discharge or satisfy any debt, obligation or liability of the Company, either in cash or in shares with or

without preferred or deferred rights in respect of dividend or repayment of capital or otherwise, or by any other securities which the Company has power to issue, or partly in one way and partly in another, and generally on such terms as may be considered expedient.

- 43. To accept payment for any property, assets or rights disposed of or dealt with or for any services rendered by the Company, or in discharge or satisfaction of any debt, obligation or liability of the Company, either in cash or in shares, with or without deferred or preferred rights in respect of dividend or repayment of capital or otherwise, or in any other securities, or partly in one way and partly in another, and generally on such terms as may be considered expedient.
- 44. To form, promote, finance or assist any other company, whether for the purpose of acquiring all or any of the undertaking, property and assets of the company or for any other purpose, which may be considered expedient.
- 45. To apply for, purchase or otherwise acquire and hold, use, develop, sell, license or otherwise dispose of or deal with patents, copyrights, designs, trade marks, secret processes, know-how and inventions and any interest therein.
- 46. To establish and maintain or procure the establishment and maintenance of any pension, superannuation funds or retirement benefit schemes (whether contributory or otherwise) for the benefit of, and to give or arrange to be given donations, gratuities, pension allowances, emoluments and any other relevant benefits to any persons who are or were at any time in the employment or services of the Company, or of any company which is a subsidiary or holding company of the Company or which is a subsidiary of any such holding company or is allied to or associated with the Company, or any such other company as aforesaid, or who may be or have been directors or officers of the company, or of any such other company as aforesaid, and the wives, widows, families and dependents of any such person, and to establish, subsidize and subscribe to any institutions, associations, societies, clubs, trusts or funds calculated to be for the benefit of, or to advance the interests and well being of the Company or of any other company as aforesaid, and to subscribe or guarantee money for charitable or benevolent objects, for any exhibition or for any public, general or useful object, and to do any of the matters aforesaid either alone or in conjunction with any such other company as aforesaid and without prejudice to the generality of the foregoing to act either alone or jointly as trustee or administrator for the furtherance of any of the aforesaid purposes.
- 47. To establish, and subject to such terms as may be considered expedient, a scheme or schemes for or in relation to the purchase of, or subscription for, any fully or partly paid shares in the capital of the Company by, or by trustees for, or otherwise for the benefit of, employees of the Company or of its subsidiary or associated Companies.
- 48. To the extent permitted by law, to give financial assistance for the purpose of the acquisition of shares of the Company or for the purpose of reducing or discharging a liability incurred for the purpose of such an acquisition and to give such assistance by means of a gift, loan, guarantee, indemnity, the provision of security or otherwise.

- 49. To subscribe or guarantee money for any national, charitable benevolent, public, general, political or useful subject, and to undertake and execute any trusts the undertaking whereof may be considered expedient, and either gratuitously or otherwise.
- 50. To enter into any arrangement with the Government of India, the Government of State or Local Authority, Country, Dominion or with any authorities local or otherwise or the purpose of carrying out the object of the Company of furthering its interest and to obtain from such Government or authority or person any charters, subsidies, loans, indemnities, grants, contracts, rights, powers, concessions, privileges or immunities which the Company may think desirable to obtain and to carry out, exercise and comply with any such arrangements, rights, privileges and concessions.
- 51. To remunerate any person or company rendering services to the Company in any manner and to pay all costs, charges and expenses incurred or sustained in or about the promotion and establishment of the Company and of any other company formed, promoted, financed or assisted by the Company, or which the Company shall consider to be in the nature of preliminary expense relating to the company or any such other company, including the cost of advertising, commissions for underwriting brokerage, printing and stationery, and the legal and other expenses of the promoters.
- 52. To procure capital, machinery, equipment, property, technical know how, patents, licenses or any other facilities from any other company, person or association having objects altogether different or similar or partly similar to those of the Company.
- 53. To amalgamate with any other company body corporate or any other entity or to amalgamate any other company body corporate or any other entity whether by sale or purchase (for fully or partly paid up shares or otherwise) of the undertaking, subject to the liabilities of this or any such other company as aforesaid, with or without winding up or by sale or purchase (for fully or partly paid up shares or otherwise) of all or a controlling interest in the shares or stock of this or any such other company as aforesaid, or by partnership, or in any other manner.
- 54. Subject to the provisions of the Companies Act, 1956, to place, to reserve or to distribute as bonus shares amongst the members of otherwise apply, as the Company may from time to time think fit, any moneys belonging to the Company including moneys received by way of premium on shares or debentures issued by the Company and money received in respect forfeited shares.
- 55. To borrow or raise money on any terms without security or on security of land, buildings, factories, machinery, tools, bills of exchange, promissory notes, bonds, bills, bills of lading, warrants, stocks, shares, debentures, book debts or undertaking of the Company and properties of every description or any one or more of them from banks, financial institutions, companies and any other person or association of persons, subject to the provisions of Section 58A and directives of Reserve Bank of India. Provided the Company shall not carry on banking business as defined in the Banking Regulation Act, 1949.

- 56. To lend money to such persons or companies and on such terms as may seem expedient, and in particular to customers and others dealing with the Company and to guarantee the performance of contracts by such person or Companies.
- 57. To give guarantee for the performance or discharge of any obligations, liabilities or duties of or the payment of money by any persons, firms or companies or Government or State and to give indemnities of all kinds.
- 58. To assist in the event of the Company occupying the position of group company, within the meaning of Section 372 of Companies Act, 1956 or any statutory modifications thereof, any company in the group in any manner and to any extent including the giving of loans and giving the guarantee or providing of securities of any kind whatsoever in connection with any loan given to the latter by any person, firm or body corporate.
- 59. To invest and deal with the moneys of the Company not immediately required for the purpose of its business in or on such investments or securities and in such manner as may be considered expedient and to dispose of or vary any such investments or securities.
- 60. To create, execute, grant or issue any mortgages, debentures, stock or bonds either at par, premium or discount and either redeemable or irredeemable, secured upon all or any part of the undertaking, rights and properties of the Company, present and future, including uncalled capital or the unpaid calls of this Company.
- 61. To promote and avail in promoting, constitute, form, organize, or partnerships for the purpose of acquiring and undertaking any property and liabilities of the Company; any having similar objects.
- 62. To enter into partnership or any other arrangements for sharing profits or losses, cooperation, joint ventures, reciprocal concession, license or otherwise with any person, firm, association, society or body corporate and to give any special rights, licenses and privileges and in particulars, the right to nominate one or more person or persons, whether they be shareholders or not, to be directors of the Company.
- 63. To act as contractors, suppliers, agents, importers and exporters for any Government or autonomous body or any firm, company, organization in the private or public sector in furtherance of any of the objects of the Company.
- 64. To establish provident funds for the benefit of the Company's employees upon the basis of joint contribution by the employees and the Company respectively or otherwise and grant pensions, allowances, gratuities and bonuses to such employees or relations of such persons.
- 65. To purchase or take on lease or in exchange or otherwise acquire any movable properties, flats and any rights or privileges which the Company may think necessary or expedient, for the purpose of the business and, in particular, any land, buildings, works and easements and on such terms as may be deemed useful.

- 66. To sell or dispose of the undertaking, property and rights of the Company or any part thereof for consideration and on such terms and conditions as the Company may think fit and in particular for shares, debentures or securities of any other company having objects altogether or in part similar to those of this Company.
- 67. To sell or subject any concessions or licenses obtained or contracts entered into any generally to sell the whole or any part of the property and business of the Company for cash or for shares or obligations of any person or persons.
- 68. To improve, manage, develop, exchange, let on lease, mortgage, sell dispose of turn to account, grant rights and privileges in respect of or otherwise deal with all or part of the properties and rights of the Company.
- 69. To establish and maintain local registers, agencies and branches, places of business, and carry on business in any part or parts of the world.
- 70. To adopt such means of making known, publishing and advertising any of the business, and services of the Company as may seem expedient and, without prejudice to the said generally and in particular, by advertising in the press, radio, television, by circulars, by purchase or acquisition or otherwise by an exhibition of art of interest, by publication of books or other literature and by granting prizes, rewards and donations.
- 71. To pay for any rights or property acquired by the Company and to remunerate any person or Company whether by cash payment or by the allotment of shares, debentures or other securities of the Company credited as paid upon in full or in part.

(C) OTHER OBJECTS:

72. To plan, locate, design, establish, build, construct, equip, operate, make use of, administer, manage and maintain, service, improve, inspect, enlarge, alter, protect, develop, extend, repair, replace, refurbish, pull down and remove, and carry out work in respect of, the whole or any part or parts of any electricity generating station, plant and or systems (including without limitation combined heat and power stations), generating sets, substation, transformer station, pumping station, building, plant, equipment, electric main works and any facilities ancillary to the operation or use of the aforesaid or any of them including structures, erections, pipes, pipelines, machinery, engines, shops and showrooms, offices, factories, works, warehouses, plants, platforms, derricks, transmission towers or pylons, rigs, wind structures, dams and associated structures, testing sites, offshore wave structures, installations (including without limitation solar power and geothermal installations), depots, distribution stations and sub-stations, pumping stations, compressor stations, laboratories, research stations, wharves, jetties, terminals, transport facilities, canals, roads, railways, branches or sidings, bridges, reservoirs, water courses, tunnels, airports, and structures of all kinds, whether for the purposes of the Company to, or in return for any consideration from any person, and to purchase or otherwise acquire, lease, charter and take or let on hire any of the same, and to contribute to, or assist in, or carry out any part of, any operation in respect of the same and to acquire, operate and maintain the licenses, consents, authorizations, wayleaves, easements and other rights

- capable or possibly capable of facilitating the aforesaid and all other rights that may or expertise any of the above.
- 73. To carry on the business of manufacturing, a running, operating, managing, advising on and supplying data processing and information retrieval systems. (whether or not remotely located and including but not limits to videotex, teletax and teletext systems) and systems utilizing the capture, storage, processing, transmission or receipt of messages and signals (including but not limited to data sounds and visual images) by, with the aid of, in conjunction with, or in way utilizing, computers, or similar equipment, and computer programs and databases and to carry on the business of operating, managing, advising on, supplying and dealing in services and facilities of all kinds which incorporate use or are used in conjunction with, in connection with or ancillary to, systems of such description as aforesaid or any of the apparatus and equipment comprised therein.
- 74. To provide remotely located office services and systems (including without prejudice to the generality of the foregoing telephone answering, calling and related services and computer bureau) and remotely located services and systems for the control of machinery utilizing telecommunication or data processing facilities to act as business and office managers, secretaries, messengers, telephone operators, commercial agents, mail order bureau, market researchers and to provide services in connection with the reception, processing and forwarding of signals and information by telephone telemessages, telegram, telex, letter, wireless telegraphy and (without limitation) another means of communication and the processing, ordering and payment for and dispatch and delivery of goods, articles and services of all kinds by any means whatsoever.
- 75. To carry on the trade or business of manufacturers, assemblers of and dealers in, contractors for, repairers and maintainers of, and importers and exporters of, all kinds of radio products, radio apparatus, including amplifiers and amplifying and public address equipment, electronics of all kinds and description, electronic devices, gadgets, modules, machinery and apparatus, medical electronic instruments and appliances and domestic electric and electronic appliances and component parts, tools, fittings and accessories connected with each of the aforesaid business.
- 76. To carry on the business of manufacturers, importers and exporters of and dealers in wires, conductors, copper aluminum fiber optic or other cables and wires (insulated or otherwise), pipes, flexible cords, rubber, polyvinychloride paper or any other insulation and/or covering materials of all kinds, lamps, values, transistors and other components apparatus and equipment and all kinds of electric, magnetic, galvanic/electric and electronic and other apparatus equipment and parts and electric, magnetic, electronic goods and articles of all kind and description.
- 77. To carry on the business of real estate and hold, deal with, manage, direct the management of, buy, sell, exchange, mortgage, charge, lease, dispose of, or grant any right or interest in over or upon any real or personal property of any kind whatsoever, including contingent and reversionary interest in any property and to undertake and carry on any business undertaking or transaction.

- 78. To carry on the business of land developers and acquire by purchase, amalgamation, grant concession, lease, license, barter or otherwise enter absolutely or conditionally, and either solely or jointly with any other houses, lands, farms rights and privileges, rights and hereditaments and any tract or tracts of country in India or elsewhere together with such rights, concessions, grants, powers and privileges as may be agreed upon, and granted by Government or the Rulers or owners, thereof, and to expend such sums of money as may be deemed requisite and advisable in the exploration, survey, farming and development thereof; and to acquire or to obtain rights over, be interested in, build, alter, construct, maintain, carry out, improve work, control, manage and regulate steam boats, aeroplanes, telephones, telegraphs, roads, tunnels, irrigation works, canals, waterways, rivers, wharves, docks, harbor works and harbors, factories, warehouses and other works and conveniences which the company may think conducive to any of its objects either by acquiring such properties outright or by acquiring the rights of others in to and over them, and generally to acquire in India or elsewhere by purchase, lease or otherwise for the company any real or personal, immovable or movable rights, easements, privileges, licenses, concessions, patents, patent rights, trade marks, machinery, rolling stock, plant, utensils, accessories and stock-in-trade whatsoever and to contribute to and take part in the constructing, carrying on, improving, working, controlling and managing any of such works or conveniences as aforesaid.
- 79. To carry on business as manufactures and dealers of and in Means & Media of communication such as intercom telephones, auto telephones, or such other intercom devices, telephone answering machines, journalism, radio, tape recorder, juke boxes or such appliances whereby sound or vision is communicated recorded, amplified, transmitted with or without remote control and with or without wires. Importers and exporters of and dealers in, dry and inert cells, batteries, flash lights of kinds and descriptions and electronic, electrical, thermal, sonic, ultrasonic, scientific, engineering, optical equipment, instruments and spares, parts and accessories thereof.
- 80. To pay for any technical know-how, rights or property acquired by the Company, and to remunerate any person or company whether by cash payment or by the allotment or shares, debentures or other securities of the Company as paid up in full or in part or otherwise.
- 81. To communicate with Chambers of Commerce and other mercantile and public bodies in India and elsewhere and concern and promote measures for the protection and advancement of trade, industry, commerce and other facilities.
- 82. To acquire by purchase, lease exchange or otherwise, and make advances on the security and deal in land, buildings of any tenure or description and any estate or interest therein and any rights over or connected with land to turn the same to account as may seem expedient and in particular by laying out and preparing land for building purpose and preparing building sites, by planting, paving, draining and cultivating land by constructing, reconstructing, pulling down, altering, improving, decorating, furnishing and maintaining offices, flats, service flats, houses, bungalows, chawls, factories, warehouses, shops, buildings, works and conveniences of all kinds and by consolidating or connecting or sub-dividing properties and by leasing, letting on building lease, building

- agreements selling (by installment or otherwise) and otherwise disposing of the same and by advancing moneys to and entering into contracts and arrangements of all kinds with builders, tenants, and others.
- 83. To plan, locate, design, establish, build, construct, equip, operate, make, lay, place, use, administer, manage and maintain, service, improve, inspect, enlarge, alter, protect, develop, extend, repair, replace, refurbish, pull down and remove and to carry on works in respect of, telephone wires (including those overhead and underground), cables, lines, plant and equipment facilities ancillary to the operation or use of any telephone service system or distribution system, and to acquire, operate and maintain the licenses, consents, authorizations, wayleaves, casements and other rights capable or possibly capable of facilitating the aforesaid.
- 84. To plan, locate, design, establish, build, construct, equip, operate, make, use, administer, manage and maintain service, improve, inspect, enlarge, alter, protect, develop, extend, repair, replace, refurbish, pull down and remove, and carry out work (including without limitation dredging works) in respect of, the whole or any part or parts buildings, plant, equipment and any facilities ancillary or incidental to the operation or use of the aforesaid or any of them including structures, erections, cables, pipelines, offices, works, warehouses, plants, platforms, laboratories, research stations, transport facilities, roads, railways, bridges, and structures of all kinds and to purchase or otherwise acquire, lease, charter, and take or let or hire any of the same, and to contribute to or assist in, or carry out any part of, any operation in respect of the same and to acquire, operate and maintain the licenses, consents, authorizations, wayleaves, casements and other rights capable or possibly capable of facilitating the aforesaid.
- 85. To invent, design, develop, construct, manufacture, produce, erect, assemble, test, alter, install, maintain, repair, renovate, refurnish, recondition, utilize, operate, manage, purchase, sell, hire, hireout, import, export, supply and otherwise deal in all kinds of equipment, apparatus, plant, machinery, appliances, articles, furniture, things, accessories, components, fittings, tools, materials, substances, products, systems, computers, computer programs and software which are required or are likely be required by the company for the purposes of, or in connection with, any of its businesses.
- 86. To purchase, subscribe for or otherwise acquire, and hold and deal with, any shares, stocks, debentures, bonds or securities of any other company.
- 87. To enter into, carry on and participate in financial transactions and operations of all kinds and to undertake, carry on and execute all kinds of financial, commercial, trading, trust agency and other operations.
- 88. To establish, design, acquire, produce, transmit, broadcast, publish, print and reproduce in any form whatsoever (including, without prejudice to the generality of the foregoing, visual or audible form and forms capable of being used by or in connection with computers), and to accept buy, sell and supply and otherwise deal in brochures, manuals, journals and periodicals, magazines newspapers, books, pictures, photographs, stationery and other documents, sound visual recordings, tapes, films and programs for radio,

television, cinema and other means of communication, (including, without prejudice to the generality of the foregoing, in forms of advertisement, publicity and promotional material for the Company or any company of which the Company is a member or which is in any manner controlled by or connected with the Company.

- 89. To carry on all or any of the business of consultants, advisers and suppliers of management, personnel, and training services, whether generally or in respect of one or more of the types of business or activity which the Company or any company of which the Company is a member or which is in any manner controlled by or connected with the Company has power to carry on, and to provide training and educational courses, instruction and materials, of every description for employees of the Company or of any company of which the Company is a member or which is in any manner controlled by or connected with the Company and for other persons.
- 90. To carry on all or any of the businesses of, and provide services associated with, engineers (including without limitation, electrical, gas, petroleum, environmental, drilling, construction, mechanical, heating, ventilation, civil, chemical, telecommunications, computer and data information engineers), environmental biologists, physicists, chemists, physicians and specialists in medicine, mechanics, technicians, geologists, draftsmen, designers, surveyors, architects, builders, painters and decorators.
- 91. To carry on all or any of the businesses of wholesalers, retailers and traders, whether generally or in relation to particular goods or commodities, advertisers, advertising agents, sales promoters, public relations agents and marketing agents and undertake promotional campaigns and competitions for itself and other persons, to undertake, promote and sponsor any product, service, event, individual or publication which in the opinion or the Company will promote advance or publicize any activity of the Company or of any company of which the Company is a member or which is in any manner controlled by or connected with the Company and generally to carry on the businesses of public relations agents, publicity consultants and marketing agents.
- 92. To carry on all or any of the businesses of running, operating, managing or co-operating in projects or works designed to restore, preserve, improve and protect the environment.
- 93. To carry on all or any of the business of running, operating, managing, maintaining, servicing, repairing, replacing, improving, altering and repairing and carry out works in respect of supplying and dealing in data processing and information retrieval systems, computers, computer programs and software, computer bureau, databases and services, facilities and equipment ancillary to, or for use in connection with, the same.
- 94. To carry on the business of purchasing, chartering, leasing, taking or letting on hire, operating, using, employing or turning to account, building, equipping, servicing, repairing, maintaining, supplying and deal in tankers and other ships and vessels and craft of every description (including without limitation, submersible craft) hovercrafts, motor vehicles, aircraft, airships, railways locomotives, wagons, trucks and any means of transport and parts and accessories of all kinds for any of the same.

- 95. To carry on the business of purchasing, taking on lease or on hire or otherwise acquiring, holding, developing, selling, hiring out, granting leases or licenses or otherwise disposing of or dealing in property of all and any kinds whether real or personal whether tangible or intangible and whether movable or immovable including intellectual property and any interest, right or privilege therein, for such consideration and on such terms as may be considered expedient.
- 96. To carry on the business of issuing, placing, underwriting, or guaranteeing, the subscription or assisting in the issuing, placing, underwriting or guaranteeing the subscription of shares, stocks, debentures, bonds and other securities of any company on such terms as to remuneration and otherwise as may be considered expedient.
- 97. To carry on business as traders, dealers, agents, importers, exporters, retailers, wholesalers and general merchants, stockman, brokers, commission agents of covering computers, peripherals, office automation, components, software and electronic goods, video equipment black and white and color television sets, cassettes, spools and cartridges, tape recorders, tape decks, tape consoles, T.V. Picture tubes, magnetic tapes of semi-conductor devices, integrated circuits, all types of capacitors, transformers, coils, transistors, and valves, radio, household appliances, mechanical, electrical, refrigeration, air conditioning and chemical machinery, apparatus tools, spare parts, appliances and goods of every description whether solid, liquid or gaseous.
- 98. The Company hereby acknowledges to adhere and comply with the conditions and terms stipulated in the licence agreement entered into by the Company with the Department of Telecommunications, Ministry of Communications and Information Technology, Government of India ("DoT") for carrying on any of the above businesses including the cellular mobile telephone service license agreement, unified access service license agreement, national long distance license agreement and the international long distance license agreement, as amended or substituted from time to time. Under extant requirements of the DoT, any violation of any such license agreements shall automatically lead to the Company being unable to carry on its business in this regard.
- IV. The liability of the members is limited.
- ^#V. The Authorised Share Capital of the Company is Rs. 10,00,00,00,000 (Rupees One Lakh Crore only) divided into 95,00,00,00,000 (Nine Thousand Five Hundred Crore) equity shares of Rs. 10/- (Rupees Ten) each and 5,00,00,00,000 (Five Hundred Crore) Preference Shares of Rs. 10/- (Rupees Ten) each, with the rights, privileges and conditions attached thereto as per the relevant provisions contained in that behalf in the Articles of Association of the Company and with the power to increase or reduce the capital of the Company and to divide the shares in the capital for the time being into several classes (being those specified in the Companies Act, 2013) and to attach thereto respectively such preferential qualified or special rights, privileges or conditions in such manner as may be permitted by the said Act or provided by the Articles of Association of the Company for the time being in force.

[^] The Authorised share capital clause of the Company has been altered and substituted vide Special Resolution passed by the members of the Company at the Extra-ordinary General Meeting held on 8th May 2024

[#] Earlier, the Authorised share capital clause of the Company has been altered and substituted vide Special Resolution passed by the members of the Company at the Extra-ordinary General Meeting held on 26th March 2022

We, the several persons whose names, address and occupation are subscribed hereunder are desirous of being formed into a Company in pursuance of these Memorandum of Association and we respectively agree to take the number of shares in the capital of the Company set opposite our respective names:

Name, Address, Description and occupation of each Subscriber	Number of Equity Shares taken by each Subscriber	Signature of Subscriber	Signature of Witness and his name, address, description and occupation	
Shri Mahesh Chandra Bagrodia S/o Shri K D Bagrodia B-2 15 th Floor , Prithvi Apartments,Altamount Road, Bombay 400 001	10 (Ten)	Sd/- M C Bargrodia		
Service				
Shri Raghuvir Bhandari S/o Dr Shree Krishna Raj Bhandari 1 Tahiti Co-op Housing Society 23 Juhu Versova Link Road Bombay 400 053	10 (Ten)	Sd/- R Bhandari		
Service				
Shri Sushil Kumar Saboo S/o Madan Lal Saboo Satnam Apartments 4 th Floor Colaba, Bombay 400 005	10 (Ten)	Sd/- S K Saboo	Witness to All Sd/- Shri Narayan Rathi S/o Shri Tulsidas Rathi R1/15 Goverdhangiri Society	
Service			Bangur Nagar	
Shri Gopi Krishna Tulsian S/o Laxmi Narayan Tulsian 301 Pradeep Worli Hill Estate Bombay 400 018	10 (Ten)	Sd/- G K Tulsian	Goregaon (W) Bombay 400 090 Company Secretary	
Service				
Shri Raghuram Raju S/o Shri K V Rama Raju E-GE DDA Flats Mayapuri New Delhi 110 064	10 (Ten)	Sd/- R Raju		
Advocate				
Shri Deepak Adalkha S/o Shri L D Adalkha E-25 A East of Kailash New Delhi 110 065	10 (Ten)	Sd/- D Adlakha		
Advocate				
Ms Jyoti Pande D/o Shri R C Pande 104 Friends Colony New Delhi 110 065	10 (Ten)	Sd/- J Pande		
Advocate				
Total	70 (Seventy)			

Place: Bombay Date: March 01, 1995

AMENDED AND RESTATED ARTICLES OF ASSOCIATION

UNDER THE COMPANIES ACT, 2013

COMPANY LIMITED BY SHARES

ARTICLES OF ASSOCIATION

OF

VODAFONE IDEA LIMITED*

PART I

INTERPRETATION

- 1. The regulations for the management of the Company and for the observance of the Members thereof and their representatives shall, subject to any exercise of the statutory powers by the Company with reference to the repeal or alterations or addition to its regulations by a Special Resolution as prescribed by the Companies Act, 2013, be such as are contained in the Articles set out herein below, and the regulations in Table F of Schedule I to the said Companies Act, 2013 shall not, except in respect of such of the matters for which no provisions exist in these Articles, apply to this Company. In the event of any conflict between Part I of the Articles of Association and Part II of the Articles of Association, the provisions of Part II of the Articles of Association shall prevail.
- 2. (a) Unless the context otherwise requires, words or expressions contained in these Articles shall bear the meanings assigned to them respectively hereunder.
 - (i) "Act" means the Companies Act, 2013 or any statutory modifications or reenactment thereof for the time being in force.
 - (ii) "Annual General Meeting" means a General Meeting of Members held in accordance with the provisions of Section 96 of the Act.
 - (iii) "Articles" or "Articles of Association" means these articles of association of the Company as originally framed or as altered from time to time in accordance with the provisions of the Act.
 - (iv) "Auditor" means and includes a Person appointed as such for the time being of the Company.
 - (v) "Audited Statement of Accounts" means the audited statement of accounts of the Company prepared by the Auditor in compliance with the Act and placed before the Members at the Annual General Meeting of the Company for approval.
 - (vi) "Auditor's Report" means a report prepared by the Auditor in addition to the Audited Statement of Accounts of the Company and placed before the Members at the Annual General Meeting of the Company for approval.

^{*} Name changed from "Idea Cellular Limited" to "Vodafone Idea Limited" at the Extra-ordinary General Meeting held on June 26, 2018 and confirmed by the Registrar of Companies, Ahmedabad (Gujarat) vide Certificate of Incorporation dated 31st August, 2018.

- (vii) "Board of Directors" or "Board" means the Board of Directors of the Company.
- (viii) "Capital" means the Share capital, for the time being raised or authorized to be raised, as the case may be, for the purposes of the Company.
- (ix) "Capital Redemption Reserve Account" has the meaning assigned to it in Article 7(a)(iv).
- (x) "Chairman" means the Chairman of the Board of Directors.
- (xi) "Chief Executive Officer" means the chief executive officer of the Company appointed, from time to time, by the Board of Directors as provided in Article 173.
- (xii) "Chief Financial Officer" means the chief financial officer of the Company appointed, from time to time, by the Board of Directors as provided in Article 174.
- (xiii) "Chief Technical Officer" means the chief technical officer or any other officer carrying on similar functions, by whatever name called, of the Company appointed, from time to time, by the Board of Directors.
- (xiv) "Committee" means a committee of the Board of Directors.
- (xv) "Company" means VODAFONE IDEA LIMITED.
- (xvi) "Company Secretary" or "the Secretary" means the company secretary of the Company appointed, from time to time, by the Board of Directors.
- (xvii) "Debenture" includes debenture-stock, bonds and other securities of the Company, whether constituting a charge on the assets of the Company, its subsidiaries, its holding company or its associate companies or not.
- (xviii) "Directors" means the Directors for the time being of the Company or, as the case may be, the Directors assembled at a Board.
- (xix) "Director's Report" means a report prepared by the Directors of the Company and placed before the Members of the Company at the Annual General Meeting, for their approval.
- (xx) "Dividend" includes bonus.
- (xxi) "DoT" means the Department of Telecommunications, Ministry of Communications and Information Technology, Government of India.
- (xxii) "Equity Capital" means the equity Shares in the Capital of the Company.
- (xxiii) "Extra-ordinary General Meeting" means an extra ordinary General Meeting of the Members other than Annual General Meeting duly called and constituted and any adjourned holding thereof.
- (xxiv) "Financial Year" has the meaning given to it under Section 2 (41) of the Act.
- (xxv) "In writing" and "Written" include printing, lithography and other modes of representing or reproducing words in a visible form.

- (xxvi) "Key Managerial Personnel" means the officers of the Company to be appointed in accordance with Section 203 of the Act.
- (xxvii) "Licence Agreements" means the licence agreements entered into by the Company with the DoT for carrying on any its business including the cellular mobile telephone service licence agreement, unified access service licence agreement, unified licence agreement, national long distance licence agreement, international long distance licence agreement and the internet service provider licence agreement or any other license agreement as amended or substituted from time to time.
- (xxviii) "Managing Director" means the managing director of the Company appointed, from time to time, by the Board of Directors as provided in Article 176.
- (xxix) "Member" means a duly registered holder, for the time being, of the Shares of the Company, and includes a subscriber to the Memorandum and Articles of Association of the Company.
- (xxx) "Meeting" or "General Meeting" means a general meeting of the Members held in accordance with the provisions of Section 96, Section 97 or Section 100 of the Act.
- (xxxi) "Memorandum" or "Memorandum of Association" means the memorandum of association of the Company as originally framed or as altered from time to time.
- (xxxii) "Office" or "Registered Office" means the registered office, for the time being, of the Company.
- (xxxiii) "Paid up" in relation to Shares includes credited as paid up.
- (xxxiv) "Person" means any individual, partnership (including any limited liability partnership), association, joint stock company, joint venture corporation, trust, unincorporated organisation or government, or agency or sub-division thereof.
- (xxxv) "Proxy" means any person who is appointed by an instrument to vote for a Member at a General Meeting in a poll.
- (xxxvi) "Register of Charges" means the register of charges to be kept pursuant to Section 85 of the Act.
- (xxxvii) "Register of Contracts" means the register of contracts to be kept pursuant to Article 168.
- (xxxviii) "Register of Directors" means the register of directors to be kept pursuant to Article 168.
- (xxxix) "Register of Members" or "Register and Index of Members" means the register of Members to be kept pursuant to the Act.
- (xl) "Register and Index of Debenture-holders" shall mean the register of debenture-holders required to be kept pursuant to the Act.
- (xli) "Register of Renewed and Duplicate Certificates" shall mean the register of renewed and duplicate Share certificates required to be kept pursuant to Article 28(c)(v).

- (xlii) "Register of Transfers and Transmissions" shall mean the register of transfers and transmission of Shares required to be kept pursuant to Article73.
- (xliii) "Registrar" means the Registrar of Companies of the State in which the Registered Office of the Company is, for the time being, situate.
- (xliv) "Regulations" or the "Company's Regulations" means the regulations or bylaws, for the time being, framed by the Company.
- (xlv) "Rules" means the applicable rules for the time being in force as prescribed under relevant Sections of the Act.
- (xlvi) "Seal" or "Common Seal" means the common seal, for the time being, of the Company.
- (xlvii) "Securities Premium Account" has the meaning given to it in Article 24(a).
- (xlviii) "Senior Officers" mean the Chief Financial Officer, the Chief Technical Officer, the Company Secretary and any other key positions of the Company as may be notified by the DoT from time to time.
- (xlix) "Share" means shares in the Capital of the Company, and includes stock except where a distinction between stock and shares is expressed or implied.
- (l) "Shareholder" means any owner of the Shares in the Capital of the Company.
- (li) "Section" or "Sections" means a section of the Act for the time being in force.
- (lii) "Special Resolution" and "Ordinary Resolution" have the meanings respectively given to them under the provisions of the Act.
- (liii) "Stock Exchange" means the relevant stock exchange on which the Shares or Debentures of the Company are listed.
- (b) Any reference in these Articles to:
 - (i) Any gender, whether masculine, feminine or neuter, shall be deemed to be construed as referring to the other gender or genders, as the case may be; and
 - (ii) Singular number be construed as referring to, the plural number and vice versa.
- (c) The headings are inserted for convenience only and do not affect the interpretation of these Articles.
- (d) Save as aforesaid, any words or expressions defined in the Act, but not defined in these Articles and not inconsistent with the subject or context, bear the same meaning herein as given to them respectively in the Act.

3. CAPITAL AND INCREASE AND REDUCTION OF CAPITAL

- (a) The authorised Capital of the Company shall be as stated in Clause V of the Memorandum of Association.
- (b) Upon an acquisition, whether by merger, issue of Shares or otherwise by the Company of another body corporate, the Company may issue new Shares in its Equity Capital at par or at such premium as may be agreed by Shareholders.

- (c) Subject to the provisions of Article 88(g) and Section 43 of the Act and the Rules, the Company may, with the approval of the Shareholders in a General Meeting, issue Shares with differential rights as to Dividend, voting or otherwise.
- 4. (a) The Company in a General Meeting, may, by Special Resolution and subject to the provisions of these Articles, from time to time, increase the authorized Capital by the creation of new Shares, such increase to be of such aggregate amount and to be divided into Shares of such respective amounts as the resolution shall prescribe.

Subject to the provisions of Sections 43, 47, 55 and 62 of the Act, the new Shares shall be issued upon such terms and conditions, and with such rights and privileges annexed thereto, as the General Meeting shall direct and if no direction be given, as the Board of Directors shall determine, and in particular, such Shares may be issued with a preferential or qualified right to Dividends and in the distribution of the assets of the Company and subject to the provisions of the said Sections with special or without any right of voting and subject to the provisions of Section 55 of the Act any preference Shares may be issued on the terms that they are liable to be redeemed.

- (b) Whenever the Capital of the Company is increased under the provisions of this Article the Board of Directors shall comply with the provisions of Section 64 of the Act.
- 5. Except so far as otherwise provided by the conditions of issue or by these Articles, any Capital raised by the creation of new Shares, shall be considered as part of the original Capital, and shall be subject to the provisions herein contained with reference to the payment of calls and instalments, forfeiture, lien, surrender, transfer and transmission, voting or otherwise.
- 6. (a) Subject to the provisions of these Articles, where at any time after the expiry of two years from the formation of the Company or at any time after the expiry of one year from the allotment of Shares made for the first time after formation of the Company, whichever is earlier, it is proposed to increase the subscribed Capital of the Company by allotment of further Shares:
 - (i) Such further Shares shall be offered to the Persons who, as on the date of the offer, are holders of the equity Shares of the Company, in proportion, as nearly as circumstances admit to the Capital paid-up on those Shares at that date;
 - (ii) Such offer shall be made by a notice specifying the number of Shares offered and stipulating a time not being less than fifteen days and not more than thirty days from the date of the offer within which the offer, if not accepted, shall be deemed to have been declined;
 - (iii) The offer aforesaid shall be deemed to include a right exercisable by the Person concerned to renounce the Shares offered to him or any of them in favour of any other Person; and the notice referred to herein above shall contain a statement of this right.
 - Provided that the Directors may decline, without assigning any reason, to allot any Shares to any Person in whose favour any Member may renounce Shares offered to him; and
 - (iv) After the expiry of the time specified in the notice aforesaid or on receipt of earlier intimation from the Person to whom such notice is given that he declines to accept the Shares offered, the Board may dispose of such Shares in such manner as the Board thinks most beneficial to the Company.

- (b) Notwithstanding anything contained in the preceding sub-Article, the Company may by a Special Resolution offer further Shares to any Person or Persons either for cash or for consideration other than cash, and such Person or Persons may or may not include the Persons who at the date of the offer, are the holders of the equity Shares of the Company.
- (c) Nothing contained in sub-Articles (a) and (b) of this Article 6 shall apply
 - (i) to the increase of the subscribed Capital caused by the exercise of an option attached to any Debentures issued or loans raised by the Company;
 - (ii) to the conversion of such Debentures or loans into Shares in the Company; or
 - (iii) to the subscription of Shares in the Company.

Provided that the terms of the issue of such Debentures or the terms of such loans include a term providing for such option and such term as may be mutually agreed upon before the issue of the Debentures or before the raising of the loans or is in conformity with the rules, if any, made by the Central Government in this behalf or by such authorities as may be laid down by the Central Government.

- (iv) in the case of Debentures or loans, other than those Debentures issued to, or loans obtained from the Central Government or any institution specified by the Central Government in this behalf, has also been approved be a Special Resolution passed by the Company in a General Meeting before the issue of the Debentures or the raising of the loans.
- (v) subject to the provisions of the Act, and these Articles, the Board of Directors may issue and allot Shares in the Capital of the Company on payment or part payment for any property or assets of any kind whatsoever sold or transferred, goods or machinery supplied or for services rendered to the Company in the conduct of its business and any Shares which may be so allotted may be issued as fully paid up or partly paid up otherwise than in cash, and if so issued, shall be deemed to be fully paid up or partly paid up Shares as the case may be.
- (d) Nothing in sub-article (a)(iii) of Article 6 hereof shall be deemed:
 - (i) To extend the time within which the offer should be accepted; or
 - (ii) To authorise any Person to exercise the right of renunciation for a second time on the ground that the Person in whose favour the renunciation was first made has declined to take Shares comprised in the renunciation.
- (e) Any Debentures, debenture-stock or other securities may be issued at a discount, premium or otherwise and may be issued on condition that they shall be convertible into Shares of any denomination and with any privileges and conditions as to redemption, surrender, drawing, allotment of Shares, attending (but not voting) at the General Meeting, appointment of Directors and otherwise. Debentures with the right to conversion into or allotment of Shares shall be issued only with the consent of the Company in the General Meeting by a Special Resolution.
- 7. (a) Subject to the provisions of Section 43, Section 55 and other applicable provisions, if any, of the Act and the Rules and the provisions of these Articles, the Company shall by a Special Resolution have power to issue or re-issue preference Shares / cumulative convertible preference Shares of one or more classes which are liable to be redeemed

or converted to equity Shares, with such rights and on such terms and conditions that are prescribed in this behalf from time to time.

Provided that:

- (i) No such Shares shall be redeemed except out of the profits of the Company which would otherwise be available for Dividend or out of the proceeds of a fresh issue of Shares made for the purposes of redemption;
- (ii) No such Shares shall be redeemed unless they are fully paid;
- (iii) The premium, if any, payable on redemption shall have been provided for out of the profits of the Company or out of the Company's Securities Premium Account before the Shares are redeemed;
- (iv) Where any such Shares are redeemed otherwise than out of the proceeds of a fresh issue, there shall, out of the profits which would otherwise have been available for Dividend, be transferred to a reserve fund, to be called "the Capital Redemption Reserve Account", a sum equal to the nominal amount of the Shares redeemed; and the provisions of the Act relating to the reduction of the Capital of the Company shall, except as provided in Section 55 of the Act and the Rules apply as if the Capital Redemption Reserve Account were paid up Capital of the Company.
- (b) Subject to the provisions of Section 55 of the Act and the Rules and subject to the provisions on which any Shares may have been issued, the redemption of preference Shares may be effected on such terms and in such manner as may be provided in these Articles or by the terms and conditions of their issue and subject thereto in such manner as the Directors may think fit.
- (c) The redemption of preference Shares under the provisions of their issue by the Company shall not be taken as reducing the amount of its authorised Capital.
- (d) Where in pursuance of this Article, the Company has redeemed or is about to redeem any preference Shares, it shall have the power to issue Shares up to the nominal amount of the Shares redeemed or to be redeemed as if those Shares had never been issued and, accordingly, the Capital of the Company shall not, for the purpose of calculating the fees payable under Section 403 of the Act, be deemed to be increased by the issue of Shares in pursuance of this Article.
 - Provided that where new Shares are issued before the redemption of the old Shares, the new Shares shall not, so far as relates to stamp duty, be deemed to have been issued in pursuance of this Article unless the old Shares are redeemed within one month of the issue of the new Shares.
- (e) The Capital Redemption Reserve Account may, notwithstanding anything contained in this Article 7, be applied by the Company, in paying up unissued Shares of the Company to be issued to Members of the Company as fully paid bonus Shares.
- 8. The Company shall be at liberty at any time, either at one time or from time to time as the Company shall think fit, by giving not less than six months' previous notice in writing to the holders of the preference Shares, to redeem at par the whole part of the preference Shares, for the time being outstanding, by payment of the nominal amount thereof with Dividend calculated up to the date or dates notified for payment (and for this purpose the Dividend shall be deemed to accrue and be due from day to day) and in the case of redemption of part of the preference

Shares the following provisions shall take effect:

- (a) The Shares to be redeemed shall be determined by drawing of lots which the Company shall cause to be made at its Registered Office in the presence of at least one of the Directors; and
- (b) Forthwith after every such drawing, the Company shall notify to the Shareholders whose Shares have been drawn for redemption, its intention to redeem such Shares by payment at the Registered Office of the Company at the time and on the date to be named against surrender of the certificates in respect of the Shares to be so redeemed and at the time and date so notified each such Shareholder shall be bound to surrender to the Company the Share certificates in respect of the Shares to be redeemed and thereupon the Company shall pay the amount payable to such Shareholders in respect of such redemption. The Shares to be redeemed shall cease to carry Dividend from the date named for payment as aforesaid. Where any such certificate comprises any Shares which have not been drawn for redemption, the Company shall issue to the holder thereof a fresh certificate therefor.
- 9. (a) The Company may (subject to the provisions of Sections 55 and 66 of the Act and the provisions of these Articles), from time to time by a Special Resolution, reduce its Capital and any Capital Redemption Reserve Account or Securities Premium Account and / or any other reserve in the nature of Capital in any manner for the time being authorised by law, and in particular Capital may be paid off on the footing that it may be called up again or otherwise.
 - (b) This Article 9 shall not derogate from any power the Company would have if it were omitted.
- 10. Subject to the provisions of Section 61 of the Act and these Articles, the Company in a General Meeting may, from time to time, by a Special Resolution alter the conditions of its Memorandum of Association so as to:
 - (a) increase its Capital by such amount as it thinks expedient by issuing new Shares;
 - (b) consolidate and divide all or any of its Capital into Shares of larger amount than its existing Shares; Provided that any consolidation and division which results in changes in the voting percentage of Members shall require the applicable approvals under the Act:
 - (c) convert all or any of its fully paid up Shares into stock; and reconvert that stock into fully paid up Shares of any denomination;
 - (d) sub-divide its Shares or any of them into Shares of smaller amount than is fixed by the Memorandum of Association of this Company subject nevertheless to the provisions of the Act in that behalf and so however, that in the sub-division, the proportion between the amount paid and the amount, if any, unpaid on each reduced Share shall be the same as it was in the case of the Share from which the reduced Share is derived, and the resolution whereby any Share is sub-divided may determine that, as between the holders of the Shares resulting from such sub-division, one or more of such Shares shall have some preference or special advantage as regards Dividend, capital or otherwise, over, or as compared with, the others or other; and
 - (e) cancel Shares which at the date of passing of the resolution in a General Meeting in that behalf have not been taken or agreed to be taken by any Person, and diminish the amount of its Capital by the amount of the Shares so cancelled.

- 11. (a) If the Company has:
 - (i) consolidated and divided its Capital into Shares of a larger amount than its existing Shares;
 - (ii) converted any Shares into stock;
 - (iii) reconverted any stock into Shares;
 - (iv) sub-divided its Shares or any of them;
 - (v) redeemed any redeemable preference Shares; or
 - (vi) cancelled any Shares otherwise than in connection with a reduction of Capital under Section 66 of the Act, the Company shall within one month after doing so, give notice thereof to the Registrar specifying as the case may be, the Shares consolidated, divided, converted, sub-divided, redeemed or cancelled or the stocks reconverted.
 - (b) The Company shall thereupon request the Registrar to record the notice and make any alterations which may be necessary in the Company's Memorandum or Articles or both.
- 12. Subject to the provisions of the Act and the Securities and Exchange Board of India (Employee Stock Option Scheme and Employee Stock Purchase Scheme) Guidelines, 1999 and other guidelines issued in this context, the Company may at any time authorize the Board to create or implement one or more employee stock option plans or employee stock purchase plans, which may run simultaneously to any issue of Shares or securities to its employees and/or any other Persons whose contributions to the Company's performance including profitability is of material importance. Subject to applicable law, the Board may, at its discretion, create one or more trusts or other special purpose vehicles of any nature, and/or any other mechanism to implement one or more employee stock option plans or employee stock purchase plans and/or use the offices of any intermediaries to conceptualize, implement, manage, and/or administer any such schemes from time to time.
- 13. (a) Whenever the Capital, by reason of the issue of preference Shares, or otherwise, is divided into different classes of Shares, all or any of the rights and privileges attached to each class (unless otherwise provided by the terms of issue of the Shares of that class) may, subject to the provisions of Section 48 of the Act and these Articles, and whether or not the Company is being wound up, be varied, modified, commuted, affected or abrogated, or dealt with the consent in writing of the holders of at least three–fourths in nominal value of the issued Shares of that class or with the sanction of a Special Resolution passed at a separate Meeting of the holders of Shares of that class and all the provisions hereinafter contained as to a General Meeting shall, *mutatis mutandis*, apply to every such Meeting.
 - (b) This Article shall not derogate from any power the Company would have if this Article were omitted.

SHARES AND SHARE CERTIFICATES

14. The rights or privileges conferred upon the holders of the Shares of any class issued with preference or other rights, shall not unless otherwise expressly provided by the terms of issue of the Shares of that class, be deemed to be varied or modified or affected by the creation or issue of further Shares ranking *pari passu* therewith.

- 15. The provisions of Sections 43 and 47 of the Act in so far as the same may be applicable shall be observed by the Company.
- 16. Subject to the provisions of the Act and these Articles, the Shares (including any Shares forming part of any increased Capital of the Company) in the Capital of the Company for the time being shall be under the control of the Board of Directors who may issue, allot or otherwise dispose of the same or any of them to such Persons, in such proportion and on such terms and conditions and either at a premium or at par or (subject to the compliance with the provisions of Section 53 of the Act) at a discount and at such time as they may from time to time think fit and with the sanction of the company in the General Meeting to give to any Person or Persons the option or right to call for any Shares either at par or premium during such time and for such consideration as the Directors think fit, and may issue and allot Shares in the Capital of the Company on payment in full or part of any property sold and transferred or for any services rendered to the company in the conduct of its business and any Shares which may so be allotted may be issued as fully paid up Shares and if so issued, shall be deemed to be fully paid Shares. Provided that option or right to call of Shares shall not be given to any Person or Persons without the sanction of the Company in the General Meeting.
- 17. The Board shall observe the restrictions as to allotment contained in Section 39 of the Act and shall cause to be made the return of allotment in accordance with Section 39 of the Act and the Rules.
- 18. In addition to and without derogating from the powers for that purpose conferred on the Board under Article 9 hereof and with the sanction of the Company in a General Meeting may, subject to the provisions of Section 62 of the Act, determine that any Shares (part of the original Capital or of any increased Capital of the Company) shall be offered to such Persons (whether Members or not) in such proportion and on such terms and conditions and either (subject to compliance with the provisions of Sections 52 and 53 of the Act) at a premium or at par or at a discount, as such General Meeting shall determine and with full power to give any Person/s (whether a Member or not) the option to call or be allotted Shares of any class of the Company either (subject to compliance with the provisions of Sections 52 and 53 of the Act) at a premium or at par or at a discount, such option being exercisable at such times and for such consideration as may be directed by such General Meeting, or the Company in General Meeting may make any other provision whatsoever for the issue, allotment and disposal of any Shares.
- 19. The Company shall cause to be kept a Register of Members and an Index of Members in accordance with Section 88 of the Act and Register and Index of Debenture-holders in accordance with Section 88 of the Act. The Company shall be entitled to keep in any State or country outside India, a branch Register of Member resident in that State or country.
- 20. The Shares in the Capital shall be numbered progressively according to their several denominations, and except in the manner herein before mentioned, no Share shall be subdivided. Every forfeited or surrendered Share shall continue to bear the number by which the same was originally distinguished.
- 21. (a) An application signed by or on behalf of an applicant for Shares in the Company, followed by an allotment of any Share therein, shall be an acceptance of Shares within the meaning of these Articles.
 - (b) Every Person who thus or otherwise accepts any Shares and whose name is entered in the Register of Members shall, for the purposes of these Articles, be a Member.

- 22. The money, if any, which the Board of Directors shall, on the allotment of any Shares being made by it, required or direct to be paid by way of deposit, call or otherwise, in respect of the Shares so allotted, shall immediately on the insertion of the name of the allottee in the Register of Members as the holder of such Shares, become a debt due to and recoverable by the Company from the allottee thereof, and shall be paid by him accordingly.
- 23. Every Member shall pay to the Company the portion of the Capital represented by his Share or Shares which may, for the time being remain unpaid thereon, in such amounts, at such time or times, and in such manner, as the Board of Directors shall, from time to time, in accordance with the Company's Regulations require or fix for the payment thereof.
- 24. (a) Where the Company issues Shares at a premium, whether for cash or otherwise, a sum equal to the aggregate amount or value of the premium on those Shares shall be transferred to an account to be called "the Securities Premium Account" and the provisions of the Act relating to the reduction of the Capital of the Company shall except as provided in this Article, apply as if the Securities Premium Account were paid up Capital of the Company.
 - (b) The Securities Premium Account may, notwithstanding anything in sub-Article (a) above, be applied by the Company:
 - (i) in paying up unissued Shares of the Company to be issued to members of the Company as fully paid bonus Shares;
 - (ii) in writing off the preliminary expenses of the Company;
 - (iii) in writing off the expenses of or the commission paid or discount allowed on, any issue of Shares or Debentures of the Company;
 - (iv) in providing for the premium payable on the redemption of any redeemable preference Shares or of any Debenture of the Company; or
 - (v) for the purchase of its own shares or other securities under Section 68 of the Act.
- 25. If and whenever, as the result of issue of new or further Shares or any consolidation or subdivision of Shares, any Shares are held by Members in fractions, the Directors shall, subject to the provisions of the Act and the Articles and to the directions of the Company in a General Meeting, if any, sell those Shares, which Members hold in fractions, for the best price reasonably obtainable and shall pay and distribute to and amongst the Members entitled to such Shares in due proportion, the net proceeds of the sale thereof. For the purpose of giving effect to any such sale the Directors may authorize any Person to transfer the Shares sold to the purchaser thereof, comprised in any such transfer and he shall not be bound to see to the application of the purchase money nor shall his title to the Shares be affected by any irregularity or invalidity in the proceedings in reference to the sale.
- 26. (a) Every Member or allottee of Shares shall be entitled, without payment, to receive one or more certificates in marketable lots, for all the Shares of each class or denomination registered in his name, or if the Directors so approve (upon paying such fee as the Directors may from time to time determine; provided that such fee does not exceed the maximum fee agreed between the Company and the Stock Exchange) to several certificates, each for one or more of such Shares and the Company shall complete and

have ready for delivery such certificates within three months from the date of allotment, unless the conditions of issue thereof otherwise provide, or within fifteen days of the date of lodgement for transfer, transmission, sub-division, consolidation, renewal or endorsement of call of any of its Shares as the case may be. Every certificate of Shares shall be under the Seal of the Company, if any and shall specify the number and distinctive numbers of Shares in respect of which it is issued and amount paid-up thereon and shall be in such form as the Directors may prescribe or approve.

(b) Such certificate shall be issued only in pursuance of a resolution passed by the Board and on surrender to the Company of its letter of allotment or its fractional coupons of requisite value, save in cases of issues against letters of advice or acceptance or letters of renunciation, or in cases of issue of bonus Shares.

Provided that if the letter of allotment is lost or destroyed, the Board may impose, such reasonable terms, if any, as it thinks fit as to evidence and indemnity and the payment of out of pocket expenses incurred by the Company in investigating the evidence.

- (c) For any further certificate the Board shall be entitled, but shall not be bound, to prescribe the charge not exceeding Rs. 50 (Rupees fifty) per certificate or such other limit as may be prescribed by the Act and the Rules.
- 27. (a) Any two or more joint allotees of a Share shall, for the purpose of this Article 27, be treated as a single Member, and the certificate of any Shares, which may be the subject of joint ownership may be delivered to any one of such joint owners on behalf of all of them and the company shall not be borne to issue more than one certificate and delivery of a certificate of Shares to one of several joint holders shall be sufficient delivery to all such holders.
 - (b) Subject to the provisions of the Act and the Rules or any statutory modification or reenactment thereof, for the time being in force, every such certificate shall be issued under the Seal, if any, which shall be affixed in the presence of (i) two Directors or Persons acting on behalf of the Directors under duly registered power of attorney and (ii) the Secretary or some other Person appointed by the Board for the purpose. The two Directors or their attorneys and the Secretary or other Person so appointed shall sign the Share certificate. Provided that if the Company does not have a Seal, the share certificate shall be signed by two directors or by a director and the Secretary. Provided further that if the composition of the Board permits of it, at least one of the aforesaid two Directors shall be a Person other than a Managing Director or Whole-time Director.
 - (c) A Director may sign a Share certificate by affixing his signature thereon by means of any machine, equipment or other mechanical material used for the purpose.
 - (d) Particulars of every Share certificate issued shall be entered in the Register of Members against the name of the Person to whom it has been issued indicating the date of issue and the amount paid up thereon.
 - (e) The Company shall comply with the provisions of Section 56 of the Act.
- 28. (a) No certificate of any Share or Shares shall be issued either in exchange for those which are sub-divided or consolidated or in replacement of those which are defaced, torn or old, decrepit, worn out, or where the cages on the reverse for recording transfers have been duly utilized, unless the certificate in lieu of which it is issued is surrendered to the Company.

- (b) The Company may charge such fee not exceeding that which may be agreed upon by the Company and the Stock Exchange per certificate, issued on splitting or consolidation of Share certificate except for marketable lots. Provided that no fee shall be charged for issue of new certificates in replacement of those which are old, defaced or worn out or where there is no further space on the back thereof for endorsement of transfer or pursuant to a scheme of arrangement sanctioned by the National Company Law Tribunal or the Central Government.
- (c) (i) Notwithstanding anything contained in the preceding Article, the Directors of the Company may in their absolute discretion refuse sub-division of Share certificates or Debenture certificates into denominations of less than the marketable lots except where such sub-division is required to be made to comply with a statutory provision or an order of a competent Court of law.
 - (ii) When a new Share certificate has been issued in pursuance of Article 28(a) of this Article, it shall state on the face of it and against the stub or counterfoil to the effect that it is "issued in lieu of share certificate No. subdivided/replaced/on consolidation of Shares".
 - (iii) If a Share certificate is lost or destroyed, a new certificate in lieu thereof shall be issued only with the prior consent of the Board upon payment of such fees as may be agreed upon between the Company and the Stock Exchange, and on such terms, if any, as to evidence and indemnity and payment of out of pocket expenses incurred by the Company in investigating the evidence, as the Board thinks fit. The Company will issue such duplicate certificate within six weeks of notification of loss and receipt of proper indemnity.
 - (iv) When a new Share certificate has been issued in pursuance of Article 28(c)(iii) of this Article, it shall state on the face of it and against the stub or counterfoil to the effect that it is "duplicate issued in lieu of Share certificate No......". The word "duplicate" shall be stamped or punched in bold letters across the face of the Share certificate.
 - (v) Where a new Share certificate has been issued in pursuance of Article 28(c)(i) or Article 28(c)(iii) of this Article, particulars of every such Share certificate shall be entered in a Register of Renewed and Duplicate Certificates indicating against the name(s) of the Person(s) to whom the certificate is issued, the number and date of issue of share certificate in lieu of which the new certificate is issued, and the necessary changes indicated in the Register of Members by suitable cross references in the "Remarks" column.
 - (vi) Provided that notwithstanding what is stated above the Directors shall comply with such rules or regulations or requirements of any Stock Exchange or the Rules or the rules made under Securities Contracts (Regulation) Act, 1956 or any other statutes or rules applicable in this behalf, including intimating the Stock Exchange within forty eight hours of receipt of information regarding loss of share certificates and issue of duplicate certificates, both by way of floppy disks and printed details.
- 29. (a) All blank forms to be used for issue of Share certificates shall be printed and the printing shall be done only on the authority of a resolution of the Board.
 - (b) The blank forms shall be consecutively machine numbered and the forms and the blocks, engravings, facsimiles and hues relating to the printing of such forms shall be kept in the custody of the Secretary or such other Person as the Board may appoint for

the purpose.

- (c) The Secretary or the Person aforesaid shall be responsible for rendering an account of these forms to the Board.
- (d) The Managing Director of the Company, for the time being, or if the Company has no Managing Director, every Director of the Company and the Secretary, if any, shall be responsible for the maintenance, preservation and safe custody of all books and documents relating to the issue of Share certificates, except the blank forms of Share certificates referred to in Article 29(a). All books referred to herein shall be preserved in good order permanently.
- 30. The provisions of the foregoing Articles relating to issue of certificates shall *mutatis mutandis* apply to the issue of certificates for any other securities including Debentures (except where otherwise provided by the Act) of the Company.
- 31. If any Share stands in the names of two or more Persons, the Person first named in the Register of Members shall, as regards receipt of Dividends or bonus, or service of notices and all other matters connected with the Company, except voting at Meetings, and the transfer of the Shares be deemed the sole holder; but the other joint holder(s) of the same shall not be relieved of his / their obligations in respect of payment of all instalments and calls due on the Share and all incidents thereof in accordance with the Rules.
- 32. (a) Except as ordered by a Court of competent jurisdiction or as required by law, the Company shall not be bound to recognize any benami, trust of equity or any equitable, contingent, future or partial interest in the Shares, or except only as is by these Articles otherwise expressly provided, any right in respect of a Share other than an absolute right thereto, in accordance with these Articles, in the Persons who are from time to time, registered as the holders thereof; but the Board shall be at liberty, at its sole discretion, to register any Share in the joint names of any two or more Persons or the survivor or survivors of them.
 - (b) Shares may be registered in the name of an incorporated Company or other body corporate but not in the name of a minor (except in case where they are fully paid) or in the name of a Person of unsound mind, or in the name of any firm or partnership or trust.
- 33. Funds of the Company shall not be applied in the purchase of any Shares of the Company, and it shall not give any financial assistance for or in connection with the purchase or subscription of any Shares in the Company or in its holding company, save as provided by Section 67 of the Act.
- 34. The Company may, by Special Resolution, purchase its own Shares or other securities, subject to such limits and on such terms and conditions specified under Sections 68 to 70 and other applicable provisions of the Act and the Rules.
- 35. The provisions of the Article under this heading shall *mutatis mutandis* apply to Debentures of the Company.
- 36. Dematerialisation of Securities

- (a) For the purpose of this Article:
 - (i) "SEBI" means the Securities and Exchange Board of India established under section 3 of the Securities and Exchange Board of India Act, 1992.
 - (ii) "Depositories Act" means the Depositories Act, 1996, including any statutory modifications thereof for the time being in force.
 - (iii) "Depository" means a company formed and registered under the Companies Act, 1956 and which has been granted a certificate of registration under subsection 1A of section 12 of the Securities and Exchange Board of India Act, 1992.
 - (iv) "Bye-laws" means bye-laws made by a Depository under section 26 of the Depositories Act.
 - (v) "Beneficial Owner" means a Person whose name is recorded as such with a depository.
 - (vi) "Member" means the duly registered holder of from time to time of the Shares of the Company and includes every Person whose name is entered as a Beneficial Owner in the records of the Company.
 - (vii) "Participant" means a Person registered as such under section 12A of the Securities and Exchange Board of India Act, 1992.
 - (viii) "Records" includes the records maintained in the form of books or stored in computer or in such other form as may be determined by regulations made by the SEBI in relation to the Depositories Act.
 - (ix) "Regulations" means the regulations made by SEBI.
 - (ix) "Security" means such security as may be specified by SEBI.
 - (x) Words imparting the singular number only include the plural number and viceversa.
 - (xii) Words and expressions used but not defined in the Act but defined in the Depositories Act, shall have the same meanings respectively assigned to them in that Act.
- (b) Either the Company or the investor may exercise an option to issue, deal in, hold the securities (including Shares) with a Depository in electronic form and the certificate in respect thereof shall be dematerialised, in which event the rights and obligations of the parties concerned and matters connected therewith or incidental thereof, shall be governed by the provisions of the Depositories Act, as amended from time to time or any statutory modification thereto or re-enactment thereof.
- (c) Notwithstanding anything contained in these Articles, the Company shall be entitled to dematerialise its existing securities, rematerialize its securities held in the Depositories and / or offer its fresh securities in a dematerialised form pursuant to the Depositories Act, and the rules framed thereunder, if any.
- (d) Every Person subscribing to or holding securities of the Company shall have the option to receive security certificates or to hold the securities with a Depository. If a Person opts to hold his security with a Depository, the Company shall intimate such Depository the details of allotment of the security, and on receipt of the information, the Depository

shall enter in its records the name of the allottees as the Beneficial Owner of the security.

- (e) All securities held by a Depository shall be dematerialised and be in fungible form. Nothing contained in Section 89 and other applicable provisions of the Act, shall apply to a Depository in respect of the securities held by it on behalf of the Beneficial Owner.
- (f) (i) Notwithstanding anything to the contrary contained in the Act, or these Articles, a Depository shall be deemed to be registered owner for the purposes of effecting transfer of ownership of security on behalf of the Beneficial Owner.
 - (ii) Save as otherwise provided in (i) above, the Depository as the registered owner of the securities shall not have any voting rights or any other rights in respect of the securities held by it.
 - (iii) Every Person holding securities of the Company and whose name is entered as the Beneficial Owner in the records of the Depository shall be deemed to be a member of the Company. The Beneficial Owner of the securities shall be entitled to all the rights and benefits and be subject to all the liabilities in respect of his securities which are held by a Depository.
- (g) Except as ordered by any Court of competent jurisdiction or as required by any law, the Company shall be entitled to treat the Person whose name appears on the Register of Members as the holder of any Share or where the name appears as the Beneficial Owner of the Shares in the records of the Depository as the absolute owner thereof and accordingly shall not be bound to recognise any benami trust or equitable, contingent, future or partial interest in any Share, or (except only as is by these Articles otherwise expressly provided) any right in respect of a Share other than an absolute right thereto in accordance with these Articles, on the part of any other Person whether or not it has express or implied notice thereof, but the Board shall be entitled at their sole discretion to register any Share in the joint names of any two or more Persons or the survivors or survivors of them.
- (h) Every Depository shall furnish to the Company information about the transfer of securities in the name of the Beneficial Owner at such intervals and in such manner as may be specified by the bye-laws and the Company in that behalf.
- (i) Upon receipt of certificates of securities on surrender by a Person who has entered into an agreement with the Depository through a Participant, the Company shall cancel such certificate and substitute in its records the name of Depository as the registered owner in respect of the said securities and shall also inform the Depository accordingly.
- (j) If a Beneficial Owner seeks to opt out of a Depository in respect of any security, the Beneficial Owner shall inform the Depository accordingly.
 - The Depository shall on receipt of information as above make appropriate entries in its records and shall inform the Company.
 - The Company shall within thirty days of the receipt of intimation from the Depository and on fulfilment of such conditions and payment of such fees as may be specified by the regulations, issue the certificate of securities to the Beneficial Owner or the transferee as the case may be.
- (k) Notwithstanding anything in the Act, or these Articles, to the contrary, where securities are held in a Depository, the records of the beneficial ownership may be served by such Depository on the Company by means of electronic mode or by delivery of floppies or discs.

- (l) Except as specifically provided in these Articles, the provisions relating to joint holders of Shares, calls, lien on Shares, forfeiture of Shares and transfer and transmission of Shares shall be applicable to Shares held in Depository so far as they apply to Shares in physical form subject to the provisions of the Depository Act.
- (m) Notwithstanding anything in the Act, or these Articles where securities are dealt with by a Depository, the Company shall intimate the details thereof to the Depository immediately on allotment of such securities.
- (n) The Shares in the Capital shall be numbered progressively according to their several denominations provided, however, that the provision relating to progressive numbering shall not apply to the Shares of the Company which are dematerialised or may be dematerialised in future or issued in future in dematerialised form. Except in the manner hereinabove mentioned, no Share shall be sub-divided. Every forfeited or surrendered Share held in material form shall continue to bear the number by which the same was originally distinguished.
- (o) The Company shall cause to keep a Register and Index of Members and Register and Index of Debenture-holders in accordance with Section 88 of the Act, and the Depositories Act, with details of Shares and debentures held material and dematerialised forms in any media as may be permitted by law including in any form of electronic media. The Register and Index of Beneficial Owners maintained by a Depository under section 11 of the Depositories Act, shall be deemed to be Register and Index of Members and Register and Index of Debenture-holders, as the case may be, for the purpose of the Act. The Company shall have the power to keep in any state or country outside India a branch Register of Members resident in that State or country.

UNDERWRITING COMMISSION AND BROKERAGE

- 37. (a) The Company may pay commission to any Person in consideration of:
 - (i) his subscribing or agreeing to subscribe whether absolutely or conditionally, for any Shares in, or Debentures of the Company, subject to the restrictions specified in sub-section (6) of Section 40 of the Act and the Rules; or
 - (ii) his procuring or agreeing to procure subscriptions, whether absolute or conditional for any Shares in, or Debentures of the Company to be offered to the public, if the following conditions are fulfilled, namely:
 - (1) the commission is paid out of the proceeds of the issue, the profits of the Company, or both;
 - (2) the commission paid or agreed to be paid does not exceed in the case of Shares, five percent of the price at which the Shares are issued and in the case of Debentures, two and half percent of the price at which the Debentures are issued;
 - (3) the name of the underwriters and the amount or rate percent of the commission paid or agreed to be paid, on Shares or Debentures offered to the public for subscription, is disclosed in the prospectus and filed before the payment of the commission with the Registrar, and where a circular or notice, not being a prospectus inviting subscription for the Shares or Debentures is issued is also disclosed in that circular or notice;
 - (4) the number of Shares or Debentures which such Persons have agreed for a commission to subscribe, absolutely or conditionally is disclosed

in the manner aforesaid; and

- (5) a copy of the contract for the payment of commission is delivered to the Registrar at the time of delivery of the prospectus for registration.
- (b) Nothing in this Article 37 shall affect the power of the Company to pay such brokerage as it has hereto before been lawful for the Company to pay.
- (c) A vendor to, promoter of, or other Person who receives payment in Shares, Debentures or money from the Company shall have and shall be deemed always to have had power to apply any part of the Shares, Debentures or money so received for payment of any commission, the payment of which, if made directly by the Company would have been legal under Section 40 of the Act.
- (d) The commission may be paid or satisfied (subject to the provisions of the Act, the Rules and these Articles) in cash, or in Shares, Debentures or debenture-stocks of the Company.

CALLS AND LIENS

- 38. (a) The Board may, from time to time, make calls upon the Members in respect of any monies unpaid on their Shares (whether on account of the nominal value of the Shares or by way of premium) and not by the conditions of allotment thereof made payable at fixed times. Provided that no call shall exceed one fourth of the nominal value of the Share or be payable at less than one month from the date fixed for payment of the last preceding call.
 - (b) Each member shall, subject to receiving at least fourteen days' notice specifying the time or times and place of payment, pay to the Company, at the time or times and place so specified, the amount called on his Shares. The call notice shall be in a standard form acceptable to the Stock Exchange.
 - (c) A call may be revoked or postponed at the discretion of the Board.
- 39. A call shall be deemed to have been made at the time when the resolution of the Board authorizing the call was passed and may be required to be paid by instalments.
- 40. The joint holders of a Share shall be jointly and severally liable to pay all calls in respect thereof.
- 41. (a) If a sum called in respect of a Share is not paid before or on the day appointed for payment thereof, the Person from whom the sum is due shall pay interest thereon from the day appointed for payment thereof to the time of actual payment at ten percent per annum or at such lower rate, if any, as the Board may determine.
 - (b) The Board shall be at liberty to waive payment of any such interest wholly or in part.
- 42. (a) Any sum which by the terms of issue of a Share becomes payable on allotment or at any fixed date, whether on account of the nominal value of the Share or by way of premium, shall, for the purposes of these Articles, e deemed to be a call duly made and payable on the date on which the terms of issue such sum becomes payable.
 - (b) In case of non-payment of such sum, all the relevant provisions of these Articles as to

payment of interest and expenses, forfeiture or otherwise shall apply as if such sum had become payable by virtue of a call duly made and notified.

43. The Board:

- (a) may, if it thinks fit, receive from any Member willing to advance the same, all or any part of the monies uncalled and unpaid upon any Shares held by him; and
- (b) Upon all or any of the monies so advanced, may (until the same would, but for such advance, become presently payable) pay interest at such rate not exceeding, unless the Company in General Meeting shall otherwise direct, twelve percent per annum, as may be agreed upon between the Board and the Member paying the sum in advance.

The provisions of these Articles relating to calls shall *mutatis mutandis* apply to any other securities of the Company including Debentures.

- 44. Subject to the provisions of the Act and these Articles, on the trial or hearing of any suit, action or other proceeding brought by the Company against any Member or his representatives for the recovery of any money claimed to be due to the Company in respect of his Shares, it shall be sufficient to prove:
 - (a) That the name of the Member, in respect of whose Shares the money is sought to be recovered, appears entered in the Register of Members as the holder, at or subsequent to the date at which the money sought to be recovered is alleged to have become due, on the Shares in respect of which such money is sought to be recovered;
 - (b) that the resolution making the call is duly recorded in the minutes book; and
 - (c) that the notice of such call was duly given to the Member or his representatives issued in pursuance of these Articles;

and it shall not be necessary to prove the appointment of the Directors who made such call nor that a quorum of Directors was present at the meeting of the Board at which any call was made was duly convened or constituted nor any other matter whatsoever, but the proof of the matter aforesaid shall be conclusive evidence of the debt.

45. (a) (i) The Board may, if it thinks fit, agree to receive from Members willing to advance the same, all or any part of the amounts of their respective Shares beyond the sums actually called up, and upon the monies so paid in advance, or upon so much thereof, from time to time, and at any time thereafter, as exceeds the amount of the calls then made upon and due in respect of the Shares on account of which such advances are made, the Board of Directors may pay or allow interest, at such rate as the Members paying the sum in advance and the Board of Directors agree upon.

Provided that any amount paid up in advance of calls on any Shares may carry interest but shall not in respect thereof confer a right to Dividends or to participate in profits.

- (ii) The Board of Directors may agree to, at any time, repay the same upon giving to the Member three months' notice in writing.
- (b) No Member paying any such sum in advance shall be entitled to voting rights in respect

of the monies so paid by him until the same would but for such payment, become presently payable.

- 46. If a sum called in respect of a Share is not paid before or on the day appointed for payment thereof, the Person from whom the sum is due shall pay interest thereon from the day appointed for payment thereof to the time of the actual payment at ten per cent per annum or at such lower rate, if any, as the Board may determine.
- 47. The provision of these Articles shall *mutatis mutandis* apply to the calls on any other securities including Debentures of the Company.
- 48. (a) The Company shall have a first and paramount lien upon all Shares and / or Debentures (other than fully paid-up Shares / Debentures) registered in the name of each Member whether solely or jointly with others and upon the proceeds of sale thereof, for all monies including his debts, liabilities and engagements solely or jointly with any other Person to or with the Company (whether presently payable or not) called or payable at a fixed time in respect of such Shares and no equitable interest in any such Share shall be created except upon the footing and condition this Article will have full legal effect. Provided that the Directors may at any time declare any Shares / Debentures wholly or in part to be exempt from the provisions of this Article 48(a).
 - (b) The Company's lien on a Share shall extend to all Dividends or interest, as the case may be, payable and bonuses declared from time to time in respect of such Shares for any money owed to the Company.
 - (c) Unless otherwise agreed, the registration of transfer of Shares shall operate as a waiver of the Company's lien, if any, on such Shares.
- 49. (a) For the purpose of enforcing such lien, the Board may sell the Shares subject thereto in such manner as it shall think fit, and for that purpose may cause to be issued a duplicate certificate in respect of such Shares and / or debentures and may authorize one of their number to execute a transfer thereof on behalf of and in the name of such Member / Debenture-holder.

Provided that no such sale shall be made:

- (i) unless a sum in respect of which the lien exists is presently payable, or
- (ii) until the expiration of fourteen days after a notice in writing of the intention to sell shall have been served on such Member, his executors or administrators or his committee, curator bonis or other legal representatives as the case may be and default shall have been made by him or them in the payment of the sum payable as aforesaid.
- (b) To give effect to any such sale, the Board may authorize any Person to transfer the Shares sold to the purchaser thereof.
- (c) The purchaser shall be registered as the holder of the Shares comprised in any such transfer.
- (d) The purchaser shall not be bound to see to the application of the purchase money, nor shall his title to the Shares be affected by any irregularity or invalidity in the proceedings in reference to the sale.

- 50. (a) The net proceeds of any such sale shall be received by the Company and applied in or towards payment of such part of the amount in respect of which the lien exists, as is presently payable and the residue, if any, shall be payable to such members his executors or administrators or assigns or his committee, curator bonis or other legal representatives as the case may be, who are entitled to the Shares on the date of the sale.
 - (b) The receipt by the Company of consideration given for the Share on sale thereof (subject, if necessary, to the execution of an instrument of transfer) constitute a good title to the Share and the purchaser shall be registered as the holder of the Share. The Company shall be entitled to treat the registered holder of any Share or Debenture as the absolute owner thereof and accordingly shall not (except as ordered by a Court of competent jurisdiction or by statute required) be bound to recognize equitable or other claim to, or interest in, such Shares or Debentures on the part of any other Person. The Company's lien shall prevail notwithstanding that it has received notice of any such claims.
- 51. The provisions of these Articles relating to lien shall *mutatis mutandis* apply to any other securities including Debentures of the Company.

FORFEITURE

- 52. If any Member or Debenture-holder fails to pay any call, or instalment of a call, the day appointed for the payment thereof, the Board may, at any time thereafter during such time as the call or instalment remains unpaid, serve notice to him requiring payment of so much of the call or instalment as is unpaid, together with any interest thereon which may have accrued.
- 53. The notice aforesaid shall:
 - (a) name a further day (not being earlier than the expiry of fourteen days from the date of service of the notice) on or before which the payment required by the notice is to be made; and
 - (b) state that, in the event of non-payment on or before the day so named, the Shares or other securities in respect of which the call was made shall be liable to be forfeited.

if the requirements of such notice as aforesaid are not complied with, any Share or other security in respect of which the notice has been given may, at any time thereafter, before the payment required by the notice has been made, be forfeited by a resolution of the Board to that effect.

- 54. (a) A forfeited Share may be sold or otherwise disposed of on such terms and in such manner as the Board thinks fit.
 - (b) At any time before a sale or disposal as aforesaid, the Board may cancel the forfeiture on such terms as it thinks fit.
- A person whose Shares have been forfeited shall cease to be a Member in respect of the forfeited Shares, but shall, notwithstanding the forfeiture, remain liable to pay the Company all monies which, at the date of forfeiture, were presently payable by him to the Company in respect of the Shares.

- (b) The liability of such Person shall cease if and when the Company shall have received payment in full of such monies in respect of the Shares.
- 56. (a) A duly verified declaration in writing that the declarant is a Director, the manager or Secretary of the Company, and that Share in the Company has been duly forfeited on a date stated in the declaration, shall be conclusive evidence of the facts therein stated as against all Persons claiming to be entitled to the Share.
 - (b) The Company shall receive the consideration, if any, given for the Share on any sale or disposal thereof and may execute a transfer of the Share in favour of the person to whom the Share is sold or disposed of.
 - (c) The transferee shall thereupon be registered as the holder of the Share.
 - (d) The transferee shall not be bound to see to the application of the purchase money, if any, nor shall his title to the Share be affected by any irregularity or invalidity in the proceedings in reference to the forfeiture, sale or disposal of the Share.
- 57. The provisions of these Articles as to forfeiture shall apply in the case of non-payment of any sum which, by the terms of issue of a Share, becomes payable at a fixed time, whether on account of the nominal value of the Share or by way of premium, as if the same had been payable by virtue of a call duly made and notified.
- 58. The provisions of these Articles as to forfeiture shall *mutatis mutandis* apply to any other securities including Debentures of the Company.

TRANSFER AND TRANSMISSION OF SHARES AND DEBENTURES

- 59. The instrument of transfer of any Share in the Company shall be executed by or on behalf of the transferor and the transferee.
- 60. The transferor shall be deemed to remain a holder of the Share until the name of the transferee is entered in the Register of Members in respect thereof.
- 61. The Board may, subject to the right of appeal conferred by Section 58 of the Act and other provisions of the Act, decline to register:
 - (a) the transfer of a Share, not being a fully paid Share, to a Person of whom they do not approve; or
 - (b) any transfer of Shares on which the Company has a lien.
- 62. the Board may decline to recognize any instrument of transfer unless:
 - (a) the instrument of transfer is in the form as prescribed in the Rules made under subsection (1) of Section 56 of the Act and approved by the Stock Exchange;
 - (b) the instrument of transfer is accompanied by the certificate of the Shares to which it relates, and such other evidence as the Board may reasonable require to show the right of the transferor to make the transfer; and

- (c) the instrument of transfer is in respect of only one class of Shares.
- 63. The Company shall verify the signatures of the Members on instruments of transfer, letters of allotment, instruments of split or consolidation or renewal of Shares when so required by the Members, or by a member of the Stock Exchange or by the clearing house of the Stock Exchange.
- 64. The Board shall not decline to register or acknowledge any transfer of Shares on the ground that the transferor holding the Shares, either singularly or jointly with another Person, is indebted to the Company on any account whatsoever and the Board shall issue certificates within fifteen days of the lodgement for transfer.
- On giving not less than seven days' previous notice in accordance with Section 91 and the Rules prescribed thereunder, the registration of transfers may be suspended at such times and for such periods as the Board may from time to time determine; provided that the registration shall not be suspended for more than thirty days at any one time and there shall be a gap of at least thirty days between two book closures; provided that the Company shall give the Stock Exchange notice of its intention to close its transfer books for the purpose of settlement of transactions and shall not close its transfer books on such days as may be inconvenient to the Stock Exchange.
- 66. The Board shall, in order to expedite the process of transfer of Shares, delegate the power of Share transfer to an officer or a Committee or to the Registrar and share transfer agents. The delegated authority shall attend to the share transfer formalities at least once in every fifteen days.
- 67. The Company shall close its transfer books at least once a year at the time of the Annual General Meeting, if they have not been otherwise closed at any time during the year, and shall give to the Stock Exchange notice in advance of at least seven working days or such other number of days as the Stock Exchange may prescribe from time to time, stating the date of closure and the purpose of closure of the transfer books, and shall send copies of such notices to the other recognized Stock Exchange in India. The Company shall further close its transfer books for purposes of declaration of Dividend, or issue of rights or bonus Shares or Shares for conversion of Debentures or Shares arising out of rights attached to the Debentures or for such other purposes as the Stock Exchange may require from time to time.
- 68. (a) On the death of a Member, the survivor or survivors where the Member was a joint holder, and his nominee or nominees or legal representatives where he was a sole holder, shall be the only Persons recognized by the Company as having any title to his interest in the Shares.
 - (b) Nothing in Article 68(a) shall release the estate of a deceased joint holder from any liability in respect of any Share which had been jointly held by him with other Persons.
- 69. (a) Any Person becoming entitled to a Share in consequence of the death or insolvency of a Member may, upon such evidence being produced as may from time to time properly be required by the Board and subject as hereinafter provided, elect, either:
 - (i) to be registered himself as the holder of the Share; or

- (ii) to make such transfer of the Share as the deceased or insolvent Member could have made.
- (b) The Board shall, in either case, have the same right to decline or suspend registration as it would have had, if the deceased or insolvent Member had transferred the Share before his death or insolvency.
- 70. (a) If the Person so becoming entitled shall elect to be registered as the holder of the Share himself, he shall deliver or send to the Company a notice in writing signed by him stating that he so elects.
 - (b) If the Person aforesaid shall elect to transfer the Share, he shall testify his election by executing a transfer of the Share.
 - (c) All the limitations, restrictions and provisions of these Articles relating to the right to transfer and the registration of transfers of Shares shall be applicable to any such notice or transfer as aforesaid as if the death or insolvency of the Member had not occurred and the notice or transfer were a transfer signed by that Member.
- 71. A Person becoming entitled to a Share by reason or insolvency of the holder shall be entitled to the same dividends and other advantages to which he would be entitled if he were the registered holder of the Share, except that he shall not, before being registered as a Member in respect of the Share, be entitled in respect of it to exercise any right conferred by membership in relation to meetings of the Company; provided that the Board may, at any time, give notice requiring any such Person to elect either to be registered himself or to transfer the Share, and if the notice is not complied with within ninety days.
- 72. Subject to the provisions of Sections 58 and 59 of the Act, section 22A of the Securities Contracts (Regulation) Act, 1956, these Articles and subject to the provisions of any other law for the time being in force, the Board shall not decline to register or acknowledge any transfer of Shares on the ground that the transferor holding the Shares, either singularly or jointly with another Person, is indebted to the Company on any account whatsoever.
- 73. The Company shall keep a book, to be called the "Register of Transfers and Transmissions", and therein shall be fairly and distinctly entered the particulars of every transfer and transmission of Shares.
- 74. The provisions of these Articles shall *mutatis mutandis* apply to the transfer or transmission by operation of law, of any other securities including Debentures of the Company.

74A#@

Article 74A was inserted vide special resolution passed by the members at the Annual General Meeting held on 30 June 2017

Provided that the foregoing shall not be applicable to any promoter, member of the promoter group or person acting in concert with any promoter or member of the promoter group of the Company.

[@] Article 74A deleted pursuant to special resolution passed by the members at the Annual General Meeting held on 30 September 2020

⁷⁴A. Until such time as the Company remains a promoter of Aditya Birla Idea Payments Bank Limited, no person / group of persons shall acquire any shares of the Company which would take his / her / its holding to a level of 5% or more (or any such percentage imposed by Reserve Bank of India from time to time) of the total issued capital of the Company unless prior approval of the Reserve Bank of India has been obtained by such person / group of persons.

CONVERSION OF SHARES INTO STOCK AND RECONVERSION

- 75. The Company may, by an Ordinary Resolution passed at a General Meeting convert any fully paid up Shares into stock and reconvert that stock into fully paid up Shares of any denomination where any Shares have been so converted into stock, the several holders of stock may henceforth transfer their respective interests therein or any part of such interests in the same manner as and subject to the same regulations under which, the Shares from which the stock arose might, before the conversion, have been transferred, or as near thereto as circumstances admit. Provided that, the Board may, from time to time, fix the minimum amount of stock transferable, so, however, that such minimum shall not exceed the nominal amount of the Shares from which the stock arose.
- 76. The holders of stock shall, according to the amount of stock held by them, have the same rights, privileges and advantages as regards Dividends, voting at the Meetings of the Company, and other matters, as if they held the Shares from which the stock arose, but no such privilege or advantage (except participation in the Dividends and profits of the Company and in the assets on winding up) shall be conferred by an amount of stock which would not, if existing in Shares, have conferred that privilege or advantage. These Articles as are applicable to paid up Shares shall apply to stock and the words "Share" and "Shareholder" in those Articles shall include "stock" and "stockholder", respectively. JOINT HOLDERS
- 77. Where two or more Persons are registered as the holders of any Share / Debentures, they shall be deemed (so far as the Company is concerned) to hold the same as joint tenants with benefits of survivorship, subject to the following and other provisions contained in these Articles.
 - (a) The Company shall be entitled to decline to register more than six Persons as the joint holders of any Shares / Debentures.
 - (b) The joint holders of any Share / Debenture shall be liable severally as well as jointly for and in respect of all calls and other payments, which ought to be made in respect of such Shares / Debentures.
 - (c) In the case of a transfer of Shares / Debentures held by joint holders, the transfer will be effective only if it is made by all the joint holders.
 - (d) On the death of any one or more such joint holders the survivor or survivors shall be the only Person or Persons recognized by the Company as having any title to the Share / Debenture, but the Directors may require such evidence of death as they may deem fit, and nothing herein contained shall be taken to release the estate of a deceased joint holder from any liability on Shares / Debentures held by him jointly with any other Person.
 - (e) Any one of such joint holders may give effectual receipts of any Dividends, interests or other monies payable in respect of such Share / Debenture.
 - (f) Only the Person whose name stands first in the Register of Members / Register of Debenture-holders as one of the joint holders of any Shares / Debentures shall be entitled to the delivery of the certificate relating to such Share / Debenture or to receive notice.
 - (g) (i) Any one of two or more joint holders may vote at any meeting either personally or by attorney or by proxy in respect of such Shares as if he were solely entitled thereto and if more than one of such joint holders be present at any meeting personally or by proxy or by attorney then that one of such Persons so present

whose name stands first or higher (as the case may be) on the Register in respect of such Share shall alone be entitled to vote in respect thereof but the other or others of the joint holders shall be entitled to be present at the meeting provided always that a joint holder present at any meeting personally shall be entitled to vote in preference to a joint holder presently by attorney or by proxy although the name of such joint holder present by an attorney or proxy stands first or higher (as the case may be) in the Register in respect of such Shares.

(ii) Several executors or administrators of a deceased Member in whose (deceased Member) sole name any Share stands shall for the purpose of this Article 77(g)(ii) be deemed joint holders.

COPIES OF MEMORANDUM AND ARTICLES, ETC. TO BE GIVEN TO MEMBERS

78. Copies of the Memorandum and Articles of Association of the Company and other documents referred in Section 117 of the Act shall be sent by the Company to every Member at his request on payment of such fees as may be fixed by the Board for each copy provided that such fees shall not exceed the maximum fees agreed between the Company and the Stock Exchange.

POWERS OF DIRECTORS

- 79. The Board of Directors shall not, except with the consent of the Company in General Meeting accorded by a Special Resolution:
 - (a) Sell, lease or otherwise dispose of the whole or substantially the whole of the undertaking of the Company or where the Company owns more than one undertaking, of the whole or substantially the whole of any such undertakings.
 - (b) Invest otherwise in trust securities the amount of compensation received by the Company as a result of any merger or amalgamation.
 - (c) Borrow money, where the money to be borrowed, together with the money already borrowed by the Company will exceed aggregate of the Company's paid-up Capital and free reserves, apart from temporary loans obtained from the Company's bankers in the ordinary course of business.
 - (d) Remit, or give time for the repayment of, any debt due from a Director.
 - (e) Dispose of shares in a material subsidiary of the Company which would reduce its shareholding (either on its own or together with other subsidiaries) to less than fifty percent or cease to exercise control over the subsidiary.
 - (f) Sell, dispose of or lease assets amounting to more than twenty percent of the assets of a material subsidiary.

Explanation 1:- Every resolution passed by the Company in a General Meeting in relation to the exercise of the power referred to in sub-Article (c) of this Article 79 shall specify the total amount up to which money may be borrowed by the Board of Directors under sub-Article (c).

Explanation 2:- 'Material subsidiary' shall mean a subsidiary of the Company which the Company has disclosed to the Stock Exchange and in its annual report as being material.

80. The Board of Directors may raise and secure the payment of such sum or sums in such manner and upon such terms and conditions in all respects as they think fit, and in particular by the issue of bonds, perpetual or redeemable, Debentures or debenture- stocks or any mortgage or

charge or other security on the undertaking of the whole or any part of the property of the Company (both present and future) including its uncalled Capital for the time being.

- 81. Any bonds, Debentures, debenture-stocks or other securities issued or to be issued by the Company shall be under the control of the Board of Directors who may issue them upon such terms and conditions and in such manner and for such consideration as they shall consider to be for the benefit of the Company and in accordance with applicable provisions of the Act and Rules. Provided that bonds, Debentures, debenture-stock or other securities so issued or to be issued by the Company with the right to allotment of or conversion into Shares shall not be issued except with the sanction of the Company in General Meeting.
- 82. Debentures, debenture-stocks, bonds or other securities may be made assignable free from any equities between the Company and the Person to whom the same may be issued.
- 83. Any bonds, Debentures, debenture-stocks or other securities may be issued, subject to the provisions of the Act, and these Articles, at a discount, premium or otherwise and with any special privileges as to redemption, surrender, drawings, appoint of Directors and otherwise and subject to the following:
 - (a) The Company shall not issue any Debentures carrying voting rights at any Meeting of the Company whether generally or in respect of particular classes of business.
 - (b) Payments of certain debts out of assets subject to floating charge in priority to claims under the charge may be made in accordance with the provisions of the Act.
 - (c) Certain charges shall be void against the liquidators or creditors unless registered as provided in Section 77 of the Act and the Rules.
 - (d) The term 'charge' shall include mortgage in these Articles.
 - (e) A contract with the Company to take up and pay for any Debentures of the Company may be enforced by a decree for specific performance.
 - (f) The Company shall, within six months after the allotment of any of its Debentures and within sixty days after the application for the registration of the transfer of any such Debentures or debenture-stocks have complete and have ready for delivery the certificate of all the Debentures and the certificates of all debenture-stocks allotted or transferred unless the conditions of issue of the Debentures or debenture-stocks otherwise provide. The expression transfer for the purpose of this Article 88(f) means a transfer duly stamped and otherwise valid and does not include any transfer which the Company is for any reason entitled to refuse to register and does not register.
 - (g) (i) A copy of the trust deed for securing any issue of Debentures shall be open for inspection to any Member or Debenture holder of the Company, in the same manner, to the same extent and on the payment of the same fees, as if it were the Register of Members of the Company; and
 - (ii) A copy of the trust deed shall be forwarded to any Member or Debenture holder of the Company, at the request, within seven days of the making thereof, on the payment of such fee as may be determined by the Board; provided that such fees shall not exceed the fees agreed between the Company and the Stock Exchange.

- 84. If any uncalled Capital of the Company is included in or charged by any mortgage or other security the Directors shall, subject to the provisions of the Act and these Articles, make calls on the Members in respect of such uncalled Capital in trust for the Person in whose favour such mortgage or security is executed.
- 85. If the Directors or any of them or any other Person shall become personally liable for the payment of any sum primarily due from the Company, the Directors may execute or cause to be executed any mortgage, charge or security over or affecting the whole or any part of the assets of the Company by way of indemnity to secure the Directors or Person so becoming liable as aforesaid from any loss in respect of such liability.
- 86. (a) The provisions of the Act relating to registration of charges shall be complied with.
 - (b) In the case of a charge created out of India and comprising solely property situated outside India, the provisions of Section 77 of the Act shall also be complied with.
 - (c) Where a charge is created in India but comprises property outside India, the instrument creating or purporting to create the charge under Section 77 of the Act and the Rules or a copy thereof verified in the prescribed manner, may be filed for registration, notwithstanding that further proceedings may be necessary to make the charge valid or effectual according to the law of the country in which the property is situate, as provided by Section 77 of the Act and the Rules.
 - (d) Where any charge on any property of the Company required to be registered under Section 77 of the Act has been so registered, any Person acquiring such property or any part thereof or any share or interest therein shall be deemed to have notice of the charge as from the date of such registration.
 - (e) In respect of registration of charges on properties acquired subject to charge, the provisions of Section 79 of the Act shall be complied with.
 - (f) The Company shall comply with the provisions of the Act relating to particulars in case of Debentures entitling holders *pari passu*.
 - (g) The Company shall comply with the provisions of the Act in regard to registration of particulars of commission, allowance or discount paid or made, directly or indirectly, in connection with the debentures.
 - (h) The Company shall comply with the provisions of Section 78 of the Act as regards registration of particulars of every charge and of every series of debentures.
 - (i) As to modification of charges, the Company shall comply with the provisions of Section 79 of the Act.
 - (j) The Company shall comply with the provisions of Section 85 of the Act regarding keeping a copy of instrument creating charge at the Registered Office of the Company and comply with the provisions of Section 84 of the Act in regard to entering in the register of charges any appointment of receiver or manager as therein provided.
 - (k) The Company shall also comply with the provisions of Section 82 of the Act as to reporting satisfaction of any charge and procedure thereafter.
 - (l) The Company shall keep at its registered office a register of charges in the prescribed form and enter therein all charges specifically affecting any property of the Company

- and all floating charges on the undertaking or on any property of the Company.
- (m) Any creditor or Member of the Company and any other Person shall have the right to inspect copies of instruments creating charges and the Company's Register of Charges in accordance with and subject to the provisions of Section 85 of the Act and the Rules.
- 87. No notice of any trust, express or implied or constructive, shall be entered on the Register of Debenture-holders.

MEETINGS OF MEMBERS

- 88. (a) The Company shall in each year hold a General Meeting as its Annual General Meeting in addition to any other Meeting in that year.
 - (b) All General Meetings other than Annual General Meetings shall be called Extraordinary General Meetings.
 - (c) Nothing contained in the foregoing provisions shall be construed as affecting the right conferred upon the Registrar under the provisions of Section 96 of the Act to extend the time within which any Annual General Meeting may be held.
 - (d) Every Annual General Meeting shall be called for a time during business hours on a day that is not a public holiday, and shall be held either at the Registered Office of the Company or at some other place within the city or town in which the Registered Office of the Company is, for the time being, situated as the Board may determine and the notice calling the Meeting shall specify it as the Annual General Meeting.
 - (e) Every Member of the Company shall be entitled to attend every General Meeting either in Person or by proxy and the Auditor of the Company shall have the right to attend and to be heard at any General Meeting which he attends on any part of the business which concerns him as such Auditor. No Person shall be permitted to act as a proxy for more than fifty Members or in respect of Shares aggregating more than ten per cent of the total Capital of the Company carrying voting rights. The proxy register with proxies and the Register of Directors' shareholdings shall remain open and accessible during the Meeting.
 - (f) At every Annual General Meeting, there shall be laid on the table the directors' report and audited statement of accounts and the Auditors' Report (if not already incorporated in the audited statement of accounts).
 - (g) A resolution of a General Meeting of the Shareholders shall be adopted by a simple majority of the Shares of Equity Capital entitled to vote at such Meeting (whether by show of hands, by poll or through electronic voting), unless a greater percentage is required by applicable law.
- 89. The Board shall cause to be prepared the annual return and balance sheet and profit and loss account and the consolidated financial statements required to be prepared under Section 129(3) of the Act and Article 212, and forward the same to the Registrar, in accordance with Sections 92 and 137 of the Act.
- 90. Section 98, Sections 101 to 107 and Section 109 of the Act with such adaptations and modifications, if any, as may be prescribed, shall apply with respect to Meetings of any class of Members or Debenture-holders of the Company in like manner as they apply with respect to General Meetings of the Company.

- 91. The Board may, whenever it thinks fit, call an Extra-ordinary General Meeting. The Board shall at the requisition made by such number of Members who hold, on the date of the receipt of the requisition, not less than one-tenth of such of the paid-up Capital of the Company as on that date carries the right of voting.
- 92. If at any time Directors capable of acting who are sufficient in number to form a quorum are not within India, any director or any two members of the company may call an Extraordinary General Meeting in the same manner, as nearly as possible, as that in which such a meeting may be called by the Board.
- 93. Any requisition so made by Members shall set out the matter or matters for the consideration of which the Meeting is proposed, shall be signed by the requisitionists, and shall be deposited at the Office; provided that such requisition may consist of several documents in like form each signed by one or more requisitionists.
- 94. If the Board does not, within twenty one days of the date of receipt of a valid requisition in regard to any matter, proceed to call a Meeting for the consideration of that matter on a day not later than forty five days from the date of receipt of such requisition, the Meeting may be called and held by the requisitionists themselves within a period of three months from the date of the requisition.
- 95. Any Meeting called under the foregoing Articles by the requisitionists shall be called in the same manner, as nearly as possible, as that in which Meeting are to be called by the Board.
- 96. (a) Twenty one days' notice of every General Meeting, Annual or Extra ordinary, and by whomsoever called, specifying the day, place and hour of the Meeting, and the general nature of the business to be transacted thereat, shall be given in the manner hereinafter provided, to such Persons as are under these Articles entitled to receive notice from the Company. Provided that, with the consent of the Members holding not less than ninety five per cent of such part of the paid up Capital of the Company as gives a right to vote at the Meeting, a Meeting may be convened by a shorter notice.
 - (b) In the case of an Annual General Meeting, any business other than (i) the consideration of the accounts, balance sheet and reports of the Board of Directors and Auditors, (ii) the declaration of a Dividend, (iii) the appointment of Directors in the place of those retiring, and (iv) the appointment of, and the fixing of the remuneration of, the Auditors, is to be transacted, and in the case of any other Meeting all business, shall be special, and there shall be annexed to the notice of the Meeting a statement setting out all material facts concerning each such item of special business, including, in particular, the nature of the concern or interest, if any, therein of every Director and Key Managerial Personnel, if any, and their Relatives.
 - (c) Where any such item of business relates to, or affects any other Company, the extent of shareholding interest in that other Company of every Director and the Manager, if any, of the Company shall also be set out in such statement if the extent of such shareholding interest is not less than twenty per cent of the paid up Capital of that other Company.
 - (d) Where any item of business consists of the according of approval to any document by the Meeting, the time and place where the document can be inspected shall be specified in the statement aforesaid.

- 97. The accidental omission to give any such notice to, of the non-receipt of notice by any Member or other Person to whom it should be given shall not invalidate any proceedings at the Meeting.
- 98. No General Meeting, Annual or Extra-ordinary, shall be competent to enter upon, discuss or transact any business, the general nature of which has not been mentioned in the notice upon which it was convened.
- 99. (a) Subject to the provisions of the Act, five Members present in Person shall be a quorum for a General Meeting.
 - (b) A body corporate, being a Member, shall be deemed to be personally present if represented in accordance with Section 113 of the Act.
- 100. (a) If the quorum is not present within half an hour from the time appointed for holding a General Meeting:
 - (i) the Meeting shall stand adjourned to the same day in the next week at the same time and place, or to such other date and such other time and place as the Board may determine; provided that in the case of an adjourned Meeting or of a change of day, time or place of meeting, the Company shall give not less than three days' notice to the Members either individually or by publishing an advertisement in the newspapers (one in English and one in vernacular language) which is in circulation at the place where the Office is situated; or
 - (ii) the Meeting if called by requisitionists under Article 91, shall stand cancelled.
 - (b) If at the adjourned Meeting also, a quorum is not present within half an hour from the time appointed for holding the Meeting, the Members present shall be the quorum.
- 101. (a) Subject to the provisions of these Articles, the Chairman of the Board of Directors shall preside as Chairman at every General Meeting, whether Annual or Extra-ordinary.
 - (b) If, at any Meeting the Chairman shall not be present within fifteen minutes of the time appointed for holding such Meeting, or shall decline to take the chair, then the Members present shall elect any other Director as Chairman, and if no Director be present or if all the Directors present at the Meeting decline to take the chair, then the Members present shall elect one of their number to be Chairman.
- 102. No business shall be discussed at any General Meeting except the election of Chairman, whilst the chair is vacant.
- 103. The Chairman may, with the consent of any Meeting at which a quorum is present, and shall, if so directed by the Meeting, adjourn the Meeting from time to time and from place to place.
- 104. No business shall be transacted at any adjourned Meeting other than the business left unfinished at the Meeting from which the adjournment took place.
- 105. When a Meeting is adjourned for thirty days or more, notice of the adjourned Meeting shall be given as in the case of an original Meeting.

- 106. Save as aforesaid, and as provided in Section 103 of the Act, it shall not be necessary to give any notice of an adjournment or of the business to be transacted at an adjourned Meeting.
- 107. At any General Meeting, a resolution put to the vote of the Meeting shall, unless a poll is demanded under Section 109 of the Act or the voting is carried out electronically, be decided on a show of hands. declaration by the Chairman of the Meeting of the passing of a resolution or otherwise by show of hands under this Article 107 and an entry to that effect in the books containing the minutes of the Meeting of the Company shall be conclusive evidence of the fact of passing of such resolution or otherwise.
- 108. In the case of an equality of votes, whether on a show of hands or electronically or on a poll, the Chairman shall, have a casting vote.
- 109. (a) If a poll is demanded as aforesaid, the same shall be taken at such time (not later than forty eight hours from the time when the demand was made) and place in the city or town in which the Registered Office of the Company is, for the time being, situate and either by open voting or by ballot, as the Chairman shall direct, and either at once or after an interval or adjournment, or otherwise, and the result of the poll shall be deemed to be the resolution of the Meeting at which the poll was demanded.
 - (b) The demand for a poll may be withdrawn, at any time, by the Persons who made the demand.
- 110. A Member may exercise his vote at a Meeting by electronic means in accordance with Section 108 of the Act (and the agreement between the Company and the Stock Exchange) and shall vote only once.
- Where a poll is to be taken, the Chairman of the Meeting shall appoint such number of scrutineers as he, in his sole discretion, deems fit to scrutinize the votes given on the poll and to report thereon to him.
 - (b) The Chairman shall have power, at any time, before the result of the poll is declared, to remove a scrutineer from office and fill the vacancy in the office of a scrutineer arising from such removal or from any other cause.
- 112. Any poll duly demanded on the election of a Chairman of a Meeting or on any question of adjournment shall be taken at the Meeting itself and without adjournment.
- 113. The demand for a poll, except on the questions of the election of the Chairman, and of an adjournment, shall not prevent the continuance of a Meeting for the transaction of any business other than the question on which the poll has been demanded.

VOTE OF MEMBERS

114. No Member shall be entitled to vote either personally, by proxy or electronically for another Member, at any General Meeting or at any Meeting of a class of Shareholders, either upon a show of hands, or upon a poll, in respect of any Shares registered in his name on which any calls or other sums presently payable by him have not been paid or in regard to which the Company has, and has exercised, any right of lien.

- 115. A Member is not prohibited from exercising his voting on the ground that he has not held his Share or other interest in the Company for any specified prescribed period preceding the date on which the vote is taken, or on any other ground not being a ground set out in the Article 114.
- Subject to the provisions of these Articles, and without prejudice to any special privileges or restrictions as to voting, for the time being attached to any class of Shares for the time being, forming part of the Capital of the Company, every Member, not disqualified by Article 114, shall be entitled to be present, and to speak and vote at such Meeting.
 - (b) Subject to any rights or restrictions for the time being attached to any class or classes of Shares:
 - (i) on a show of hands, every Member present in person shall have one vote; and
 - (ii) on a poll, the voting rights of Members shall be in proportion to his share in the paid-up Capital of the Company.

Provided that if any preference share holder be present at any Meeting of the Company, save as provided in the second proviso to Section 47(2) of the Act, he shall have a right to vote only on resolutions placed before the Meeting which directly affect the rights attached to his preference Shares.

- 117. (a) A body corporate may, if it is a Member of the Company, by a resolution of its board of directors or other governing body, authorize such person as it thinks fit to act as its representative at any General Meeting, or at any meeting of any class of Members of the Company.
 - (b) A person authorized by a resolution under Article 117(a) shall be entitled to exercise the same rights and powers, including the right to vote by proxy and by postal ballot, on behalf of the body corporate which he represents as that body corporate could exercise if it were an individual Member of the Company.
- 118. If there be joint registered holders of any Shares, any one of such Persons may vote at any Meeting or may appoint another Person (whether a Member or not) as his proxy in respect of such Shares, as if he were solely entitled thereto and if more than one such joint holder be present at any Meeting either in person or by proxy, that one of the said Persons so present whose name stands higher on the Register of Members shall alone be entitled to speak and to vote in respect of such Shares, but the other or others of the joint holders shall be entitled to be present at the Meeting.
- 119. Subject to the provisions of the Act and in accordance with these Articles, any Person entitled under the Articles pertaining to transmission to any Shares may vote at any General Meeting in respect thereof as if he was the registered holder of such Shares, provided that at least forty eight hours before the time of holding the meeting or adjourned meeting, as the case may be, at which he proposes to vote, he shall duly satisfy the Board of his right to such Shares unless the Board shall have previously admitted his right to vote at such meeting in respect thereof.
- 120. (a) Subject to the provisions of these Articles, votes may be given by Members either in Person or by proxy.

- (b) Any Member of the Company entitled to attend and vote at a Meeting of the Company shall be entitled to appoint any other Member as his proxy to attend and vote instead of himself. A Member (and in case of joint holders, all holders) shall not appoint more than one Person as proxy.
- (c) In every notice called a meeting of the Company there shall appear with reasonable prominence a statement that a Member entitled to attend and vote is entitled to appoint a proxy to attend and vote instead of himself, and that proxy need not be a Member.
- 121. (a) The instrument appointing a proxy shall be in the form as specified in the Rules made under Section 105 of the Act and shall:
 - (i) be in writing; and
 - (ii) be signed by the appointee or his attorney duly authorised in writing or, if the appointee is a body corporate, be under its Seal, if any or be signed by an officer or an attorney duly authorised by it.
 - (b) The proxy so appointed shall not have any right to speak at the Meeting.
- 122. No Member present only by proxy shall be entitled to vote on a show of hands. The representative of a body corporate appointed in terms of Section 113 of the Act, however, shall have a vote on a show of hands.
- 123. (a) The President of India or the Governor of a State if he is a Member of the Company may appoint such Person as he thinks fit to act as his representative at any meeting of the Company or at any Meeting of any class of Members of the Company in accordance with provisions of Section 112 of the Act or any other statutory provision governing the same.
 - (b) A Person appointed to act as aforesaid shall for the purposes of the Act be deemed to be a Member of such a Company and shall be entitled to exercise the same rights and powers (including the right to, vote by proxy) as the President or the Governor, as the case may be, could exercise, as a Member of the Company.
- 124. The instrument appointing a proxy and the power of attorney or other authority, if any, under which it is signed, or a notarially certified copy of that power of authority, shall be deposited at the office not less than forty eight hours before the time for holding the Meeting or adjourned Meeting at which the Person named in the instrument proposes to vote, or in the case of a poll, not less than twenty four hours before the time appointed for the taking of the poll, and in default, the instrument of proxy shall not be treated as valid.
- 125. Every instrument of proxy shall be in the form specified in the Rules and any other Rules made under Section 105 to the Act, or in a form as near thereto as circumstances admit.
- 126. A vote given in accordance with the terms of an instrument of proxy shall be valid notwithstanding the previous death, insanity, winding up of the principal, or revocation of the proxy or of any power of attorney under which such proxy was signed, or the transfer of the share in respect of which the vote is given. Provided that no intimation in writing of such death, insanity, winding up, revocation or transfer shall have been received at the Office before the commencement of the Meeting or adjourned Meeting at which the proxy is used.

- 127. No objection shall be raised to the qualification of the voter or to the validity of any vote, except at the Meeting or at the adjourned Meeting or on a poll at which such vote shall be given or tendered, and every vote whether given personally or by proxy, not disallowed at such Meeting or adjourned Meeting or poll shall be deemed valid for all purposes of such Meeting or poll whatsoever. Provided that, any such objection raised in due time shall be referred to the Chairman of the Meeting, whose decision shall be final and conclusive.
- 128. Member of unsound mind, or in respect of whom an order has been made by any Court having jurisdiction in lunacy, may vote, whether on a show of hands or on a poll, by his committee or other legal guardian, and any such committee or guardian may, on a poll, vote by proxy.
- 129. The Chairman of any Meeting shall be the sole judge of the validity of every vote given or tendered at such Meeting. The Chairman present at the time of taking of a poll shall be the sole judge of the validity of every vote tendered at such poll.
- 130. (a) The Company shall cause minutes of all proceedings of every meeting of every class of Shareholders or creditors' meeting and every resolution passed by postal ballot to be kept by making entries thereof in books kept for that purpose with their pages consecutively numbered within thirty days of the conclusion of every such meeting concerned or passing of resolution by postal ballot.
 - (b) Each page of every such book shall be initialled or signed and the last page of the record of proceedings of each Meeting in such book shall be dated and signed by the Chairman of the same Meeting within the aforesaid period of thirty days or in the event of the death or inability of that Chairman within this period, by a Director duly authorised by the Board for the purpose.
 - (c) The minutes of each Meeting shall contain a fair and correct summary of the proceedings thereat.
 - (d) All appointments of officers made at any of the Meetings aforesaid shall be included in the minutes of the Meeting.
 - (e) (i) Nothing herein contained shall require or be deemed to require the inclusion in any such minutes of any matter which in the opinion of the Chairman of the Meeting:
 - (1) is, or could reasonably be regarded as defamatory of any Person;
 - (2) is irrelevant or immaterial to the proceedings; or
 - (3) is detrimental to the interest of the Company.
 - (ii) The Chairman of the Meeting shall exercise an absolute discretion in regard to the inclusion or non-inclusion of any matter in the minutes on the aforesaid grounds.
 - (f) Any such minutes shall be kept in accordance with the provisions of the Act and shall be evidence of the proceedings recorded therein.
 - (g) (i) The books containing the minutes of the proceedings of any General Meeting shall be kept at the Office and shall be open, during business hours, for a period of two hours in the aggregate in each day, to the inspection of any Member without charge.

- (ii) Any Member shall be entitled to be furnished, within the time prescribed by the Act, after he has made a request in writing in that behalf to the Company and on payment of such fees as may be fixed by the Board (which shall in no event exceed the fees agreed by the Company and the Stock Exchange), with a copy of any minutes referred to in Article 130(a).
- 131. The Board, and any Person(s) authorised by it, may take any action before the commencement of any General Meeting, or any Meeting of a class of Members in the Company, which they may think fit to ensure the security of the Meeting, the safety of the people attending the Meeting, and the future orderly conduct of the Meeting. Any decision made in good faith under this Article shall be final, and rights to attend and participate in the Meeting concerned shall be subject to such decision. DIRECTORS
- 132. The persons hereinafter named shall be the first Directors, that is to say:
 - (a) Shri Aditya Vikram Birla;
 - (b) Shri Kumar Mangalam Birla; and
 - (c) Shri Mahesh Chandra Bagrodia.
- 133. (a) Until otherwise determined by a General Meeting and subject to Section 149 of the Act, and the provisions of these Articles, the number of Directors shall not be less than three and not more than fifteen.
 - (b) The majority Directors on the Board shall be resident Indian citizens.
- 134. (a) The Company shall in general, subject to the provisions of the Act, be entitled hereafter to agree with the Central or any State Government, Person, firm or corporation or any financial or lending Institution, the he or it shall have right to appoint his or its nominee(s) on the Board of the Company, upon such terms and conditions mutually agreed on.
- 135. Subject to the provisions of Section 152 of the Act, the number of Directors appointed under Article 134shall not exceed in the aggregate one-third of the total number of Directors for the time being in office.
- 136. (a) The Board of Directors shall be entitled to appoint an alternate Director to a Director who is not present in India for a period of not less than three months. No Person shall be appointed as an alternate Director for an independent Director unless he is qualified to be appointed as an independent Director under the provisions of the Act.
 - (b) An alternate Director appointed under this Article 136 shall vacate office if and when the original Director returns to such State in which meetings of the Board are ordinarily held.
 - (c) If the term of office of the original Director is determined before he so returns to that State, any provision in the Act or in these Articles for the automatic re-appointment of retiring Directors in default of another appointment shall apply to the original Director, and not to the alternate Director.

- (d) An alternate Director shall not hold office as such for a longer period than that permissible to the original Director in whose place he has been appointed.
- 137. (a) Subject to the provisions of Section 149 of the Act and the other applicable provisions of these Articles, the Board of Directors shall also have power any time and from time to time to appoint any person, other than a person who fails to get appointed as a Director in a General Meeting, as an additional Director, but so that the total number of Directors shall not, at any time exceed the maximum strength fixed for the Board by the Articles.
 - (b) Any Person so appointed as an additional Director shall remain in office only up to the date of the next Annual General Meeting or the last date on which the Annual General Meeting should have been held, whichever is earlier, but shall be eligible for the reappointment at such Meeting subject to the provisions of the Act.
- 138. If the office of any Director appointed by the Company in General Meeting is vacated before his term of office will expire in the normal course, the resulting casual vacancy may, in default of and subject to these Articles, be filled by the Board of Directors at a meeting of the Board. The Director so appointed shall hold office only up to the date which the Director in whose place he is appointed would have held office if it had not been vacated.
- 139. No Director shall be required to hold any Shares as qualification Shares.
- 140. (a) At a General Meeting of the Company a motion shall not be made for the appointment of two or more Persons as Directors by a single resolution, unless a resolution that it shall be so made has first been agreed to by the Meeting without any vote being given against it.
 - (b) A resolution moved in contravention of Article 140(a) shall be void, whether or not objection was taken at the time of its being so moved. Provided that where a resolution so moved is passed, no provision for the automatic re-appointment of the retiring Director in default of another appointment shall apply, as herein before provided.
 - (c) For the purpose of this Article 140, a motion for approving a Person's appointment, or for nominating a Person for appointment, shall be treated as a motion for his appointment.
- 141. (a) A Person who is not a retiring Director shall be eligible for appointment to the office of Director at any General Meeting, if he or some Member intending to propose him has, not less than fourteen days before the Meeting, left at the Registered Office a notice in writing under his hand signifying his candidature for the office of Director or the intention of such Member to propose him as a candidate for that office, as the case may be, to such member, if the Person succeeds in getting elected as a Director.
 - (b) The Company shall inform its Members of the candidature of a Person for the office of a Director or the intention of a Member to propose such Person as a candidate for that office, by serving individual notices on the Members not less than seven days before the Meeting. Provided that it shall not be necessary for the Company to serve individual notices upon the Members as aforesaid, if the Company advertises such candidature or intention, not less than seven days before the Meeting, in at least two newspapers circulating in the place where the Office of the Company is located, of which one is published in the English language and the other in the regional language of that place.

(c) Every Person (other than a Director retiring by rotation or otherwise or a Person who has left at the Office of the Company, a notice under Section 160 of the Act signifying his candidature for the office of Directors) proposed as a candidate for the office of a Director shall sign and file with the Company his consent in writing to act as a Director, if appointed.

142. A Person other than -

- (a) A Director re-appointed after retirement by rotation or immediately on the expiry of his term of office; or
- (b) An additional or alternate Director, or a Person filling a casual vacancy in the office of a Director under Section 161 of the Act, appointed as a Director or re-appointed as an additional or alternate Director, immediately on the expiry of his term of office; or
- (c) A person named as a Director of the Company under its Articles as first registered,

shall not act as a Director of the Company unless he has, within thirty days of his appointment, signed and filed with the Registrar his consent in writing to act as such Director.

- 143. (a) The fee payable to a Director for attending a meeting of the Board or Committee thereof or a General Meeting shall be decided by the Board of Directors from time to time and shall not exceed Rs. 1,00,000 (Rupees one lakh) per meeting or such other limit as may be prescribed by the Act. The remuneration payable to a Director shall, in so far as it consists of a monthly payment, be deemed to accrue from day to day.
 - (b) Subject to the provisions of the Act, the Directors may be paid such further or additional remuneration, if any, as the Company in General Meeting shall, from time to time, determine, and such additional or further remuneration shall be divided among the Directors in such proportion and manner as the Board may, from time to time, determine, and in default of such determination shall be divided equally among the Directors entitled to remuneration.
 - (c) Subject to the provisions of the Act, if any Director be called upon to perform extra services or special exertions or efforts (which expression shall include work done by a Director as a member of any Committee formed by the Directors), the Board may arrange with such Director for such special remuneration for such extra services or special exertions or efforts, either by a fixed sum or otherwise, as may be determined by the Board and such remuneration may be either in addition to or in substitution for his remuneration above provided.
- 144. The Board of Directors may allow and pay to any Director, who is not a resident of the place where the meetings of the Board or Committees thereof or General Meeting of the Company are held and who shall come to such place for the purpose of attending a meeting or for attending its business at the request of the Company, such sum as the Board may consider fair compensation for traveling, and other incidental expenses, in addition to his fee, if any, for attending such meeting as above specified, and if any Director be called upon to go or reside out of the ordinary place of his residence of the Company's business, he shall be entitled to be reimbursed all traveling and other expenses incurred in connection with the business of the Company.
- 145. The continuing Director(s) may act notwithstanding any vacancy in their body, but, if and so long as their number is reduced below the quorum fixed by these Articles for a meeting of the Board, the continuing Director(s) may act for the purpose of increasing the number of Directors to that fixed for the quorum, or for summoning a General Meeting, but for no other purpose.

- 146. Subject to Section 167 of the Act, the office of a Director shall become vacant, if:
 - (a) he is found to be of unsound mind by a Court of competent jurisdiction;
 - (b) he applies to be adjudicated as an insolvent and his application is pending;
 - (c) he is an undischarged insolvent;
 - (d) he fails to pay any call made on him in respect of Shares of the Company held by him whether alone or jointly with others, within six months from the last date fixed for the payment of the call, unless the Central Government has, by notification in the Official Gazette, removed the disqualification incurred by such failure;
 - (e) he absents himself from all meetings of the Board held during a period of twelve months with or without seeking leave of absence of the Board;
 - (f) he becomes disqualified by an order of court under the provisions of the Act;
 - (g) he is removed in pursuance of Section 169 of the Act;
 - (h) he fails to disclose his interest in any contract or arrangement in which he is directly or indirectly interest, in contravention of the provisions of Section 184 of the Act;
 - (i) he acts in contravention of Section 184 of the Act;
 - (j) he is convicted by a court of any offense involving moral turpitude and sentenced in respect thereof to imprisonment for not less than six months; provided that the office shall be vacated by the Director even if he has filed an appeal against the order of such court;
 - (k) having been appointed a Director by virtue of his holding any office or other employment in the Company, he ceases to hold such office or other employment in the Company.
- 147. (a) A Director of the Company, who is in any way, whether directly or indirectly, concerned or interested in a contract or arrangement entered into or a proposed contract or arrangement to be entered into by or on behalf of the Company, shall disclose the nature of his concern or interest at a meeting of the Board in the manner provided in Section 184 of the Act. Provided that it shall not be necessary for a Director to disclose his concern or interest in any contract or arrangement entered into with any other company where any of the Directors of the Company holds or two or more of them together hold, not more than two per cent of the paid up Capital in any such other company.
 - (b) (i) In the case of a proposed contract or arrangement, the disclosure required to be made by a Director under Article 147(a) shall be made at the meeting of the Board at which the question of entering into the contract or arrangement is first taken into consideration, or if the Director was not, at the date of that meeting concerned or interested in the proposed contract or arrangement, at the first meeting of the Board held after he becomes so concerned or interested;
 - (ii) In the case of any other contract or arrangement, the required disclosure shall be made at the first meeting of the Board held after the Director becomes concerned or interested in the contract or arrangement.

- (c) A Director shall give notice of his interests to the Company in the prescribed form at the first meeting of the Board of Directors in every financial year.
- 148. No Director shall as such interested Director, take any part in the discussion of, or vote on, any contract or arrangement entered into or to be entered into by or on behalf of the Company, if he is in any way, whether directly or indirectly, concerned or interested in such contract or arrangement, nor shall his presence count for the purpose of forming a quorum at the time of any such discussion or vote, and if he does vote, his vote shall be void. Provided however, that nothing herein contained shall apply to:
 - (a) any contract of indemnity against any loss which the Directors, or any one or more of them, may suffer by reason of becoming or being sureties or a surety for the Company;
 - (b) any contract or arrangement entered into or to be entered into with a public company or a private company which is a subsidiary of a public company in which the interest of the Director consists solely in his being a member holding not more than two per cent of its paid up Capital.
- 149. (a) The Company shall keep one or more Registers in accordance with Section 189 of the Act, and shall within the time specified therein, enter in such Register(s) the particulars of all contracts or arrangements to which Section 184 or Section 188 of the Act applies in the form prescribed by the Act and the Rules.
 - (b) The Registers shall be kept at the Office and shall be open to inspection at the Office and extracts may be taken there from and copies thereof may be required by any Member of the Company to the same extent, in the manner and on payment of the same fee as in the case of the Register of Members of the Company and the provisions of Section 94 of the Act shall apply accordingly.
- 150. Subject to the provisions of the Act and any other law for the time being in force, a Director may be or become a Director of any company promoted by the Company, or in which it may be interested as a vendor, Shareholder, or otherwise, and, subject to the provisions of the Act, no such Director shall be accountable for any benefits received as Director or shareholder of such other company.
- 151. (a) The Company shall keep at the Office a register containing the particulars of Directors, Managers, Secretaries, Key Managerial Personnel and other Persons mentioned in Section 170 of the Act and shall send to the Registrar, a return containing the particulars specified in such register and shall otherwise comply with the provisions of the said Section in all respects.
 - (b) The Company shall also keep at the Office a register in respect of the Shares or Debentures of the Company held by the Directors and Key Managerial Personnel as required by Section 170 of the Act, and shall otherwise duly comply with the provisions of the said Section in all respects.
- 152. (a) Every Director, Managing Director, Manager, Secretary or Key Managerial Personnel of the Company shall immediately upon his appointment to any of the above offices in other body corporate, disclose to the Company the particulars relating to his office in the other body corporate or bodies corporate which are required to be specified under Section 170 of the Act and the Rules.

- (b) Every Director and Key Managerial Personnel, shall give notice to the Company of such matters relating to himself as may be necessary for the purpose of enabling the Company to comply with the provisions of Section 170 of the Act and the Rules.
- 153. (a) The Company may (subject to the provisions of Section 169 and other applicable provisions of the Act and these Articles) by passing a Special Resolution at the General Meeting remove any Director other than Special Directors or Debenture Directors before the expiry of his period of office.
 - (b) Special notice as provided by Section 115 of the Act shall be required of any resolution to remove a Director under this Article 153 or to appoint some other Person in place of a Director so removed at the meeting at which he is removed.
 - (c) On receipt of notice of a resolution to remove a Director under this Article 153, the Company shall forthwith send a copy thereof to the Director concerned and the Director (whether or not he is member of the Company) shall be entitled to be heard on the resolution at the meeting.
 - (d) Where notice is given of a resolution to remove a Director under this Article 153 and the Director concerned makes with respect thereto representations in writing to the Company (not exceeding a reasonable length) and requests their notification to Members, the Company shall unless the representations are received by it too late for it to do so.
 - (i) In the notice of the resolution given to Members of the Company state the fact of the representations having been made, and
 - (iii) Send a copy of the representation to every Member of the Company to whom notice of the Meeting is sent (whether before or after receipt of the representations by the Company), and if a copy of the representations, is not sent as aforesaid because they were received too late or because of the Company's default, the Director may (without prejudice to his right to be heard orally) require that the representations be read out at the Meeting, provided that copies of the representations need not be sent or read out at the Meeting if so directed by the Court.
 - (e) Subject to the provisions of the other Articles hereof and in particular Article 151 hereof a vacancy created by the removal of a Director under this Article may, if he had been appointed by the Company in General Meeting or by the Board in pursuance of Section 161 of the Act be filled by the appointment of another Director in his stead by the Meeting at which he is removed, provided special notice of the intended appointment has been given under Article 153(b) hereof. A Director so appointed shall hold office until the date up to which his predecessor would have held office if he had not been removed as aforesaid.
 - (f) If the vacancy is not filled under Article 153(e), it may be filled as a casual vacancy in accordance with the provisions, in so far as they may be applicable, of the said Article 136 and of Section 161 of the Act, and all the provisions of that Section shall apply accordingly;

Provided that the Director who was removed from office under this Article 153 shall not be reappointed as a Director by the Board of Directors.

- (g) Nothing contained in this Article 153 shall be taken:
 - (i) as depriving a Person removed thereunder of any compensation or damages

- payable to him in respect of the termination of his appointment as director or of any appointment terminating with that as director; or
- (ii) as derogating from any power to remove a Director which may exist apart from this Article 153.
- 154. The continuing Directors may act notwithstanding any vacancy in the Board; but if so long as their number is reduced below the quorum prescribed by the Act for a meeting of the Board, the continuing Directors or Director may act for the purpose of increasing the number of Directors to that fixed for the quorum, or of summoning a General Meeting of the Company, but for no other purpose.

ROTATION OF DIRECTORS

- 155. Not less than two-third of the total number of Directors shall;
 - (a) be Persons whose period of office is liable to determination by retirement of Directors by rotation, and
 - (b) save as otherwise expressly provided in the Act, be appointed by the Company in General Meeting.

The remaining Directors shall, in default of and subject to any regulations in the Articles of the Company, also be appointed by the Company, in General Meeting. For the purposes of this Article 155 "total number of Directors" shall not include independent Directors on the Board, whether appointed under this Act or any other law for the time being in force.

- 156. (a) At every Annual General Meeting one-third of such of the Directors for the time being as are liable to retire by rotation, or if their number is not three or a multiple of three, then the number nearer to one-third, shall retire from office.
 - (b) The Directors to retire by rotation at every Annual General Meeting shall be those who have been longest in office since their last appointment, but as between Persons who became Directors on the same day, those who are to retire shall, in default of and subject to any agreement amongst themselves, be determined by lot.
 - (c) At the Annual General Meeting at which a Director retires as aforesaid, the Company may fill up the vacancy by appointing the retiring Director or some other Person thereto.
 - (d) (i) If the place of the retiring Director is not so filled up and that meeting has not expressly resolved not to fill the vacancy, the meeting shall stand adjourned till the same day in the next week, at the same time and place.
 - (ii) If at the adjourned meeting also, the place of the retiring Director is not filled up and that meeting also has not expressly resolved not to fill the vacancy, the retiring Director shall be deemed to have been re-appointed at the adjourned meeting, unless
 - (1) at that meeting or at the previous meeting a resolution for the reappointment of such Director has been put to the meeting and lost;
 - (2) the retiring Director has, by a notice in writing addressed to the Company or its Board of Directors, expressed his unwillingness to be

so re-appointed;

- (3) he is not qualified or is disqualified for appointment;
- (4) a resolution, whether special or ordinary, is required for his appointment or re-appointment in virtue of any provisions of the Act; or

Explanation: In this Article the expression "Retiring Director" means Director retiring by rotation.

PROCEEDINGS OF DIRECTORS

- 157. The Board of Directors may meet for the conduct of business, adjourn and otherwise regulate its meetings, as it thinks fit and shall so meet at least once in every quarter and at least four such meetings shall be held in every calendar year and not more than one hundred and twenty days shall intervene between two consecutive meetings. The meetings of the Board may be called by the Company Secretary on instructions of any member of the Board or by any member of the Board or by the Chairman. The provisions of this Article 157 shall not be deemed to be contravened merely by reason of the fact that meetings of the Board, which had been called in compliance with the terms herein mentioned could not be held for want of quorum.
- 158. The participation of Directors in a meeting of the Board or a meeting of a Committee of the Board may be either in person or through video conferencing or audio visual means or teleconferencing, as may be prescribed by the provisions of the Act and the Rules.
- 159. (a) At least seven calendar days' notice of every meeting of the Board shall be given in writing to every Director. Such notice shall be accompanied by the agenda setting out the business proposed to be transacted at the meeting of the Board, provided, however, that a meeting of the Board may be called at shorter notice to transact urgent business subject to the condition that at least one independent Director, if any, shall be present at the meeting. in case of absence of independent Directors from such a meeting of the Board, decisions taken at such a meeting shall be circulated to all the Directors and shall be final only on ratification thereof by at least one independent director, if any. Notice of Board Meetings to all Directors shall be given sent by hand delivery or by post or by electronic means to every Director at his address registered with the Company.
 - (b) The Board shall only transact the business set out in the agenda accompanying the notice to the Directors provided however that with the consent of the Board, any other business not set out in the agenda may be transacted.
- 160. The participation of directors in a meeting of the Board may be either in person or through video conferencing or other audio visual means, as may be prescribed by the Act and the Rules, which are capable of recording and recognising the participation of the directors and of recording and storing the proceedings of such meetings along with date and time.
- 161. The quorum for a meeting of the Board shall be one-third of the total strength of the Board for the time being or three Directors whichever is higher, and the participation of the Directors by video conferencing or by other audio visual means shall also be counted for the purposes of quorum under this Article 161. Where a meeting of the Board could not be held for want of quorum, then the meeting shall automatically stand adjourned to the same day at the same time

and place in the next week or if that day is a national holiday till the next succeeding day, which is not a national holiday, at the same time and place. If at the adjourned meeting also, there is no quorum, the Directors present at such adjourned meeting being not less than three in number shall constitute quorum for that particular meeting and the business as per the agenda already circulated to the Directors, in respect of the original meeting transacted by such Directors at such adjourned meeting shall be valid and binding.

- 162. A meeting of the Board at which a quorum is present shall be competent to exercise all or any of the authorities, powers and discretion which by or under the Act or the Articles or the Regulations of the Company are, for the time being, vested in or exercisable by the Board generally.
- 163. (a) The Board of Directors may create such Committees as it deems appropriate or as may be required by applicable law. Permanent invitees of the Committees, if any, shall be determined by the Board of Directors.
 - (b) The Board may, from time to time, dissolve or discharge any such Committee of the Board either wholly or in part and either as to Persons or purposes, but every Committee of the Board to be formed shall in the exercise of the powers so delegated conform to any regulations that may, from time to time, be imposed on it by the Board.
 - (c) All acts done by any such Committee of the Board in conformity with such regulations and in fulfilment of the purpose of their constitution but not otherwise shall have the like effect as is done by the Board.
- 164. The meetings and proceedings of any such Committee of the Board consisting of two or more members shall be governed by the provisions herein contained for regulating the meetings and proceedings of the Directors, so far as the same are applicable thereto, and are not superseded by any regulations made by the Directors under Article 163. The Board may subject to the provisions of the Act from time to time fix the remuneration to be paid to any member or members of their body constituting a Committee appointed by the Board in terms of these Articles and may pay the same.
- 165. Save as otherwise expressly provided in the Act, a resolution in writing, signed, whether manually or by secure electronic mode, by a majority of the members of the Board or of a Committee thereof, for the time being entitled to receive notice of a meeting of the Board or Committee, shall be valid and effective as if it had been passed at a meeting of the Board or Committee, duly convened and held.
- 166. All acts done by any meeting of the Board or by a Committee of the Board, or by any Person acting as a Director shall, notwithstanding that it shall afterwards be discovered that there was some defect in the appointment of such Directors or Persons acting as aforesaid or that they or any of them were disqualified or that the appointment of any of them be terminated by virtue of any provisions contained in the Act in these Articles, be as valid as if every such Person had been duly appointed, and was qualified to be a Director. Provided that nothing in this Article 166 shall be deemed to give validity to acts done by a Director after his appointment has been shown to the Company to be invalid or to have terminated.
- 167. (a) The Company shall cause minutes of all proceedings of every meeting of the Board and of every Committee of the Board to be kept by making entries thereof in books kept for that purpose with their pages consecutively numbered within thirty days of the

conclusion of every such meeting.

- (b) Each page of every such book shall be initialled or signed and the last page of the record of proceedings of each book shall be dated and signed by the Chairman of that meeting of the Board or of the Committee, as the case may be, or the Chairman of the next succeeding meeting of the Board or the Committee, as the case may be.
- (c) The minutes of each meeting shall contain a fair and correct summary of the proceedings thereat.
- (d) All appointments of officers made at any of the meetings aforesaid shall be included in the minutes of the meeting.
- (e) The minutes shall also contain details of:
 - (i) the names of Directors and other members of the Committee present at the meeting;
 - (ii) all orders made by the Board and any Committee of the Board;
 - (iii) all resolutions and proceedings of meetings of the Board; and
 - (iv) in the case of each resolution passed at the meeting, the names of the Directors, if any, dissenting from, or not concurring in, the resolution.
- (f) Nothing contained in Articles 167(a) to 167(e) shall be deemed to require the inclusion in such minutes of any matter which, in the opinion of the Chairman of the Meeting:
 - (i) is, or could reasonably be regarded as defamatory of any Person;
 - (ii) is irrelevant or immaterial to the proceedings, or
 - (iii) is detrimental to the interest of the Company.
- (g) The Chairman shall exercise an absolute discretion in regard to the inclusion or non-inclusion of any matter in the minutes on the grounds specified in this Article 167.
- (h) Minutes kept in accordance with the aforesaid provisions shall be evidence of the proceedings recorded therein.
- 168. (a) The Directors shall cause to be kept at the registered office of the Company:
 - (i) A Register of the Directors, Managing Director, Manager and Secretary of the Company containing the particulars required by Section 170 of the Act;
 - (ii) A Register of Contracts with companies and firms in which the Directors are interested, containing the particulars required by Section 189 of the Act; and
 - (iv) A Register of Directors shareholding containing the particulars required by Section 170 of the Act; and
 - (iv) Other registers and indexes as required by the Act.
 - (b) The Company shall comply with the provisions of Sections 170, 189, 190 and other Sections of the Act with regard to the inspection of registers and furnishing copies or extracts so far as the same be applicable to the Company.

POWER OF DIRECTORS

- 169. The business of the Company shall be managed by the Board, and the Board may exercise all such powers of the Company and do all such acts and things as are not, by the Act, or any other law or by the Memorandum of Association of the Company or by these Articles required to be exercised by the Company in a General Meeting, subject nevertheless to the provisions of the Act, any other law, or in the Memorandum of the Company or these Articles or any regulations, being not inconsistent therewith and duly made thereunder including regulations, made by the Company in a General Meeting, but no regulation made by the Company in General Meeting shall invalidate any prior act of the Board which would have been valid if that regulation had not been made.
- 170. (a) Without derogating from the powers vested in the Board of Directors under these Articles, the Board shall exercise the following powers on behalf of the Company and they shall do so only by means of resolutions passed at meetings of the Board.
 - (i) The power to make calls on Shareholders in respect of money unpaid on their Shares;
 - (ii) The power to authorize buy-back of securities of the Company under Section 68 of the Act;
 - (iii) The power to issue securities, including Debentures, whether in or outside India;
 - (iv) The power to borrow moneys;
 - (v) The power to invest the funds of the Company;
 - (vi) The power to grant loans or give guarantee or provide security in respect of loans;
 - (vii) The power to approve financial statements and the Directors' report;
 - (viii) The power to diversify the business of the Company;
 - (ix) The power to approve amalgamation, merger or reconstruction;
 - (x) The power to take over a company or acquire a controlling or substantial stake in another company;
 - (xi) The power to make political contributions;
 - (xii) The power to appoint or remove Key Managerial Personnel of the Company;
 - (xiii) To appoint internal auditors and secretarial auditor;
 - (xiv) The power to take note of appointment(s) or removal(s) of employees of the Company one level below the Key Managerial Personnel of the Company;
 - (xv) The power to buy, sell investments held by the Company (other than trade investments), constituting five percent or more of the paid up share capital and free reserves of the investee company;

- (xvi) The power to invite or accept or renew public deposits and related matters;
- (xvii) The power to review or change the terms and conditions of public deposits; and
- (xviii) The power to approve quarterly, half yearly and annual financial statements or financial results, as the case may be.

Provided that the Board may by resolution passed at the meeting, delegate to any Committee of Directors, the Managing Director, the Manager or any other principal officer of the Company or in the case of a branch office of the Company, a principal officer of the branch office, the powers specified in Articles 170(a)(iv), 170(a)(v) and 170(a)(vi) above on such condition as the Board may prescribe.

- (b) Nothing in this Article shall be deemed to affect the right of the Company in a General Meeting to impose restrictions and conditions on the exercise by the Board of any of the powers referred to in items (i) to (xviii) of Article 170(a) above.
- 171. Without prejudice to the general powers conferred by Article 170 and so as not in any way to limit or restrict these powers, and without prejudice to the other powers conferred by the Act and these Articles, but subject to the restrictions contained in the other Articles hereof, it is hereby declared that the Directors shall have the following powers:
 - (a) to pay/reimburse the costs, charges, and expenses, preliminary and incidental to the incorporation, promotion, establishment and registration of the Company;
 - (b) to purchase or otherwise acquire for the Company any lands, buildings, machinery, premises, assets, hereditaments property, effects, rights or privileges, credits, royalties, bounties and goodwill of any Person, firm or company which the Company is authorized to acquire, at or for such price or consideration and generally on such terms and conditions as they may think fit, and in any such purchase or other acquisition accept such title as the Directors may believe or may be advised to be reasonably satisfactory;
 - (c) to pay and charge to the Capital account of the company any commission or interest lawfully payable thereat under the provisions of Section 40 of the Act;
 - (d) at their discretion, and subject to the provisions of the Act, to pay for any property, rights or privileges acquired by or services rendered to the Company, either wholly or partly, in cash or in Shares, stock, bonds, Debentures, debenture-stock, mortgages or other securities of the Company, and any such Shares may be issued either as fully paid up or with such amount credited as paid up thereon as may be agreed upon; and any such bonds, Debentures, debenture-stock, mortgages or other securities may be either specifically charged upon all or any part of the property of the Company and its uncalled Capital or not so charged;
 - (e) to secure the fulfilment of any contract or engagements entered into by the Company by mortgage or charge of all or any of the property of the Company and its uncalled Capital for the time being or in such manner as the Directors may think fit;
 - (f) to accept from any Member, so far as may be permissible by law, a surrender of his Shares or any part thereof, on such terms and conditions as shall be agreed;
 - (g) to appoint any Person to accept and hold in trust for the Company any property belonging to the Company, or in which it is interested, or for any other purpose; and to

- execute and do all such deeds and things as may be required in relation to any such trust, and to provide for the remuneration of such trustee or trustees;
- (h) to institute, conduct, defend, compound or abandon any legal proceedings by or against the Company or its officers, or its other employees or otherwise concerning the affairs of the Company, and also to compound and allow time for payment on satisfaction of any debts due, and of any claims or demands by or against the Company, and to refer any differences to arbitration, and to observe and perform any awards made thereon;
- (i) subject to the provisions of the Act, to give in the name and on behalf of the Company such indemnities and guarantees as may be necessary;
- (j) to act on behalf of the Company in all matters relating to bankrupts and insolvents;
- (k) to make and give receipts, release, and other discharges for moneys payable to the Company and for the claims and demands of the Company;
- (l) subject to the provisions of Sections 179, 180 and 186 of the Act, to invest and deal with any moneys of the Company upon such security (not being Shares of this Company), or without security and in such manner as they may think fit and, from time to time, vary or realize such investments. Save as provided in Section 187 of the Act, all investments shall be made and held in the Company's own name;
- (m) to execute in the name and on behalf of the Company in favour of any Director or other Person who may incur or be about to incur any personal liability whether as principal or surety, for the benefit of the Company, such mortgages of the Company's property (present and future) as they think fit; and any such mortgage may contain a power of sale and such other powers, provisions, covenants and agreements as shall be agreed upon;
- (n) to determine, from time to time, who shall be entitled to sign, on the Company's behalf, bills, promissory notes, receipts, acceptances, endorsements, cheques, Dividend warrants, releases, contracts, instruments and documents, and general correspondence, and to give the necessary authority for such purpose;
- subject to the provisions of the Act and the Rules, to provide for the welfare of Directors (o) or ex-Directors or employees or ex-employees of the Company and other Persons who are or were working for the Company delegated or seconded by any other organizations and the wives, widows and families or the dependents or connections of such Persons by building or contributing to the building of houses, dwellings, or by grants of money, pensions, gratuities, bonus, allowances or other payments; or by creating and from time to time subscribing or contributing to provident fund, including acceptance of transfer of money or from any other provident fund and any superannuation fund for being credited to the relevant fund created by the Company and to other associations, institutions, funds or trusts including any research and development organizations, training schools, by providing or subscribing or contributing towards research and development centres and places of instructions and recreation, hospitals and dispensaries, medical and other attendance and other assistance as the Directors shall think fit; and to subscribe or contribute or otherwise to assist or to guarantee money to charitable, benevolent, religious, scientific, educational, cultural, social and other institutions for objects which shall have any moral or other claims to support or aid by the Company, either by reason of locality of operation, or of public and general utility or otherwise;
- (p) (i) before recommending any Dividend, to set aside, out of the profits of the Company such sums as they may think proper for depreciation in accordance with the provisions of the Act, or as a reserve or reserves which shall, at the

discretion of the Board, be applicable for any purpose to which the profits of the Company may be properly applied, including provision for meeting contingencies or for equalising dividends; and pending such application, may, at the like discretion, either be employed in the business of the Company or be invested in such investments (other than Shares) as the Board may, from time to time, think fit;

- (ii) before the declaration of any dividend in any Financial Year, transfer such percentage of its profits for that financial year as it may consider appropriate to the reserves of the Company; and
- (iii) carry forward any profits which it may consider necessary not to divide, without setting them aside as a reserve.
- (q) to distribute by way of bonus amongst the staff of the Company a Share or Shares in the profits of the Company, and to give to any Director, officer or other Person employed by or working for the Company, a commission on the profits of any particular business or transaction; and to charge such bonus or commission as part of the working expenses of the Company;
- (r) to effect, make and enter into, on behalf of the Company, all transactions, agreements and other contracts within the scope of the business of the Company; and to appoint, constitute and at their discretion, remove or dissolve any consultant, advisors and Committee(s) as they may from time to time think fit, and to determine their powers and duties and fix their remuneration;
- (s) from time to time and at any time, to make such arrangements as the Directors may consider appropriate for managing any of the affairs of the Company in any specified locality in India or elsewhere and to appoint any Person(s) to be in charge of such offices;
- (t) subject to Section 179 of the Act, from time to time, and at any time to appoint any Person and to delegate to the Person so appointed, any of the powers, authorities and discretion for the time being vested in the Directors; and to authorize any Person to fill up any vacancies therein and to act notwithstanding vacancies; and such appointment or delegation may be made on such terms, and subject to such conditions as the Directors may think fit, and the Directors may, at any time remove any Person so appointed, and may annul or vary any such delegation;
- (u) at any time, and from time to time, by power of attorney to appoint any Person or Persons to be the attorney or attorneys of the Company, for such purposes and with such powers, authorities and discretion (not exceeding those vested in or exercisable by the Directors under these presents) and subject to the provisions of Section 179 of the Act and for such period and subject to such conditions as the Directors may, from time to time, think fit, and any such appointment may (if the Directors think fit) be made in favour of any Person or in favour of any Company, or the Shareholders, Directors, nominees or Managers of any Company or firm or otherwise in favour of any fluctuating body of Persons whether nominated directly or indirectly by the Directors and any such power of attorney may contain such powers for the protection of convenience of Persons dealing with such attorneys as aforesaid to sub-delegate all or any of the powers, authorities and discretion for the time being vested in them;
- (v) subject to Section 188 of the Act, for or in relation to any of the matters aforesaid or otherwise for the purposes of the Company to enter into all such negotiations and contracts and rescind and vary all such contracts, and execute and do all such acts, deeds and things in the name and on behalf of the Company as they may consider expedient;

- (w) to purchase or otherwise acquire or obtain licence for the use of, and to sell, exchange or grant licence for the use of any trade mark, patent, invention or technical know-how;
- (x) to undertake on behalf of the Company any payment of all rents and the performance of the covenants, conditions and agreements contained in or reserved by any lease that may be granted or assigned or to otherwise acquired by the Company, and to purchase the reversion or reversions, and otherwise to acquire the fee simple of all or any of the lands of the Company for the time being held under lease or for an estate less than freehold estate;
- (y) to improve, manage, develop, exchange, lease, sell, re-sell, and re-purchase, dispose of, deal or otherwise turn to account, any property (movable or immovable) or any rights or privileges belonging to or at the disposal of the Company or in which the Company has or may have interest;
- (z) to let, sell or otherwise dispose of, subject to the provisions of Section 180 of the Act any property of the Company, either absolutely or conditionally and in such manner and upon such terms and conditions in all respects as they think fit and accept payments of satisfaction for the same in cash or otherwise as they think fit;
- (aa) from time to time to make, vary and repeal bye-laws, regulations and other rules, guidelines or instructions for regulating the business of the Company, its officials employees and other Persons having dealings with the Company;
- (bb) to get insured and keep insured against loss or damage by fire or otherwise for such period and to such extend as they may think proper, all or any part of the building, machinery, goods, stores, produce and other movable property of the Company either separately or co-jointly, also to insure all or any portion of the goods, produce, machinery and other articles imported or exported by the Company and to assign, surrender or discontinue any policies of assurance effected in pursuance of this power; and
- (cc) Subject to Section 179 of the Act, to open accounts with any bank or bankers or with any Company, firm or individual and to pay money into and draw money from any account from time to time as the Directors may think fit.

DECISIONS OF THE BOARD OF DIRECTORS

172. A resolution of the Board of Directors shall be adopted by the affirmative vote of the majority of the Directors present at a meeting, at which a quorum of the Board of Directors is present.

CHIEF EXECUTIVE OFFICER

173. (a) The Chief Executive Officer /Managing Director shall be vested with the day-to-day responsibility and discretion for managing the business and operations of the Company and the authority conferred on him by the Board of Directors. The Chief Executive Officer/Managing Director shall have, in addition to the powers and authorities normally incidental to the office of Chief Executive Officer / Managing Director, and the powers and duties set forth in these Articles, if any, the following authorities and accountabilities: (i) accountability to the Board of Directors to achieve the milestones, requirements and objectives as set forth in annual operating and capital budget or otherwise; (ii) day-to-day administration of the Company and co-ordination of the subcontractors; (iii) representing the Company in dealings with the Shareholders and third parties; (iv) proposing to the Board of Directors updates and amendments to

annual operating and capital budgets; (v) delegating authority to other officers or employees of the Company the authority to discharge the functions on behalf of, and enter into transactions in the name of, the Company consistent with the Act and these Articles; and (vi) managing the personnel resources of the Company.

- (b) The Chief Executive Officer / Managing Director may be appointed by the Board for such term, at such remuneration and upon such terms and conditions as it may think fit; and any Chief Executive Officer / Managing Director so appointed may be removed by means of a resolution of the Board.
- (c) A Director may be appointed as Chief Executive Officer.

CHIEF FINANCIAL OFFICER

- 174. (a) The Chief Financial Officer may be appointed by the Board for such term, at such remuneration and upon such terms and conditions as it may think fit; and any Chief Financial Officer so appointed may be removed by means of a resolution of the Board. Provided that, the Board may appoint one or more Chief Financial Officers for its multiple businesses.
 - (b) A Director may be appointed as a Chief Financial Officer.
- 175. The Company shall have Senior Officers as decided by the Board of Directors. The Senior Officers shall be resident Indian citizens and shall discharge such functions as may be decided by the Board of Directors.

MANAGING DIRECTOR/ WHOLE TIME DIRECTOR

- 176. (a) Subject to the provisions of the Act and Articles 173 to 175, the Board shall have the power to appoint and reappoint and from time to time remove one or more Persons to be Managing Director(s) and whole time Director(s) of the Company for a fixed period as the Board thinks fit, and subject to the provisions of Article 178, the Board may by resolution vest in such Managing Director such of the powers hereby vested in the Board generally as it thinks fit, and such power may be made exercisable for such period or periods and upon such conditions and subject to such restrictions as it may determine.
 - (b) A Managing Director or a Whole-time Director shall receive such remuneration (whether by way of salary, perquisites, commission or participation in profits, or otherwise or partly in one way and partly in another) as the Directors may, subject to the provisions of the Act, or any other law applicable for the time being in force in that behalf, determine.
 - (c) Subject to the provisions of the Act, the Board of Directors may entrust to and confer upon a Managing Director or Whole-time Director any of the powers exercisable by the Board upon such terms and conditions and with such restrictions as the Board may think fit, and either collaterally with or to the exclusion of powers of the Board, and may, from time to time, revoke, withdraw, alter or vary any of such powers.
- 177. (a) The Company shall not appoint or employ, or continue the appointment or employment of, a Person as its Managing or Whole-time Director who;

- (i) is an undischarged insolvent, or has, at any time, been adjudged an insolvent;
- (ii) suspends, or has, at any time, suspended with his creditors, or makes, or has at any time made, a composition with them; or
- (iii) is, or has at any time been convicted by a Court of an offense involving moral turpitude.
- (b) If the Managing or Whole-time Director ceases to hold the office of Director he shall ipso facto and immediately cease to be a Managing Director or whole time Director, as the case may be, of the Company.

MANAGER

- 178. (a) Subject to the provisions of the Act, if a Managing Director has not been appointed as provided for in the Articles, the Board may appoint a Manager for such term and on such remuneration and upon such conditions as it may deem fit; and any Manager so appointed may be removed by the Board.
 - (b) The Manager shall exercise such power or powers and for such period or periods and upon such conditions and subject to such restrictions as the Board may determine.

THE SECRETARY

179. Subject to the provisions of the Act, the Board of Directors may, from time to time appoint and, at their discretion remove any individual (hereinafter called "the Secretary" or the "Company Secretary") who shall have such qualifications as may be prescribed to perform any functions, which by the Act or these Articles are to be performed by the Secretary, and to execute any other purely ministerial or administrative duties which may from time to time be assigned to the Secretary. The Board of Directors may also at any time appoint some Persons (who need not be the Secretary) to keep the registers required to be kept by the Company. The Secretary may be appointed by the Board of Directors on such terms and conditions and at such remuneration as it may deem fit.

THE SEAL

- 180. The Board may, at its discretion, provide a Common Seal for the purposes of the Company, and shall have from time to time, power to destroy the same and substitute a new Seal in lieu thereof, and the Board shall provide for the safe custody of the Seal for the time being.
- 181. The Seal, if any, shall not be affixed to any instrument except by the authority of a resolution of the Board of Directors or a Committee of the Board authorized by it in that behalf and except in the presence of at least one Director and of the Secretary or such other Person as the Board may appoint for the purpose, and such one Director and Secretary or other Person as aforesaid shall sign every instrument to which the Seal of the Company, if any, is so affixed in their presence.

Provided further that the certificates of Shares or Debentures shall be sealed in the manner and in conformity with the provisions of the Rules and any statutory modifications thereof, for the time being in force.

DIVIDENDS

- 182. The profits of the Company, subject to the provisions of these Articles, shall be divisible among the Members in proportion to the amount of Capital paid up or credited as paid up on the Shares held by them respectively but if any Share is issued on terms providing that it shall rank for Dividend as from a particular date such share shall rank for Dividend accordingly.
- 183. The Company, in General Meeting, may declare Dividends to be paid to Members according to their respective rights but no Dividend shall exceed the amount recommended by the Board, but the Company in General Meeting may declare a lower Dividend.
- 184. No Dividend shall be declared or paid otherwise than out of profits of the Financial Year arrived at after providing for depreciation in accordance with the provisions of Section 123 of the Act, or out of the profits of the Company for any previous Financial Year or years arrived at after providing for depreciation in accordance with those provisions and remaining undistributed or out of both.

Provided that:

- (a) if the Company has not provided for depreciation for any previous Financial Year or years, it shall, before declaring or paying any Dividend for any Financial Year, provide for such depreciation out of the profits of that Financial Year or out of the profits of any other previous Financial Year or years;
- (b) if the Company has incurred any loss in any previous Financial Year or years, the amount of the loss or an amount which is equal to the amount provided for depreciation for that year or those years whichever is less, shall be set off against the profits of the Company for the year for which the Dividend is proposed to be declared or paid or against the profits of the Company for any previous Financial Year or years arrived at in both cases after providing for depreciation in accordance with the provisions of Schedule II of the Act or against both.

Provided further that, no Dividend shall be declared or paid for any Financial Year out of the profits of the Company for that year arrived at after providing for depreciation as above, except after the transfer to the reserves of the Company of such percentage of its profits for that year as may be prescribed in accordance with Section 123 of the Act or such high percentage of its profits as may be allowed in accordance with that Section.

Provided further that, the Board may carry forward any profits which it may consider necessary not to divide, without setting them aside as reserve.

- 185. The Board may, from time to time, pay to the Members such interim Dividend as in its judgment the position of the Company justifies.
- 186. Where the Capital is paid in advance of the calls upon the footing that the same shall carry interest, such Capital shall not, whilst carrying interest, confer a right to participate in profits.
- 187. The Company shall pay Dividends in proportion to the amount paid up or credit as paid-up on some Shares than on others.
- 188. The Board may retain the Dividends payable upon Shares in respect of which any Person has become entitled to be a Member under Article 56 or any Person under that Article is entitled to transfer until such Person becomes a Member in respect of such Shares or shall duly transfer the same.

- 189. A waiver in whole or in part of any Dividend on any Share by any document (whether or not under Seal, if any) shall be effective only if such document is signed by the Member (or the Person entitled to the Share in consequence of the death or bankruptcy of the holder) and delivered to the Company and if or to the extent that the same is accepted as such or acted upon by the Board.
- 190. Any one of the several Persons who are registered as joint holders of any Share may give effectual receipts for all Dividends or bonus and payments on account of Dividends or bonus or other moneys payable in respect of such Share.
- 191. No Member shall be entitled to receive payment of any interest or Dividend or bonus in respect of his Share whilst any moneys may be due or owing from him to the Company in respect of such Share or otherwise, however, either alone or jointly with any other Person or Persons; and the Board may deduct from the interest or Dividend payable to any Member all such sums of money so due from him to the Company.
- 192. (a) Unless otherwise directed, any Dividend payable in cash in respect of Shares may be paid by electronic mode, cheque or warrant payable only in India, or by a pay slip or receipt having the force of a cheque or warrant, sent through the post to the registered address of the Member or Person entitled, or in case of joint holders to that one of them first named in the Register of Members in respect of the joint holding.
 - (b) Every such cheque or warrant shall be made payable to the registered holder of Shares or to his order or to his bankers.
 - (c) Payment in any way whatsoever shall be made at the risk of the Person entitled to the money paid or to be paid. The Company shall not be liable or responsible for any cheque or warrants or payslip or receipt lost in transmission, or for any Dividend lost to the Member or Person entitled thereto by the forged endorsement of any cheque or warrant or the forged signature on any payslip or receipt or the fraudulent or improper recovery of the Dividend by and other means. The Company will be deemed to having made a payment and received a good discharge for it if a payment using any of the foregoing permissible means is made.
- 193. The Company shall pay the Dividend or send the cheque or warrant or given instructions for payment in electronic mode in respect thereof to the Member entitled to the payment of Dividend in accordance with the provisions of the law from the date of the declaration unless:
 - (a) where the Dividend could not be paid by reason of the operation of any law;
 - (b) where a Shareholder has given directions regarding the payment of the Dividend and those directions cannot be complied with;
 - (c) where there is a dispute regarding the right to receive the Dividend; or
 - (d where for any other reason, the failure to pay the Dividend or to post the warrant within the period aforesaid was not due to any default on the part of the Company.
- 194. Subject to the rights of persons, if any, entitled to shares with special rights as to dividends, all dividends shall be declared and paid according to the amounts paid or credited as paid on the shares in respect whereof the dividend is paid, but if and so long as nothing is paid upon any of the shares in the Company, dividends may be declared and paid according to the amounts of the Shares.

- 195. No unclaimed Dividend shall be forfeited by the Board before the claim becomes barred by law, and such forfeiture, when affected, shall be annulled in appropriate cases, and the Company shall comply with the provisions of Section 124 of the Act, as applicable, in respect of such Dividend.
- 196. Unclaimed Dividends shall be transferred to the unpaid Dividend account of the Company as hereinafter provided:
 - (a) Where the Dividend has been declared but not paid but the warrant in respect thereof has not been posted, within thirty days from the date of the declaration to any Shareholder entitled to the payment thereof, the Company shall within seven days from the date of expiry of the said period of thirty days transfer the total amount of Dividend which remains unpaid or in relation to which no Dividend warrant has been posted within the said period of thirty days to a special account to be opened by the Company in that behalf in any Scheduled Bank to be called "Unpaid Dividend Account of ".....".
 - A claim to any money so transferred to the general revenue account may be preferred to the Central Government by the Shareholders to whom the money is due.
 - (b) Any money transferred to the unpaid Dividend account of the Company in pursuance of sub-Article (a) hereof which remains unpaid or unclaimed for a period for seven consecutive years or more from the date of such transfer, shall be transferred by the Company to the Investor Education and Protection Fund;
 - (c) The Company shall, when making any transfer under Article 196(b) hereof to the general revenue account of the Central Government of any unpaid or unclaimed Dividend, furnish to such officer as the Central Government may appoint in this behalf a statement in the prescribed form setting forth in respect of all sums included in such transfer, the nature of the sums, the names and last known addresses of the Persons entitled to receive the sum, the amount to which each Person is entitled, and the nature of his claim thereto and such other particulars as may be prescribed by the Central Government.
- 197. Any General Meeting declaring a Dividend may, on the recommendations of the Board of Directors, make a call on the Members of such amount as the Meeting fixes but so that the call on each Member shall not exceed the Dividend payable to him, and so that the call be made payable at the same time as the Dividend; and the Dividend may, if so arranged between the Company and the Members, be set off against the call.
- 198. No Dividend shall be payable except in cash. Provided that nothing in this Article 198 shall be deemed to prohibit the capitalization of the profits or reserves of the Company for the purpose of issuing fully paid up bonus Shares or paying up any amount for the time being unpaid on any Shares held by Members of the Company.
- 199. Subject to the rights of Persons, if any, entitled to Shares with special rights as to Dividends, all Dividends shall be declared and paid according to the amounts paid or credited as paid on the Shares in respect whereof the Dividend is paid, but if and so long as nothing is paid upon any of the Shares in the Company, Dividends may be declared and paid according to the amounts of the Shares. No amount paid or credited as paid on a Share in advance of calls shall be treated for the purposes of this Article 199 as paid on the Share.
- 200. The Board may deduct from any Dividend payable to any Member all sums of money, if any, presently payable by him to the Company on account of calls or otherwise in relation to the Shares of the Company.

201. No Dividend shall bear interest against the Company.

CAPITALISATION BUY-BACK OF SHARES

- 202. (a) The Company in General Meeting may, upon the recommendation of the Board:
 - (i) that it is desirable to capitalise any part of the amount for the time being standing to the credit of any of the company's reserve accounts, or to the credit of the profit and loss account, or otherwise available for distribution; and
 - (ii) that such sum be accordingly set free for distribution in the manner specified in Article 202(b) amongst the Members who would have been entitled thereto, if distributed by way of dividend and in the same proportions.
 - (b) The sum aforesaid shall not be paid in cash but shall be applied, subject to the provisions of the Act, either in or towards:
 - (i) paying up any amounts for the time being unpaid on any Shares held by such Members respectively;
 - (ii) paying up in full, unissued shares of the company to be allotted and distributed, credited as fully paid-up, to and amongst such members in the proportions aforesaid:
 - (iii) partly in the way specified in Article 202(b)(i) and partly that specified in Article 202(b)(ii);
 - (v) A securities premium account and a capital redemption reserve account may, for the purposes of this Article, be applied in the paying up of unissued Shares to be issued to Members of the Company as fully paid bonus Shares;
 - (vi) The Board shall give effect to the resolution passed by the Company in pursuance of this Article 202.
- 203. (a) Whenever such a resolution as aforesaid shall have been passed, the Board shall:
 - (i) make all appropriations and applications of the undivided profits resolved to be capitalised thereby, and all allotment and issues of fully paid Shares if any; and
 - (ii) generally do all acts and things required to give effect thereto.
 - (b) The Board shall have the power:
 - (i) make such provisions, by the issue of fractional certificates or by payment in cash or otherwise as it thinks fit, for the case of shares becoming distributable in fractions; and
 - (ii) to authorise any Person to enter, on behalf of all the Members entitled thereto, into an agreement with the company providing for the allotment to them respectively, credited as fully paid-up, of any further Shares to which they may be entitled upon such capitalisation, or as the case may require, for the payment by the Company on their behalf, by the application thereto of their respective proportions of profits resolved to be capitalised, of the amount or any part of the amounts remaining unpaid on their existing Shares;

(iii) Any agreement made under such authority shall be effective and binding on such Members.

BUY-BACK OF SHARES

204. Notwithstanding anything contained in these Articles but subject to the provisions of Sections 68 to 70 of the Act, or any other law for the time being in force, the Company may purchase its own Shares or other specified securities.

ACCOUNTS

- 205. (a) The Company shall keep at the office or at such other place in India as the Board thinks fit, proper books of accounts in accordance with Section 128 of the Act with respect to:
 - (i) all sums of moneys received and expended by the Company and the matters in respect of which the receipts and expenditure take place;
 - (ii) all sales and purchases of goods by the Company; and
 - (iii) the assets and liabilities of the Company.

The books of accounts may also be maintained in electronic form.

- (b) Where the Board decides to keep all or any of the books of accounts at any place other than the Office of the Company, the Company shall, within seven days of the decision, file with the Registrar a notice in writing giving the full address of that other place.
- (c) The Company shall preserve in good order the books of accounts relating to a period of not less than eight years preceding the current year together with the vouchers relevant to entries in such books of account.
- (d) When the Company has a branch office, whether in or outside India, the Company shall be deemed to have complied with this Article 205 if proper books of account relating to the transactions effected at the branch office are kept at the branch office and proper summarized returns made up to dates at intervals of not more than three months, are sent by the branch office to the Company at the registered office or other place in India, at which the Company's books of account are kept as aforesaid.
- (e) The books of account shall give a true and fair view of the state of the affairs of the Company.
- (f) The books of account shall be open for inspection by any Director during business hours. No Member (not being a Director) shall have any right to inspect any books of account or books and papers or documents of the Company except as conferred by law or authorised by the Board.
- 206. The Board shall, from time to time, in accordance with the provisions of the Act, cause to be prepared and to be laid at each Annual General Meeting a profit and loss account and a balance sheet, containing a summary of the property and assets and of the Capital and liabilities of the Company, made up to a date not earlier than the date of the Meeting by more than six months or such extended period as may be permitted under the Act by the Registrar of Companies.
- 207. Every balance sheet and profit and loss account of the Company shall give a true and fair view of the affairs and the profit or loss of the Company for the financial year and shall comply with the requirements of Schedule III of the Act, so far as they are applicable thereto.

- 208. (a) Every balance sheet laid before the Company in Annual General Meeting shall be accompanied by a report of the Board of Directors as to the state of the Company's affairs and as to the amounts, if any, which it recommends should be paid by way of Dividend and material changes and commitments, if any, affecting the financial position of the Company for which the balance sheet relates and the date of the report.
 - (b) The Board's report shall so far as is material for the appreciation of the state of the Company's affairs by its Members and which will not, in the Board's opinion, be harmful to the business of the Company, deal with any change which have occurred during the Financial Year in the nature of the Company's business and generally in the classes of business in which the Company has an interest.
 - (c) The Board shall also give the fullest information and explanations in its Report aforesaid, or in an addendum to that report on every reservation, qualification or adverse remark contained in the Auditor's Report.
 - (d) The Board's Report and any addendum thereto shall be signed by its Chairman if he is authorized in that behalf by the Board, and where he is not so authorized, shall be signed by such number of Directors as are required to sign the balance sheet and the profit and loss account of the Company.
 - (e) The Board shall have the right to charge any Person not being a Director with the duty of seeing that the provisions of Articles 208(a) to 208(c) are complied with.
- 209. (a) The profit and loss account and balance sheet shall be signed on behalf of the Board of Directors by the Manager or Secretary, if any, and by not less than two Directors, one of whom shall be a Managing Director where there is one, provided that if there is only one Director present in India at the time the profit and loss account and balance sheet shall be signed by such Director but in such a case there shall be attached to the Profit and Loss Account and Balance Sheet a statement signed by such Director explaining the reason for non-compliance with the aforesaid provision requiring the signatures of two Directors.
 - (b) The profit and loss account and balance sheet shall be audited by the Auditor and the Auditor's Report (including the Auditors separate, special or supplementary report, if any) shall be attached thereto, and such Report shall be read before the Company in General Meeting and shall be open to inspection by any Member.
- 210. The Board of Directors shall from time to time determine whether and to what extent and at what times and places and under what conditions or regulations the accounts, books and documents of the Company or any of them, shall be open to the inspection of the members, and no members (not being a Director) shall have any right of inspecting any account or books or documents of the Company except as conferred by statute or authorized by the Directors or by a resolution of the Company in General Meeting.
- 211. A copy of every balance sheet including the profit and loss account, the Auditors' Report and every other document required by law to be annexed or attached as the case may be, to the balance sheet, which is to be laid before the Company in a General Meeting, shall be made available for inspection at the Registered Office, of the Company during working hours for a period of twenty one days before the date of the Meeting.
- 212. Subject to provisions of the Act and the Rules, where the Company has one or more subsidiaries, it shall, in addition to the financial statements to be prepared in accordance with these Articles, prepare a consolidated financial statement of the company and of all the subsidiaries in the same form and manner as that of its own.

213. A statement containing the salient features of such documents in the prescribed form or the copies of the documents aforesaid, as the Company may deem fit, will be sent to every member of the Company and to every trustee for the holders of any Debentures issued by the Company not less than twenty one days before the date of the meeting as laid down in Section 136 of the Act and all other provisions of this Section shall apply in respect of the matters referred to in this Article 213.

DOCUMENTS AND NOTICES

- A document or notice may be served or given by the Company on any Member or officer of the Company either personally or by sending it by post to him to his registered address or (if he has no registered address in India) to the address, if any, in India supplied by him to the Company for serving documents or notices in him or by electronic mail.
 - (b) Where a document or notice is sent by post, service of the documents or notice shall be deemed to be effected by properly addressing, prepaying and posting a letter containing the document or notice. Provided that where a Member has intimated to the Company in advance that documents or notices should be sent to him under a certificate of posting or by registered post with or without acknowledgment due and has deposited with the Company a sum, sufficient to defray the expenses of doing so, service of the document or notice shall not be deemed to be effected unless, it is sent in the manner intimated by the Member, and such service shall be deemed to have been effected in the case of a notice of a Meeting, at the expiration of forty eight hours after the letter containing the documents or notice is posted, and in any other case, at the time at which the letter would be delivered in the ordinary course of post.
 - (c) Where a document or notice is sent by electronic mail, the document or notice shall be deemed to have been delivered upon an electronic mail containing the document or notice being sent to the email address provided to the Company by the Member.
- 215. A document or notice advertised in a newspaper circulating in the neighbourhood of the Office shall be deemed to be duly served or sent on the day on which the advertisement appears, on every Member who has no registered address in India or has not supplied to the Company an address within India for the serving of document on or the sending of notices to him.
- 216. A document or notice may be served or given by the Company on or to the joint-holders of a Share / Debenture by serving or giving the document or notice on or to the joint-holder named first in the Register of Members/ Register of Debenture-holders in respect of the Share/Debenture.
- 217. A document or notice may be served or given by the Company on or to the Persons entitled to a Share in consequence of death, insolvency or winding up of a Member by sending it through the post in a prepaid letter addressed to them by name or by the title of representatives of deceased, official assignee, receiver or liquidator of the Member in insolvency or winding-up or by any like description, at the address, if any, in India, supplied for the purpose by the Persons claiming to be so entitled, or (until such an address has been so supplied) by serving the document or notice in any manner in which the same might have been given if the death, insolvency or winding up had not occurred.
- 218. Documents of every General Meeting shall be served or given in the same manner hereinbefore authorized on or to every Member and to the Auditor or Auditors for the time being of the Company; and shall be served in the manner provided in Article 214 on every Person entitled to a share in consequence of the death, insolvency or winding up of a Member.

- 219. Every Person who, by operation of law, transfer or other means whatsoever, shall become entitled to any share, shall be bound by every document or notice in respect of such share, which previously to his name and address being entered on the register of Members, shall have been duly served on or given to the Person from whom he derives his title to such share.
- 220. Any document or notice to be served or given by the Company may be signed by any Director, Secretary or some Person duly authorized by the Board of Directors for such purpose and the signature may be written, printed or lithographed.
- 221. All documents or notice to be served or given by Members on or to the Company or any Officer thereof shall be served or given by sending the same to the Company or officer at the Registered Office by post under a certificate of posting or by registered post, or by leaving the same at its Registered Office.
- 222. A Notice may be served on the Registrar by sending it to him at his office by post under a certificate of posting or by registered post, or by delivering it to, or leaving it for him at his office.

WINDING UP

- 223. (a) Subject to the provisions of the Companies Act, 1956 or the Act, as applicable, and these Articles, if the Company shall be wound up, the liquidator, may with the sanction of a special resolution of the Company and any other sanction required by the Act, divide amongst the members in specie or kind the whole or any part of the assets of the Company, whether they shall consist of property of the same kind or not.
- (b) For the purpose of aforesaid, the liquidator may set such value as he deems fair upon any property to be divided as aforesaid and may be determined how such division shall be carried out as between the Members or different classes of Members.
 - (c) The liquidator may, with the like sanction, vest the whole or any part of such assets in trustees upon such trust for the benefit of the contributors as the liquidator, with the like sanction shall think fit, but so that no member shall be compelled to accept any Shares of other securities whereon there is any liability.

INDEMNITY AND RESPONSIBILITY

- Save and except so far as the provisions of this Article 224 shall be avoided by Section 197(13) of the Act, the Board of Directors, Managing Director, Managers, Secretary, and other officers or other employees/servants for the time being of the any, Auditor and the trustees, if any, for the time being acting in relation to any of the affairs of the Company and every one of them and every one of their heirs, executors and administrators shall be indemnified and secured harmless out of the assets and profits of the Company from and against all actions, costs, charges, losses, damages and expenses which they or any of them, their or any of their executors or administrators shall or may incur or sustain by or reason of any act done, concurred in or omitted in or about the execution of their duty, or supposed duty, in their respective offices or trust, except such, if any as they shall incur or sustain through or by their own wilful neglect or default respectively.
- (b) None of them shall be answerable for the acts, receipts, neglects or defaults of the other or others of them, or for joining in any receipt for the sake of conformity, or for any bankers or other Person with whom any moneys or effects belonging to the Company shall or may be lodged or deposited for safe custody, or for the insufficiency or deficiency of any security upon which any moneys of or belonging to the Company

shall be placed out or invested, or for any other loss, misfortune or damage which may happen in the execution of their respective offices or trusts or in relation thereto, except the same shall happen by or through their own wilful dishonesty, neglect or default respectively.

SECRECY

- 225. (a) Every Director, Manager, Auditor, treasurer, trustee, Member of a Committee, officer, servant, agent, accountant or any other Person employed in the company shall, if so required by the Directors, or by any other Person authorized in this behalf before entering upon his duties, sign a declaration pledging himself to observe strict secrecy respecting all transactions and affairs of the Company with the customers and the state of the accounts with individuals and in matters relating thereto, and shall by such declaration pledge himself not to reveal any of the matters which may come to his knowledge in discharge of his duties except when required to do so by the Directors or by law or by the Person to whom such matters relate and except so far as may be necessary in order to comply with any of the provisions in these presents contained.
 - (b) Subject as aforesaid, every Director, Managing Director, Manager, Company Secretary or other officer of the Company shall be indemnified against any liability incurred by him in defending any proceedings, whether civil or criminal in which judgment is given in his favour or in which he is acquitted or discharged or in connection with any application under applicable provisions of the Act in which relief is given to him by the Court.
 - (c) Subject to the provisions of the Act, no member shall be entitled to visit or inspect any works of the Company without the permission of the Director or to require discovery of or any information respecting any detail of the Company's trading or any matter which is or may be in the nature of a trade secret, mystery of trade, secret process, or any other matter which may be relate to the conduct of the business of the Company and which in the opinion of the Directors, would be inexpedient in the interest of the Company to disclose.
 - (d) The Company may take and maintain any insurance as the Board may think fit on behalf of its present and/or former Directors and Key Managerial Personnel for indemnifying all or any of them against any liability for any acts in relation to the Company for which they may be liable but have acted honestly and reasonably.

GENERAL POWER

226. Wherever in the Act, it has been provided that the Company shall have right, privilege or authority or that the Company could carry out any transaction only if the Company is so authorized by its articles, then and in that case this regulation hereto authorizes and empowers the Company to have such rights, privilege or authority and to carry such transactions as have been permitted by the Act, without there being any specific regulation in that behalf herein provided.

COMPLIANCE WITH THE LICENCE AGREEMENTS

227. The Company will adhere and comply with the conditions and terms stipulated in the Licence Agreements. Under extant requirements of the DoT, any violation of any such License Agreements shall automatically lead to the Company being unable to carry on its business in this regard.

Part II

1. OVERRIDING EFFECT

In the event of any conflict between Part I of the Articles of Association and Part II of the Articles of Association, the provisions of Part II of the Articles of Association shall prevail.

2. DEFINITIONS & INTERPRETATION

2.1 **Definitions**

Unless the context otherwise requires, the following words and terms shall have the meanings set forth below:

"Acceptance Notice" shall have the meaning given to it in Article 13.3.3;

"Act" means the Indian Companies Act, 2013 and shall include the provisions of the Indian Companies Act, 1956, to the extent the corresponding provision in the Indian Companies Act, 2013 has not been notified:

"Affiliate" of a Person means any other Person that directly or indirectly, through one or more intermediaries, (a) owns greater than 26% of the voting equity or interest of such Person or is similarly owned by such Person; and (b) Controls, is Controlled by, or is under common Control with, such first Person, and in the case of a natural Person, shall include his or her Relatives;

"Agreed Shared Costs" shall have the meaning as agreed, *inter alia*, among the Parties in the Implementation Agreement;

"Big Four Accounting Firm" shall mean any of (i) KPMG, (ii) Deloitte Touche Tohmatsu Limited (iii) Ernst and Young LLP, or (iv) PricewaterhouseCoopers, or any of their Indian associates and affiliates.

"Board" means the board of directors of the Company constituted in accordance with the Articles of Association from time to time;

"Books and Records" means all accounting, financial reporting, tax, business, marketing and corporate files, documents, instruments, papers, books, registers and records (statutory or otherwise) of the Company and its Subsidiaries, including technical records, financial statements, journals, deeds, manuals, minute books, customer and client lists, reports, files, documents, electronic information and operating data, contracts, memoranda of understanding and agreements, in whatever form;

"Business" means the provision of fixed and mobile telecommunications services to consumer and enterprise customers, including direct-to-consumer video and content services that are bundled with telecommunications services by the Company and its Subsidiaries in the Territory, and subject to amendment in accordance with Article 10 (*Reserved Matters*), any other business carried on by the Company and its Subsidiaries;

[#] Sections of Part II have been amended pursuant to a special resolution passed by the members of the Company at the Annual General Meeting held on August 27, 2019.

"Business Day" means a day other than Saturday and Sunday on which banks are open for normal banking business in London, United Kingdom, Mauritius, the Netherlands and Mumbai, India;

"Business Plan" means the detailed operating budget and the financial and strategic plan of the Company as prepared, approved and amended from time to time in accordance with the Articles of Association;

"Call Option 1" shall have the meaning given to it in Article 12.3.1(a);

"Call Option 1 Equity Share Value" means the Call Option 1 Equity Value divided by the sum of:

- (a) the number of Equity Shares of the Company (on a fully diluted basis) as on the Effective Date:
- (b) (without double counting) the number of Equity Shares issued pursuant to all Rights Recapitalisations occurring under Article 4; and
- (c) the number of Equity Shares issued as bonus shares with respect to any of the Equity Shares falling within (a) or (b) above,

and adjusted in customary manner for any split or reverse-split made with respect to such Equity Shares on or after the date on which they were issued;

"Call Option 1 Equity Value" means, in relation to a Call Option 1 Notice, the amount that is equal to the sum of:

- (a) Rs.945,524 million; and
- (b) (without double counting) the aggregate of value of all gross consideration received or receivable by the Company pursuant to all Rights Recapitalisations occurring under Article 4 and before the date of that Call Option 1 Notice;

"Call Option 1 Notice" shall have the meaning given to it in Article 12.3.1(c);

"Call Option 1 Period" means a period of 36 (thirty six) months and one (1) Business Day from the Effective Date;

"Call Option 1 Price" shall have the meaning given to it in Article 12.3.1(c);

"Call Option 1 Purchaser" shall have the meaning given to it in Article 12.3.1(c);

"Call Option 1 Shares" shall have the meaning given to it in Article 12.3.1(c);

"Call Option 2" shall have the meaning given to it in Article 12.3.2(a);

"Call Option 2 Notice" shall have the meaning given to it in Article 12.3.2(d);

"Call Option 2 Period" shall have the meaning given to it in Article 12.3.2(a);

- "Call Option 2 Price" shall have the meaning given to it in Article 12.3.2(e);
- "Call Option 2 Purchaser" shall have the meaning given to it in Article 12.3.2(d);
- "Call Option 2 Shares" shall have the meaning given to it in Article 12.3.2(b);
- "Call Option Cap" means at any specified time, the number of Equity Shares that is equal to 50% of the Excess Equity Shares at such time (rounded down to the nearest whole Equity Share) (it being acknowledged that, at the end of the Effective Date, the Call Option Cap will not exceed 10.75% of the Share Capital);
- "Capped Options" means Call Option 1, Call Option 2, Step Down Option 1 and the Rights Recapitalisation Call Option;
- "CEO" means the chief executive officer of the Company, appointed from time to time in accordance with the Articles of Association;
- "CFO" means the chief financial officer of the Company, appointed from time to time in accordance with the Articles of Association;
- "Chairperson" shall have the meaning given to it in Article 5.7.1;
- "Charged Rights" shall have the meaning given to it in Article 13A;
- "Charged Shares" shall have the meaning given to it in Article 13A;
- "Circular Resolution" shall have the meaning given to it in Article 5.8.1;
- "CoC Exercise Notice" shall have the meaning given to it in Article 16.2.1;
- "CoC Notice" shall have the meaning given to it in Article 16.1;
- "CoC Shareholder" shall have the meaning given to it in Article 16.1;
- "CoC Shares" shall have the meaning given to it in Article 16.2.1;
- "Committee" shall have the meaning given to it in Article 5.4.1;
- "Company" means Vodafone Idea Limited, a company incorporated under the laws of India having its registered office at Suman Tower, Plot No. 18, Sector 11, Gandhinagar, Gujarat 382011, India;
- "Competing Business" means a business in the Territory that is the same as or substantially similar to the Business:
- "COO" means the chief operating officer of the Company, appointed from time to time in accordance with the Articles of Association:

"Control" (including with correlative meaning, the terms "Controlled by" and "under common Control" with) means the right to appoint the majority of the directors or to control the management or policy decisions of a Person, exercisable by a Person or Persons acting individually or in concert, directly or indirectly, including by virtue of their shareholding or management rights or shareholders agreements or voting agreements or in any other manner;

"Deadlock" shall have the meaning given to it in Article 14.1;

"Deadlock Notice" shall have the meaning given to it in Article 14.2;

"Deed of Adherence" means the deed of adherence set out in the Shareholders' Agreement;

"**Defaulting Promoter Group**" shall have the meaning given to it in Article 15.1.1;

"Defaulting Shareholder Group" shall have the meaning given to it in Article 15.2.1;

"Diluted Group" shall have the meaning given to it in Article 4.8.1;

"Directors" mean the members of the Board appointed in accordance with the Articles of Association;

"Draft Revised Business Plan" shall have the meaning given to it in Article 11.1;

"Dynamo Agreement" means the agreement between Vodafone Plc and Dynamo Finance Designated Activity Company dated 2 February 2019 (as amended from time to time);

"EBITDA" means the consolidated profit before tax of the Company as per the Financial Statements for that relevant period after adding back:

- (a) any amount attributable to amortisation of intangible assets and goodwill, and depreciation of tangible assets;
- (b) Finance Charges;
- (c) items treated as exceptional;
- (d) Integration Costs; and
- (e) Agreed Shared Costs,

in each case, to the extent added, deducted or taken into account, as the case may be, in determining the consolidated profit before tax of the Company as per the relevant Financial Statements;

"Effective Date" means 31 August 2018;

"Encumbrance" (including with correlative meaning, "Encumber") means, (i) any mortgage, charge (whether fixed or floating), pledge, lien, hypothecation, assignment, deed of trust, title retention, security interest or other encumbrance of any kind securing, or conferring any priority of payment in respect of, any obligation of any Person, or other agreement or arrangement having

a similar effect (including any sale and leaseback agreement or arrangement and any sale and repurchase agreement or arrangement), including any right granted by a transaction which, in legal terms, is not the granting of security but which has an economic or financial effect similar to the granting of security under applicable Law, (ii) any proxy for exercising voting rights issued to any third party, power of attorney issued to any third party for transferring and/or exercising any rights, voting trust agreement, interest, option, right of first offer, refusal or transfer restriction in favour of any Person, and (iii) any adverse claim as to title, possession or use;

"Enforced Entity" shall have the meaning given to it in Article 13.2.3A;

"Enforcement Event" shall have the meaning given to it in the Dynamo Agreement (and any material amendment to the definition of "Enforcement Event" in the Dynamo Agreement after 2 February 2019 which adversely affects the ICL Group Shareholders shall require the prior written consent of the ICL Group Shareholders, such consent not to be unreasonably withheld, conditioned or delayed);

"Equal Offer Notice" shall have the meaning given to it in Article 12.6.3(a);

"Equal Offer Period" shall have the meaning given to it in Article 12.6.3(b);

"**Equal Shareholding Date**" means the first date on which the number of Excess Equity Shares becomes zero;

"Equity Shares" means fully-paid up equity shares issued from time to time forming part of the Share Capital;

"Event of Default" shall have the meaning given to it in Article 15.2.1;

"Excess Equity Shares" means, at any specified time and subject to Articles 4.6, 12.1.1, and 12.3.2(i), the number of Equity Shares that is equal to the greater of:

- (a) zero; and
- (b) (i) the Shareholding of the Vodafone Group Shareholders at such time minus (ii) the Shareholding of the ICL Group Shareholders at such time;

"Excluded Financial Investor" means any Financial Investor:

- (a) where 33% or more of that Financial Investor's assets under management comprise an equity holding in a single Person that conducts a business that is similar to the Business within or outside the Territory; or
- (b) whose Investment Manager is Controlled by a Person who conducts a business that is similar to the Business within or outside the Territory;

"Extended RCO Period" shall have the meaning given to it in Article 4.7.1(c);

"Facility Agreement" means the facility agreement between inter alios Dynamo Finance Designated Activity Company and HSBC Bank plc as agent, among others, dated 2 February 2019 (as amended from time to time);

"Fair Market Value" means the Volume Weighted Average Market Price for a period of three (3) months preceding the Relevant Date, as traded on the Recognised Stock Exchange where the maximum volume of trading in the Equity Shares of the Company is recorded during the three-month period prior to the Relevant Date;

"Finance Charges" means, for any relevant period, the aggregate amount of interest, commission, fees, discounts, prepayment penalties or premiums, Forex Losses or Gains (if net losses) and other finance payments in respect of Financial Indebtedness whether accrued, paid or payable in respect of that relevant period, net of any treasury income (representing income from investing surplus cash in securities as per the treasury policy of the Company), or interest or similar income and Forex Losses or Gains (if net gains) whether accrued, received or receivable, and:

- (a) including the interest element of leasing and hire purchase payments;
- (b) including the mark to market gains or losses, whether realised or unrealised, on foreign exchange rate and interest rate derivative financial instruments; and
- (c) including any amounts in the nature of interest payable in respect of any shares other than ordinary equity share capital;

"Financial Indebtedness" means any borrowings or indebtedness for or in respect of:

- (a) moneys borrowed;
- (b) accrued interest payable;
- (c) any interest bearing amount raised by acceptance under any acceptance credit, bill acceptance or bill endorsement facility or dematerialised equivalent;
- (d) any amount raised pursuant to any note purchase facility or the issue of bonds, notes, debentures, loan stock or any similar instrument;
- (e) the amount of any liability in respect of any finance lease;
- (f) receivables sold or discounted (other than any receivables to the extent they are sold on a non-recourse basis);
- (g) any amount raised under any other transaction (including any forward sale or purchase agreement) having the commercial effect of a borrowing under Ind AS;
- (h) any derivative transaction entered into in connection with protection against or benefit from fluctuation in any rate or price (and, when calculating the value of any derivative transaction, only the marked to market value shall be taken into account); and
- (i) shares which are expressed to be redeemable or shares or instruments convertible into shares (other than compulsorily convertible instruments),

provided in each case that there shall be no double-counting of any indebtedness;

"Financial Investor" means any organisation (including banks, insurance companies, hedge funds, endowment funds, pension funds, sovereign wealth funds and other financial institutions) engaged in the business of holding and managing assets (including securities) or wealth management, for and on behalf of its clients, other than an Excluded Financial Investor;

"Financial Statements" means in relation to the Company the consolidated quarterly financial statements of the Company and its Subsidiaries prepared under Ind AS;

"Financial Year" means the Company's fiscal year beginning on 1 April of each calendar year and ending on 31 March of the immediately succeeding calendar year, or such other period as the Board or the Shareholders, as the case may be, determine in accordance with applicable Law;

- "Financier" means any Agent, Security Agent or Lender (as such terms are defined in the Facility Agreement) under the Facility Agreement from time to time and any agent, attorney, delegate, receiver or trustee of or for such Agent, Security Agent or Lender;
- "First Refusal Right" shall have the meaning given to it in Article 13.3.1;
- "Forex Losses or Gains" means the net foreign exchange gains or losses with respect to Financial Indebtedness denominated in currency other than INR;
- "General Meeting" shall have the meaning given to it in Article 6.1;
- "GIL" shall have the meaning given to it in Article 9.2.1(b);
- "Governmental Authority" means any national, regional or local government or governmental, administrative, fiscal, judicial, or government-owned body of any nation or any of its ministries, departments, secretariats, agencies or any legislative body, commission, authority, court or tribunal or entity, and shall include any authority exercising jurisdiction over any Person;
- "Group" means, (i) the ICL Group and/or the Vodafone Group, as the context may require, (ii) in relation to the Company, the Company and its Subsidiaries, and (iii) in relation to any other company, means that company and any Affiliate of that company;
- "Higher Number" shall have the meaning given to it in Article 4.3.2(ii)(A);
- "Higher Shareholder" shall have the meaning given to it in Article 12.6.3;
- "ICL Bank" shall have the meaning given to it in Article 4.3.2(ii);
- "ICL CoC Period" shall have the meaning given to it in Article 16.2.1;
- "ICL CoC Price" means an amount equal to the Vodafone Purchase Price;
- "ICL Confirmation Notice" shall have the meaning given to it in Article 12.2.2(b);
- "ICL Financier" means any agent, security agent or lender under and ICL Financing Agreement;
- "ICL Financing Arrangement" means an arrangement between (i) KMB and/ or any ICL Group Shareholder and (ii) any lender or group of lenders, the purpose of which is to finance, from 20 March 2017, the participation of any ICL Group Shareholder in a Rights Recapitalisation or other equity fund raising by the Company or any acquisitions of Equity Shares by KMB and/or any ICL Group Shareholder from any Vodafone Group Shareholder;
- "ICL Group" means the ICL Group Shareholders and their respective Affiliates, excluding the Company and its Subsidiaries;
- "ICL Group Directors" shall have the meaning given to it in Article 5.2.2(a);
- "ICL Group Shareholders" shall mean (i) Grasim Industries Limited, (ii) Pilani Investments and Industries Limited, (iii) Hindalco Industries Limited, (iv) Birla TMT Holdings Private

Limited (v) Elaine Investments Pte Ltd. (vi) Oriana Investments Pte Ltd. and (vii) IGH Holdings Private Limited, together with any Affiliates that execute a Deed of Adherence;

- "ICL Opposition Notice" shall have the meaning given to it in Article 12.2.1(a);
- "ICL Surplus Shareholding" means the Equity Shares representing the following percentage of Share Capital: (i) the percentage Shareholding of the ICL Group Shareholders at the end of the Effective Date (on a fully diluted basis) minus (ii) 26.0%;
- "Implementation Agreement" means the Implementation Agreement dated 20 March 2017, as amended, among, inter alia, the ICL Group Shareholders, the Vodafone Group Shareholders, the Company and the Vodafone Confirming Party;
- "Ind AS" means Indian Accounting Standards as notified by Ministry of Corporate Affairs, Government of India;
- "Indian Competitor" means: (a) any Person, including its Affiliates, engaged in a Competing Business; or (b) any Person who holds, directly and/or indirectly through its Affiliates, 26% (twenty six percent) or more in, and is categorised as a promoter of, a Person referred to in (a) above;
- "Integration Costs" means costs incurred on or after the Effective Date in connection with the combination of the Company and Vodafone India Limited as agreed, *inter alia*, among the Parties in the Implementation Agreement, which would not have been incurred otherwise;
- "Intellectual Property" means all domestic and foreign intellectual property rights, including with respect to all patents, patent applications, and trademarks, service marks, trade names, trade dress, logos, corporate names, brand names, domain names, all copyrights, designs and mask works, and all registrations, applications and renewals in connection therewith, and software and all website content (including text, graphics, images, audio, video and data) and trade secrets, confidential business information and other proprietary information;
- "Investment Bank" means a Category I merchant banker registered with the Securities and Exchange Board of India;
- "Investment Manager" in respect of any Person, means any general partner, investment manager or other person who controls the investment decisions of such Person;
- "KMB" means Kumar Mangalam Birla, an individual residing in India;
- "Law" means any statute, law, ordinance, rule, regulation, press note, notification, circular, order, writ, injunction, directive, judgment or decree issued by any Governmental Authority;
- "Leverage Ratio" means, at any time, the ratio of the Net Financial Debt to LTM EBITDA, each of which shall have been determined with reference to the same time:
- "Leverage Ratio Trigger" is met in the case where the then current Leverage Ratio is above:
- (a) 6:1 in the financial year ended 31 March 2018;

- (b) 5.75:1 in the first quarter of the financial year ended 31 March 2019;
- (c) 5.5:1 in the second quarter of the financial year ended 31 March 2019;
- (d) 5.25:1 in the third quarter of the financial year ended 31 March 2019; or
- (e) 5:1 in the fourth quarter of the financial year ended 31 March 2019 or at any time thereafter;

"Lower Number" shall have the meaning given to it in Article 4.3.2(ii)(A);

"LTM EBITDA" means, at any time, the EBITDA (by reference to the Financial Statements) for the 12 (twelve) months up to the end of the most recent calendar quarter ended 31 March, 30 June, 30 September or 31 December. Where LTM EBITDA requires EBITDA to be determined for periods prior to the Effective Date, EBITDA for these periods shall be taken from the Financial Statements and the Vodafone Financial Statements and aggregated;

"Net Assets" means, at any time in relation to a Person, the aggregate of its assets (excluding intangible assets) less the aggregate of its liabilities (other than share capital and reserves, and provisions against intangible assets), in each case calculated on a consolidated basis in accordance with applicable accounting standards;

"Net Assets Threshold" means Rs.167,375 million;

"Net Financial Debt" means, at any time, the aggregate amount of all obligations of the Company for or in respect of Financial Indebtedness at that time but:

- (a) deducting the aggregate amount of cash and cash equivalent investments held by the Company at that time; and
- (b) deducting the aggregate amount of interest receivable by the Company at that time,

and so that no amount shall be included or excluded more than once;

"New Qualifying Shareholder" shall have the meaning given to it in Article 13.2.3 or 13.2.3A, as applicable;

"Non-Diluted Group" shall have the meaning given to it in Article 4.8.1;

"Non-Equal Shareholder" shall have the meaning given to it in Article 12.6.3;

"Non-transferring Shareholder" shall have the meaning given to it in Article 13.3.2;

"Offered Shares" shall have the meaning given to it in Article 13.3.2;

"Offer Period" shall have the meaning given to it in Article 13.3.3;

"Option Transfer" shall have the meaning given to it in Article 12.7.1;

"Party" means any of the ICL Group Shareholders, the Vodafone Group Shareholders and the Company in its capacity as a party to the Shareholders' Agreement;

"Permitted Security" shall have the meaning given to it in Article 13A;

"Person" means any individual, general or limited partnership, corporation, limited liability company, joint stock company, trust, joint venture, unincorporated organisation, association or any other entity, including any Governmental Authority, or any group consisting of two (2) or more of the foregoing;

"Promoter Group" means the Vodafone Group Shareholders collectively and/or the ICL Group Shareholders collectively, as the context may require;

"Proposed Transferee" shall have the meaning given to it in Article 13.3.1;

"Public Shareholder" means any Person holding Public Shareholding;

"Public Shareholding" means, with respect to the Company, its public shareholding (as defined under rule 2(e) of the Securities Contracts (Regulation) Rules, 1957);

S"Qualifying Threshold" means 13% of the Share Capital, provided that, solely for purposes of calculation of the percentage of the Qualifying Threshold and determination of whether any Shareholding meets the Qualifying Threshold, any outstanding employee stock options granted by the Company after the Effective Date shall not be taken into account and the effect of such options on the Share Capital shall be disregarded;

"RCO Notice" shall have the meaning given to it in Article 4.7.1(b);

"RCO Period" shall have the meaning given to it in Article 4.7.1;

"RCO Price" means, if the Vodafone Group Shareholders elect to Transfer Equity Shares: (i) on a Recognised Stock Exchange, the higher of the maximum price per Equity Share permitted under Law for such Transfer without the approval of any Governmental Authority (or if applicable Law does not prescribe a price, the per share Fair Market Value) and the subscription price per share for the Rights Recapitalisation; and (ii) in an off-exchange Transfer, the higher of the maximum price per Equity Share permitted under Law for such Transfer without the approval of any Governmental Authority (or if applicable Law does not prescribe a price, the per share Fair Market Value) and the subscription price per share for the Rights Recapitalisation;

"RCO Purchaser" shall have the meaning given to it in Article 4.7.1(b);

"RCO Shares" shall have the meaning given to it in Article 4.7.1;

"RCO Withholding Computation" shall have the meaning given to it in Article 4.7.1(g);

"Recognised Stock Exchange" means any stock exchange where the Equity Shares are listed;

⁵ The definition of Qualifying Threshold has been altered pursuant to Special Resolution passed by the members of the Company at the Extra-ordinary General Meeting held on 26th March 2022

- "Relative" with respect to a natural Person, shall have the meaning given to the term in the Act;
- "Relevant Date" means, for the purpose of determination of the Fair Market Value in: (i) Article 4.7.1(b), the date of the RCO Notice; (ii) Article 16, the date of the CoC Notice; and (iii) the context of Relevant India Telecom Equity Value, the earlier of the date of public announcement of and the date of execution of binding documentation for a Vodafone Permitted Group Sale Disposal or a Vodafone Restricted Group Sale Disposal, as applicable;
- "Relevant Holdco Equity Value" means, in relation to: (i) item (a) of the definitions of Vodafone Permitted Group Sale Disposal and Vodafone Restricted Group Sale Disposal, the aggregate proportionate equity value of the entities or assets proposed to be included within such transaction, as derived from the agreed consideration for such transaction; and (ii) item (b) of the definitions of Vodafone Permitted Group Sale Disposal and Vodafone Restricted Group Sale Disposal, the aggregate proportionate equity value of the entities proposed to be included within such transaction as derived from a valuation opinion prepared by any one of the Persons specified in **Schedule 1** selected by the Vodafone Group by a draw of lots in the presence of an authorised representative of the ICL Group Shareholders;
- "Relevant India Telecom Equity Value" means, with respect to Shareholder(s) proposed to be included within a Vodafone Permitted Group Sale Disposal or a Vodafone Restricted Group Sale Disposal, the equity value of the Company based on the Fair Market Value multiplied by the percentage Shareholding of such Shareholder(s);
- "Remaining ICL Shareholders" shall have the meaning given to it in Article 16.2.1;
- "Remote Participation" shall have the meaning given to it in Article 5.9.1;
- "Representatives" means, with respect to any Person, its directors, officers, employees, consultants, agents, investment bankers, financial advisors, legal advisors, accountants, other advisors and authorised representatives;
- "Reserved Matters" has the meaning given to it in Article 10;
- "Rights Cure Period" shall have the meaning given to it in Article 17.3.2;
- "Rights Recapitalisation" shall have the meaning given to it in Article 4.3;
- "Rights Recapitalisation Call Option" shall have the meaning given to it in Article 4.7.1;
- "Rights Recapitalisation Cap" means with respect to any Rights Recapitalisation, the lower of:
 (a) the number of new Equity Shares subscribed for by the Vodafone Group Shareholders in excess of the new Equity Shares to which they are entitled and (b) the number of new Equity Shares to which the ICL Group Shareholders were entitled under the Rights Recapitalisation but for which they did not subscribe;
- "Rights Recapitalisation Notice" shall have the meaning given to it in Article 4.3;
- "Shareholder" means any Person who holds Equity Shares in the Company;

"Shareholding" means, with respect to:

- (a) any Person as a Shareholder, at any time, that Person's total direct and indirect shareholding in the Company; and
- (b) a group of Persons directly and indirectly holding shares in the Company, the aggregate of the total direct and indirect shareholding of each Person in the group in the Company without any duplication or double counting of shareholdings among such Persons,

in each case, on a fully diluted basis (except that solely for purposes of determination of whether any Shareholding meets the Qualifying Threshold, any outstanding employee stock options granted by the Company after the Effective Date shall not be taken into account and the effect of such options shall be disregarded), it being understood that the indirect shareholding of any such Person in the Company means the voting interest held indirectly by such Person through its subsidiaries. Shareholding shall refer to the number of Equity Shares or the percentage of Share Capital, as the context may require;

"Share Capital" means the equity share capital of the Company on a fully diluted basis. For the purposes of Article 10.4 (*Reserved Matters*), Share Capital shall mean share capital of the Company on a fully diluted basis; However, solely for purposes of calculation of the Qualifying Threshold and determination of whether any Shareholding meets the Qualifying Threshold and threshold specified in Article 9.2.2, any outstanding employee stock options granted by the Company after the Effective Date shall not be taken into account and the effect of such options on the Share Capital shall be disregarded;

"Shareholders' Agreement" means the Shareholders' Agreement dated 20 March 2017, as amended and restated on 3 May 2019, by and among the ICL Group Shareholders, the Vodafone Group Shareholders, the Company, KMB and the Vodafone Confirming Party;

"Step Down 1 Excess Shareholding" shall have the meaning given to it in Article 12.4;

"Step Down Option 1" shall have the meaning given to it in Article 12.4;

"Step Down Option 1 Period" shall have the meaning given to it in Article 12.4;

"Step Down Option 2 Period" shall have the meaning given to it in Article 12.5;

"Step Down Share Value" means, as of a particular date, the Step Down Value divided by the number of Equity Shares of the Company (on a fully diluted basis) as on the date of the Transfer of Equity Shares in accordance with Article 12.5;

"Step Down Value" means, in relation to a proposed Transfer of Equity Shares in accordance with Article 12.5, the amount that is equal to the sum of:

- (a) US\$14,123 million;
- (b) the aggregate of value of all gross consideration, whether in cash or otherwise, received or receivable by the Company in respect of each and every allotment of Equity Shares (or securities convertible into or exchangeable for Equity Shares), or grant of rights to

- subscribe for or otherwise acquire Equity Shares, in each case occurring between the date of the Shareholders' Agreement and the date of that proposed Transfer of Equity Shares;
- (c) the aggregate value of all gross consideration, whether in cash or otherwise, received or receivable by the Company and/or a Subsidiary in respect of each and every allotment of equity shares in a Subsidiary (or securities convertible into or exchangeable for equity shares in a Subsidiary), or grant of rights to subscribe for or otherwise acquire equity shares in a Subsidiary (excluding any allotment or grant to the Company or another Subsidiary that is wholly owned by the Company), in each case occurring between the date of the Shareholders' Agreement and the date of that proposed Transfer of Equity Shares,

and for the purpose of this definition: (i) any adjustment to the gross consideration received or receivable by the Company or such Subsidiary occurring or liable to occur after such allotment or grant shall be disregarded and (ii) the total amount of the maximum gross consideration receivable shall be brought into account for the purpose of this definition notwithstanding that all or any part of it is deferred or contingent);

"Subsidiary" means a subsidiary of the Company;

"Tag-Along Right" shall have the meaning given to it in Article 13.4.1;

"Tag Exercise Notice" shall have the meaning given to it in Article 13.3.3;

"Tagged Shares" shall have the meaning given to it in Article 13.4.3;

"**Takeover Code**" means the Securities and Exchange Board of India (Substantial Acquisition of Shares and Takeovers) Regulations, 2011;

"Target Group" shall have the meaning as agreed, *inter alia*, among the Parties in Implementation Agreement;

"Target Leverage Ratio" means a Leverage Ratio of: (i) 4.5:1 during the financial years ended 31 March 2018 and 31 March 2019 and (ii) 4:1 at any time thereafter;

"Tax" means any tax payable under the Indian Income-tax Act, 1961, as amended;

"Terms" shall have the meaning given to it in Article 4.3.2(i);

"Territory" means India;

"Third Banker" shall have the meaning given to it in Article 4.3.2(ii)(C);

"Third Number" shall have the meaning given to it in Article 4.3.2(ii)(C)(a);

"Transferring Shareholder" shall have the meaning given to it in Article 13.3.1;

"Transfer" means to transfer, assign, pledge or otherwise alienate or dispose of, in any way, any Equity Shares, or any rights relating to such Equity Shares, and "Transferred" shall be construed accordingly;

"Transfer Embargo" means any prohibition on the Transfer of any Equity Shares pursuant to an order of a Governmental Authority issued in respect of any Party;

"Transfer Notice" shall have the meaning given to it in Article 13.3.2;

"Trigger Event" shall have the meaning given to it in the Dynamo Agreement (and any material amendment to the definition of "Trigger Event" in the Dynamo Agreement after 2 February 2019 which adversely affects the ICL Group Shareholders shall require the prior written consent of the ICL Group Shareholders, such consent not to be unreasonably withheld, conditioned or delayed);

"Ultimate Parent" in relation to any Person, means the Person (if any) which is not itself subject to Control but which has Control of that first Person, either directly or through a chain of Persons each of which has Control over the next Person in the chain (being, as at the date of the Shareholders' Agreement, Vodafone Plc in the case of the Vodafone Group Shareholders);

"Underwriting Promoter Group" shall have the meaning given to it in Article 4.3.2(ii);

"Vodafone Bank" shall have the meaning given to it in Article 4.3.2(ii);

"Vodafone Confirmation Notice" shall have the meaning given to it in Article 12.2.1(b);

"Vodafone Confirming Party" means Vodafone International Holdings B.V., a company incorporated under the laws of The Netherlands and having its registered office at Rivium Quadrant 173, 2909 LC Capelle aan den IJssel, The Netherlands;

"Vodafone Direct Spin-off Disposal" means a demerger or spin off (effected by a solvent reconstruction or otherwise) of the entire Shareholding held by the Vodafone Group Shareholders on a *pro rata* basis to the shareholders of the Ultimate Parent;

"Vodafone Financial Statements" means the consolidated financial statements of Vodafone India Limited and its subsidiaries prepared for group reporting purposes under IFRS;

"Vodafone Group" means the Vodafone Group Shareholders and their respective Affiliates, excluding the Company and its Subsidiaries;

"Vodafone Group Directors" shall have the meaning given to it in Article 5.2.2;

"Vodafone Group Shareholders" shall mean (i) Al-Amin Investments Ltd., (ii) Asian Telecommunication Investments (Mauritius) Ltd., (iii) CCII (Mauritius) Inc, (iv) Euro Pacific Securities Ltd., (v) Mobilvest, (vi) Prime Metals Ltd., (vii) Trans Crystal Ltd., (viii) Vodafone Telecommunications (India) Ltd., (ix) Omega Telecom Holdings Private Limited, (x) Telecom Investments India Private Limited, (xi) Jaykay Finholding (India) Private Limited, and (xii) Usha Martin Telematics Limited, together with any Affiliates that execute a Deed of Adherence and in each case, a Vodafone Group Shareholder shall cease to be a Vodafone Group Shareholder upon (a) ceasing to be a member of the Vodafone Plc Group and/or (b) enforcement of Permitted Security over the Charged Shares and/or the Charged Rights of (i) such Vodafone Group Shareholder and/or (ii) a direct or indirect holding company of such Vodafone Group Shareholder;

"Vodafone Opposition Notice" shall have the meaning given to it in Article 12.2.2(a);

"Vodafone Permitted Group Sale Disposal" means: (a) a transfer of shares, voting rights, assets or any economic interest in a Vodafone Group Shareholder(s) or an entity(ies) within the chain(s) of entities between a Vodafone Group Shareholder(s) and its Ultimate Parent; or (b) a demerger or spin off (effected by a solvent reconstruction or otherwise) involving the transfer or distribution of shares in any entity within the chain(s) of entities between the Vodafone Group Shareholders and their Ultimate Parent on a *pro rata* basis to the shareholders of the Ultimate Parent, in each case, where the Relevant India Telecom Equity Value represents 33% or less of the Relevant Holdco Equity Value;

"Vodafone Permitted Transactions" shall have the meaning given to it in Article 16.6;

"Vodafone Plc" means, as at the date of the Shareholders' Agreement, Vodafone Group Plc, a company incorporated under the laws of England with its registered office at Vodafone House, The Connection, Newbury, Berkshire, RG14 2FN, and shall instead mean, if applicable in the future, any company which becomes the holding company of Vodafone Group Plc provided that:

- (a) such holding company (directly or indirectly) owns 100% of the previous Vodafone Plc's share capital (excluding any treasury shares);
- (b) such holding company is listed on a recognised stock exchange; and
- (c) the shareholders of such holding company when it becomes the holding company of the previous Vodafone Plc, include all or substantially all of the shareholders of the previous Vodafone Plc immediately prior to such event;

"Vodafone Plc Group" means Vodafone Plc and its Affiliates from time to time (excluding, for the avoidance of doubt, the Company and its Subsidiaries from time to time and the members of the ICL Group from time to time);

"Vodafone Plc Shareholder Approval" shall mean the approval (by means of an ordinary resolution) of the shareholders of Vodafone Plc at a general meeting of Vodafone Plc;

"Vodafone Purchase Price" means (a) the lower of: (i) the minimum price per Equity Share under Law for Transfers on a Recognised Stock Exchange and (ii) the minimum price per Equity Share permitted under Law for an off-exchange Transfer, in each case, without the approval of any Governmental Authority in respect of Transfers of Equity Shares by the resident to the non-resident; or (b) if applicable Law does not prescribe a price, the per share Fair Market Value;

"Vodafone Restricted Group Sale Disposal" means: (a) a transfer of shares, voting rights, assets or any economic interest in a Vodafone Group Shareholder(s) or an entity(ies) within the chain(s) of entities between a Vodafone Group Shareholder(s) and its Ultimate Parent; or (b) a demerger or spin off (effected by a solvent reconstruction or otherwise) involving the transfer or distribution of shares in any entity within the chain(s) of entities between the Vodafone Group Shareholders and their Ultimate Parent on a *pro rata* basis to the shareholders of the Ultimate Parent, in each case, where the Relevant India Telecom Equity Value represents more than 33% of the Relevant Holdco Equity Value but "Vodafone Restricted Group Sale Disposal" shall not include any transfer or distribution referred to in (a) or (b) where such transfer or distribution is as a result of the occurrence of a Trigger Event or an Enforcement Event under the Dynamo

Agreement and is effected in accordance with the terms of the Dynamo Agreement;

- "Vodafone Sale Price" means (a) the higher of (i) the maximum price per Equity Share permitted under Law for Transfers on a Recognised Stock Exchange and (ii) the maximum price per Equity Share permitted under Law for an off-exchange Transfer, in each case, without the approval of any Governmental Authority in respect of Transfers of Equity Shares by the non-resident to the resident; or (b) if applicable Law does not prescribe a price, the per share Fair Market Value;
- "Vodafone Surplus Shareholding" means the Equity Shares representing the following percentage of Share Capital: (i) the percentage Shareholding of the Vodafone Group Shareholders at the end of the Effective Date (on a fully diluted basis) minus (ii) 45.1%;
- "Volume Weighted Average Market Price" means the product of the number of Equity Shares traded on a Recognised Stock Exchange and the closing price of each Equity Share divided by the total number of Equity Shares traded on the Recognised Stock Exchange;
- "Voting Default" shall have the meaning given to it in Article 15.1.1; and
- "Withholding Computation" shall have the meaning given to it in Article 12.3.3.

2.2 Interpretation

Unless the context otherwise requires, in this Part II of the Articles of Association:

- 2.2.1 the expression "Articles" or "Articles of Association" shall mean the Articles included in this Part II;
- 2.2.2 any reference to any statute or statutory provision shall include:
 - (i) all subordinate legislation made from time to time under that provision (whether or not amended, modified, re-enacted or consolidated);
 - (ii) such provision as from time to time modified or re-enacted or consolidated and (so far as liability thereunder may exist or can arise) shall include also any past statutory provision (as from time to time modified or re-enacted or consolidated) which such provision has directly or indirectly replaced, provided that nothing in this Article 2.2.2 shall operate to increase the liability of any Party beyond that which would have existed had this Article 2.2.2 been omitted;
- 2.2.3 any reference to the singular shall include the plural and *vice-versa* and references to any gender includes the other gender;
- 2.2.4 the terms "herein", "hereof", "hereto", "hereunder" and words of similar purport refer to the Articles of Association as a whole and not to any particular provision of the Articles of Association;
- 2.2.5 any references to a "company" shall include a body corporate;
- 2.2.6 references to a document shall be a reference to that document as modified, amended, novated or replaced from time to time;
- 2.2.7 the expression "this Article" shall, unless followed by reference to a specific provision,

- be deemed to refer to the whole Article (not merely the sub-Article, paragraph or other provision) in which the expression occurs;
- 2.2.8 headings are for convenience only and shall be ignored in construing or interpreting any provision of the Articles of Association;
- 2.2.9 if the last day of any period of days specified in the Articles of Association is not a Business Day, then such period shall include the following Business Day;
- 2.2.10 a reference to a specific time for the performance of an obligation is a reference to that time in the place where that obligation is to be performed;
- 2.2.11 the words "include" and "including" shall be construed without limitation;
- 2.2.12 reference to any Person shall include that Person's successors in title and permitted assigns or transferees that have executed a Deed of Adherence in accordance with the Shareholders' Agreement;
- 2.2.13 where a wider construction is possible, the words "other" and "otherwise" shall not be construed *ejusdem generis* with any foregoing words;
- 2.2.14 any reference to any Indian legal term or concept (including for any action, remedy, judicial proceeding, document, legal status, statute, court, official governmental authority or agency) shall, in respect of any jurisdiction other than India, be interpreted to mean the nearest and most appropriate analogous term to the Indian term in the legal language in that jurisdiction as the context reasonably requires so as to produce as nearly as possible the same effect in relation to that jurisdiction as would be the case in relation to India;
- 2.2.15 any undertaking by any of the Parties not to do any act or thing will be deemed to include an undertaking not to permit or suffer or assist the doing of that act or thing (to the extent that such action or omission will be under the control or influence of the relevant Party);
- 2.2.16 where any obligation is imposed on the Company under the Articles of Association, it will be deemed that the Promoter Groups have a corresponding obligation to exercise all their powers (including voting powers) and take all necessary steps and do or cause to be done all acts, deeds and things, commissions or omissions as required to ensure compliance with such obligation of the Company;
- 2.2.17 where any obligation is imposed on any member of a Promoter Group under the Articles of Association (irrespective of whether or not such obligation on such member is independent of or in conjunction with the same obligation being placed on the Company), the members of such Promoter Group will have a corresponding obligation to cause themselves as well as each of the other members of such Promoter Group to exercise all their powers (including voting powers) and take all necessary steps and do or cause to be done all acts, deeds and things, commissions or omissions as required to ensure compliance with such obligation of the Promoter Group;
- 2.2.18 unless otherwise expressly specified in the Articles of Association, all Shareholders who are members of the same Group shall be deemed to be one (1) Shareholder and shall act together in the exercise of their rights;
- 2.2.19 unless otherwise expressly specified in the Articles of Association, the rights and

- obligations of the ICL Group Shareholders contained in the Articles of Association shall be exercised and performed jointly and severally by the ICL Group Shareholders;
- 2.2.20 unless otherwise expressly specified in the Articles of Association, the rights and obligations of the Vodafone Group Shareholders contained in the Articles of Association shall be exercised and performed jointly and severally by the Vodafone Group Shareholders;
- 2.2.21 any obligation, covenant, warranty, representation or undertaking hereto that is expressed to be made, undertaken or given by the ICL Group Shareholders will be deemed to be jointly and severally undertaken and given by each of the ICL Group Shareholders:
- 2.2.22 any obligation, covenant, warranty, representation or undertaking hereto that is expressed to be made, undertaken or given by the Vodafone Group Shareholders will be deemed to be jointly and severally undertaken and given by each of the Vodafone Group Shareholders;
- 2.2.23 notices issued in respect of options for shares shall be irrevocable except as provided in Article 12.1.1 or agreed among the Parties in writing;
- 2.2.24 a Person may exercise its votes as a Shareholder in accordance with the Articles of Association in any manner permitted by applicable Law, including at a General Meeting, through postal ballot or through e-voting;
- 2.2.25 references to "INR" or "Rs." are to Indian National Rupees;
- 2.2.26 references to "US\$" or "U.S. Dollars" are to United States Dollars;
- 2.2.27 References to "EUR" or "€" are to Euros.
- 2.2.28 "fully diluted basis" means a calculation assuming that all outstanding mandatorily convertible securities and any options issued or reserved for issuance under the employee stock option plan or any other stock option plan or scheme by whatever name called existing at the time of determination have been exercised or converted into equity shares, and equity shares under all outstanding commitments to issue equity shares or other ownership interests have been issued, in each case, as adjusted for any stock splits or any capital or other restructuring or consolidation or reduction of capital;
- 2.2.29 references to number of shares of a company and price at which any option for shares can be exercised shall be adjusted for bonus issue, reduction, reclassification, buy-back, split, sub-division or consolidation of share capital, or any similar corporate action, of such company; and
- 2.2.30 the expressions "holding company" and "subsidiary" shall have the same meanings in the Articles of Association as their respective definitions in the Act

3. ARTICLES AND OTHER MATTERS

3.1 Articles of Association

3.1.1 The rights of the Promoter Groups in the Company shall be governed by, and enforceable against each of them, in accordance with the terms of the Articles. The Promoter Groups shall perform and comply with, and pursuant to exercise of their

- voting and other rights, ensure that the Company performs and complies with, all of their respective obligations under the Articles.
- 3.1.2 The Company shall take all necessary steps to amend and alter the Articles from time to time to reflect any changes made to the Shareholders' Agreement in accordance with the terms thereof from time to time.
- 3.1.3 In the event of any ambiguity or discrepancy between the provisions of the Shareholders' Agreement and the Articles of Association, the provisions of the Shareholders' Agreement shall prevail. The Company shall take all necessary steps to amend and alter the Articles of Association from time to time to resolve any such ambiguity or discrepancy.

3.2 **Promoters**

Based on their Shareholding and rights under the Shareholders' Agreement on the Effective Date, each ICL Group Shareholder and each Vodafone Group Shareholder shall be categorised as a "promoter" of the Company.

3.3 Subsidiaries

- 3.3.1 The Company shall, and each Promoter Group shall procure that the Company shall, cause each Subsidiary to take all actions necessary to amend the articles of association of such Subsidiary to include (a) the governance provisions set forth in the Articles of Association (including with respect to board representation, quorum requirements and Reserved Matters), and (b) a provision stating that no resolution shall be adopted by the board or shareholders of such Subsidiary unless it is in compliance with the articles of association of such Subsidiary and the Shareholders' Agreement.
- 3.3.2 With respect to each Subsidiary, the Company shall procure the appointment of the maximum permissible number of directors nominated, and such number of independent directors as may be required under applicable Law from among the Persons recommended for appointment, by each Promoter Group, in the same proportion as is applicable to the constitution of the Board in Article 5.2.
- 3.3.3 If and to the extent the Promoter Groups have not exercised their respective rights with respect to nomination of directors to the boards of the Subsidiaries, the Board shall have the power to select the proposed directors of the Subsidiaries.
- 3.3.4 All resolutions to be considered by the shareholders of the Subsidiaries shall be subject to prior consideration by and approval of the Board in accordance with the Articles of Association.
- 3.3.5 The Company shall exercise its voting rights in each Subsidiary (in its capacity as a shareholder of such Subsidiary) in accordance with the Articles of Association. The Company shall vote in favour of only those resolutions which have been approved by the Board in accordance with the Articles of Association and shall vote against such resolutions which have not been so approved.

4. FUNDING

4.1 It is the intention of the Promoter Groups and the Company that the Company is self-funding and that the Company and its Group should be capable of financing their activities on a standalone basis.

- 4.2 Neither Promoter Group shall be obliged to provide any funding, whether in the form of equity or debt, to the Company or its Group, except for the purposes of Article 9.2.2 in the case of the ICL Group Shareholders.
- 4.3 If the Leverage Ratio Trigger is met, either Promoter Group may give written notice to the other Promoter Group and the Company directing the Company to implement a rights issue (a "Rights Recapitalisation", and such notice, a "Rights Recapitalisation Notice") in order to reduce the Leverage Ratio to the Target Leverage Ratio as soon as reasonably practicable. If a Rights Recapitalisation Notice is given after 31 March 2019:
 - 4.3.1 Within 15 (fifteen) Business Days of receipt of the Rights Recapitalisation Notice, each Promoter Group shall give written notice to the Company and to the other Promoter Group as to whether it is willing to underwrite the Rights Recapitalisation in proportion to its Shareholding relative to the total Shareholding.

4.3.2 If:

- (i) each Promoter Group gives such notice that it is willing to so underwrite the Rights Recapitalisation, the pricing (being a discount to the then current market price), timing and other terms of the Rights Recapitalisation (the "**Terms**") shall, subject to Article 4.3.3, be as the Promoter Groups shall agree;
- only one Promoter Group gives such notice that it is willing to so underwrite the Rights Recapitalisation (such Promoter Group being the "Underwriting Promoter Group"), the Terms shall be in accordance with the recommendation of an Investment Bank which is selected by both Promoter Groups. If the Promoter Groups are unable to agree on the selection of an Investment Bank within five (5) Business Days of the notice specified above, each Promoter Group shall promptly appoint one Investment Bank, each of whom shall determine the pricing of the Rights Recapitalisation ("ICL Bank" and "Vodafone Bank") and the following procedure shall apply:
 - (A) the higher of the prices determined by the Vodafone Bank and the ICL Bank shall be the "Higher Number" and the lower of the prices determined by the Vodafone Bank and the ICL Bank shall be the "Lower Number";
 - (B) if the Higher Number is not more than 110% of the Lower Number, the price will be the arithmetic average of such two numbers;
 - (C) if the Higher Number is more than 110% of the Lower Number, a third Investment Bank shall promptly be appointed by the Board from among the Investment Banks listed in **Schedule 1** ("**Third Banker**") by a draw of lots to determine the pricing of the Rights Recapitalisation and the price of the Rights Recapitalisation shall be:
 - (a) the Higher Number, if the price determined by the Third Banker ("Third Number") is greater than the Higher Number;
 - (b) the Lower Number, if the Third Number is less than the Lower Number;
 - (c) the arithmetic average of the Third Number and the other number (Higher Number or Lower Number) that is closer to the Third Number,

if the Third Number falls within the range between (and including) the Lower Number and the Higher Number; or

(d) the Third Number, if the Lower Number and the Higher Number are equally close to the Third Number.

Further, the Underwriting Promoter Group shall be entitled to underwrite all or part of the proportion of the Rights Recapitalisation which the other Promoter Group has not agreed to underwrite and/or procure that one or more Investment Banks underwrites all or a part of the proportion which the other Promoter Group has not agreed to underwrite on such terms as the Underwriting Promoter Group chooses; or

- (iii) neither Promoter Group gives such notice, the Rights Recapitalisation shall only proceed if there is a decision of the Board to do so, and in such case the Board shall decide the Terms, it being understood that such decision shall be a Reserved Matter.
- 4.3.3 If Article 4.3.2(i) applies and the Promoter Groups are unable to agree on the Terms within fifteen (15) Business Days of delivery of notices under Article 4.3.2(i), the Terms shall be in accordance with the recommendation of an Investment Bank which is selected by both Promoter Groups. If the Promoter Groups are unable to agree on the selection of an Investment Bank within five (5) Business Days of the expiration of the 15-Business Day period mentioned above, the Vodafone Group Shareholders shall appoint the Vodafone Bank and the ICL Group Shareholders shall appoint the ICL Bank, and the procedure set out in Article 4.3.2(ii) shall apply *mutatis mutandis*.
- 4.3.4 All costs, fees and other expenses of the Investment Bank(s) appointed pursuant to this Article 4.3 shall be borne by the Company.
- 4.4 The Company shall proceed with and promptly implement any Rights Recapitalisation in accordance with this Article 4. Regardless of whether it agrees to underwrite the Rights Recapitalisation, each Promoter Group shall take all steps necessary to procure that the Company proceeds with and promptly implements any Rights Recapitalisation in accordance with this Article 4 (except pursuant to Article 4.9).
- 4.5 To the extent required by applicable Law, any participation by the Promoter Groups in a Rights Recapitalisation shall be subject to compliance by the Company with the minimum public shareholding, if any, prescribed under applicable Law.

4.6 Calculation of Excess Equity Shares following a Rights Recapitalisation

In any Rights Recapitalisation, if the Vodafone Group Shareholders (or their nominated Affiliates) subscribe to a higher percentage of their entitlement than the ICL Group Shareholders (or their nominated Affiliates), the number of Equity Shares subscribed to by the Vodafone Group Shareholders shall, to the extent it relates to the greater relative participation of the Vodafone Group Shareholders in the Rights Recapitalisation, be excluded from the Shareholding of the Vodafone Group Shareholders for calculating the Excess Equity Shares to the extent of such higher relative participation. It is clarified that Equity Shares acquired by the ICL Group Shareholders (or their nominated Affiliates) under Article 4.7 shall be considered for the purposes of calculating the Excess Equity Shares, as per the preceding sentence.

4.7 Rights Recapitalisation Call Option prior to the Equal Shareholding Date

- 4.7.1 If, at any time prior to the Equal Shareholding Date, the percentage Shareholding of the ICL Group Shareholders is diluted pursuant to their non-participation or partial participation in a Rights Recapitalisation under this Article 4, then for a period of six (6) months from the date of completion of the relevant Rights Recapitalisation (the "RCO Period"), the ICL Group Shareholders shall have the right to acquire from the Vodafone Group Shareholders, directly or through their Affiliates, such number of Equity Shares not exceeding the Rights Recapitalisation Cap (the "RCO Shares") in the manner set forth in this Article 4.7 (the "Rights Recapitalisation Call Option"), provided that the Shareholding of the Vodafone Group Shareholders does not fall below the Qualifying Threshold pursuant to the exercise of such right:
 - (a) The Rights Recapitalisation Call Option may be exercised only once during the RCO Period.
 - (b) The ICL Group Shareholders may exercise the Rights Recapitalisation Call Option by issuing a written notice to the Vodafone Group Shareholders (the "RCO Notice"), which shall specify: (i) the identity of the purchaser(s) (the "RCO Purchaser(s)"); (ii) certification that the RCO Purchaser(s) is an ICL Group Shareholder or an Affiliate of an ICL Group Shareholder; and (iii) the number of RCO Shares.
 - (c) The Vodafone Group Shareholders shall, at their sole discretion, determine: (i) the identity of the Vodafone Group Shareholder(s) that shall Transfer the RCO Shares to the RCO Purchaser and the number of Equity Shares that each such Vodafone Group Shareholder will Transfer; (ii) whether such Transfer shall occur on a Recognised Stock Exchange or off-exchange; and (iii) the RCO Price, and shall, within five (5) Business Days of the receipt of the RCO Notice, notify the foregoing details (in writing) to the RCO Purchaser. Notwithstanding anything contained in the Articles of Association, no Vodafone Group Shareholder shall be required to Transfer any RCO Shares to the RCO Purchaser at a price less than the RCO Price. If applicable Law does not permit the RCO Purchaser to pay the RCO Price to the Vodafone Group Shareholders for the purchase of the RCO Shares during the RCO Period without the approval of a Governmental Authority, then at the option of the Vodafone Group Shareholders, (i) Clause 25.14.2 of the Shareholders Agreement shall apply or (ii) the RCO Period shall be extended for a period of three (3) months (the "Extended RCO Period").
 - (d) For the purposes of the sale and purchase of the RCO Shares, the RCO Purchaser and the relevant Vodafone Group Shareholders shall execute a share purchase agreement in the form set out in the Shareholders' Agreement. Such share purchase agreement, the Shareholders' Agreement and such other terms as may be mutually agreed shall be the sole terms which govern the sale and purchase of the RCO Shares.
 - (e) The consummation of the sale and purchase of the RCO Shares shall be completed within 10 (ten) Business Days of the date of receipt of the RCO Notice by the Vodafone Group Shareholders (such period to be extended *protanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of the RCO Notice).

- (f) The ICL Group Shareholders shall ensure that the exercise of the Rights Recapitalisation Call Option does not result in any requirement to make a public announcement of an open offer with respect to the Company under the Takeover Code.
- (g) The ICL Group Shareholders and the Vodafone Group Shareholders shall jointly appoint a Big Four Accounting Firm, or if no Big Four Accounting Firm is able or willing to act, another accounting firm of international standing, to provide:
 - (i) a certificate confirming (i) the RCO Price; and (ii) whether any Taxes are required to be withheld with respect to the sale and purchase of the RCO Shares, if such certificate is required under applicable Law; and
 - (ii) an opinion on computation of capital gains Taxes in connection with (a) above along with the necessary supporting documents in respect of cost of acquisition of the RCO Option Shares,

(together, the "RCO Withholding Computation"). The Vodafone Group Shareholders shall promptly provide any information required by the appointed accounting firm for purposes of issue of such certificate and shall confirm to the ICL Group Shareholders that such information is true and correct. On the date of completion of the Transfer of the RCO Shares, the RCO Purchaser shall pay the RCO Price to the relevant Vodafone Group Shareholders after withholding or deduction of any Tax required pursuant to the RCO Withholding Computation. All costs, fees and other expenses of the accounting firm appointed for the purposes of provision of the RCO Withholding Computation shall be borne equally by each Promoter Group.

(h) The ICL Group Shareholders and the Vodafone Group Shareholders shall execute all such documents, take all such actions and shall render all such assistance to each other as may be reasonably required to complete the Transfer of the RCO Shares.

4.7.2 If:

- (a) the ICL Group Shareholders have issued one or more RCO Notices and have acquired the RCO Shares specified therein up to the Rights Recapitalisation Cap pursuant to Article 4.7.1, the ICL Group Shareholders shall, for a period of three (3) months from the date of the last such acquisition, be entitled to acquire from the market such number of Equity Shares out of the entitlement of the ICL Group Shareholders that were subscribed to by the Public Shareholders (for avoidance of doubt, in excess of the Public Shareholders' entitlement) in the Rights Recapitalisation;
- (b) the ICL Group Shareholders have issued one or more RCO Notices but, due to restrictions under applicable Law, have been unable to acquire all of the RCO Shares specified therein up to the Rights Recapitalisation Cap during the RCO Period and the Extended RCO Period pursuant to Article 4.7.1(c), the ICL Group Shareholders shall, for a period of three (3) months from the expiration of the Extended RCO Period, be entitled to acquire from the market (i) any remaining RCO Shares as well as (ii) as such number of Equity Shares out of the entitlement of the ICL Group Shareholders that have been subscribed to by

- the Public Shareholders (for avoidance of doubt, in excess of the Public Shareholders' entitlement) in the Rights Recapitalisation; or
- (c) the Rights Recapitalisation Cap is zero, the ICL Group Shareholders shall, for a period of six (6) months from the date of completion of the relevant Rights Recapitalisation, be entitled to acquire from the market such number of Equity Shares out of the entitlement of the ICL Group Shareholders that have been subscribed to by the Public Shareholders (for avoidance of doubt, in excess of the Public Shareholders' entitlement) in the Rights Recapitalisation.

4.8 Dilution pursuant to a Rights Recapitalisation after the Equal Shareholding Date

- 4.8.1 At any time after the Equal Shareholding Date, if the percentage Shareholding of any Promoter Group is diluted (the "**Diluted Group**") to a level below the Shareholding of the other Promoter Group (the "**Non-Diluted Group**") pursuant to its non-participation or partial participation in a Rights Recapitalisation under this Article 4, then the Diluted Group shall, for a period of six (6) months of the date of completion of the relevant Rights Recapitalisation, have the right to acquire Equity Shares from the market, directly or through its Affiliates, to equalise its Shareholding with the Non-Diluted Group.
- 4.8.2 The Diluted Group shall ensure that the exercise of its rights under Article 4.8.1 does not result in any requirement to make a public announcement of an open offer with respect to the Company under the Takeover Code.

4.9 Initial Rights Recapitalisation Period

- 4.9.1 During the period from the Effective Date until 31 March 2019, if any Promoter Group seeks to implement a Rights Recapitalisation, it shall notify the other Promoter Group in writing. Within a period of 30 (thirty) days of such notice, the Promoter Groups shall discuss, in good faith, whether the Company requires additional equity capital taking into account the performance of and outlook for the Company at that time and the expected timing for realisation of synergies pursuant to the combination of the Company and Vodafone India Limited.
- 4.9.2 If the Promoter Groups cannot agree whether the Company requires additional equity capital within the 30-day period specified in Article 4.9.1, the matter shall be referred to the chief executive officer of Vodafone Plc and the group chairperson of the ICL Group for their consideration, and such persons shall be required to resolve the matter within 30 (thirty) days of the reference.
- 4.9.3 In the event such representatives of the Promoter Groups are unable to resolve such matter within the 30-day period specified in Article 4.9.2, the Promoter Groups shall jointly appoint an Investment Bank to advise the Board on the Terms of the Rights Recapitalisation proposed pursuant to Article 4.9.1. If the Promoter Groups are unable to agree on the selection of an Investment Bank within five (5) Business Days of the expiration of the 30-day period specified in Article 4.9.2, an Investment Bank shall promptly be appointed by the Board from among the Investment Banks listed in **Schedule 1** by draw of lots.
- 4.9.4 The Board shall consider the Terms of the Rights Recapitalisation proposed by the Investment Bank appointed by the Promoter Groups or by draw of lots, as applicable, and the Rights Recapitalisation shall proceed only if there is a decision of the Board to do so by a simple majority vote, it being understood that such decision shall not be a Reserved Matter.

4.9.5 All costs, fees and other expenses of the Investment Bank(s) appointed pursuant to this Article 4.9 shall be borne by the Company.

5. BOARD OF DIRECTORS OF THE COMPANY

5.1 Authority of the Board

Subject to the provisions of the Articles of Association and applicable Law, the Board shall be responsible for the management of the Company. The Board shall give due consideration to the views of Committees, however, the Board shall be responsible for taking final decisions on matters considered by such Committees. The approval of the Shareholders will be obtained for such matters as may be required under applicable Law or pursuant to the Articles of Association.

5.2 Composition of the Board

- 5.2.1 The Board shall consist of twelve (12) Directors as follows:
 - (a) three (3) nominee Directors of the ICL Group Shareholders;
 - (b) three (3) nominee Directors of the Vodafone Group Shareholders; and
 - (c) six (6) independent Directors,

in each case, appointed in accordance with this Article 5.2.

- 5.2.2 Subject to Article 17.3, each Promoter Group shall be entitled, by notice in writing to the Company (with a copy to the other Promoter Group), to require the Company to:
 - (a) appoint three (3) Directors nominated by it (the "Vodafone Group Directors" and the "ICL Group Directors", as applicable); and
 - (b) appoint three (3) independent Directors from among the persons recommended by it for such appointment.
- 5.2.3 To the extent the entitlement of any Promoter Group to nominate Directors and/or recommend persons for appointment as independent Directors is extinguished pursuant to any provision of the Articles of Association, such entitlement shall be transferred to the other Promoter Group and the entitlement of such other Promoter Group pursuant to Article 5.2.2 shall be increased automatically, provided that the Shareholding of such other Promoter Group is equal to or higher than the Qualifying Threshold and such other Promoter Group has rights under this Article 5.2.
- 5.2.4 If, at any time, the entitlement of any Promoter Group to nominate Directors and/or recommend persons for appointment as independent Directors is extinguished pursuant to any provision of the Articles of Association, then such Promoter Group shall procure that an appropriate number of Directors nominated or recommended for appointment by that Promoter Group shall resign and vacate office as promptly as practicable.

5.3 **Qualification**

The Directors shall not be required to hold any qualification Equity Shares.

5.4 **Board Committees**

5.4.1 Subject to Article 17.3, the Board shall constitute and determine the terms of reference of committees of the Board (each, a "Committee") to the extent required under applicable Law, including an audit committee, a nomination and remuneration committee, a stakeholders' relationship committee, a risk management committee and a corporate social responsibility committee.

5.4.2 Each Committee shall include:

- (a) such number of independent Directors as may be required under applicable Law from among the Persons recommended for appointment by the Promoter Groups; and
- (b) the maximum permissible number of ICL Group Directors and Vodafone Group Directors,

in each case, in the same proportion as is applicable to the constitution of the Board in Article 5.2.

5.4.3 The provisions of this Article 5, including with respect to conduct of meetings, quorum and manner of approval of business, and Article 10, as they apply to the Board, shall apply *mutatis mutandis* to Committees. If any Committee cannot agree on any matter, the Committee shall refer the matter to the Board.

5.5 Removal of Directors; Casual Vacancy

- 5.5.1 Each Promoter Group shall be entitled, by notice in writing to the Company (with a copy to the other Promoter Group and the concerned Director), to require any Director nominated by it to be removed from such position and the Company and the Promoter Groups shall promptly take steps for the removal of such Director in accordance with such request. In the event of such removal or if any Director nominated by a Promoter Group ceases to hold office for any other reason, such Promoter Group shall be entitled to require the Company to appoint another Director in his or her place pursuant to Article 5.2.2, as promptly as practicable.
- 5.5.2 In the event that an independent Director appointed from among the persons recommended by any Promoter Group ceases to hold office as a Director for any reason, such Promoter Group shall be entitled to recommend another person in his/her place.
- 5.5.3 Except as set forth in Article 5.2.4, the removal of a Director nominated by a Promoter Group or an independent Director appointed from among the persons recommended by any Promoter Group shall be subject to the prior written consent of such Promoter Group.

5.6 **Notice of Board Meetings**

- 5.6.1 A Board meeting may be called by the Chairperson or any two (2) other Directors by giving notice in writing to the company secretary of the Company, who shall convene a Board meeting within ten (10) days of such notice.
- 5.6.2 A notice of a Board meeting shall (i) be in English; (ii) specify a reasonably detailed written agenda specifying the date, time and agenda of such Board meeting; (iii)

include copies of all papers relevant for such Board meeting; and (iv) be sent via e-mail and in addition via courier. Unless waived in writing by at least one (1) Vodafone Group Director and at least one (1) ICL Group Director, no discussion, action, vote or resolution with respect to any item not included in the agenda of any meeting shall be taken at any meeting of the Board.

5.7 Chairperson of the Board

- 5.7.1 Subject to Article 17.3, the ICL Group Shareholders shall have the right to appoint the group chairperson of the ICL Group (or his successor) as the chairperson of the Company ("Chairperson"). The Chairperson shall chair all meetings of the Board that he attends.
- 5.7.2 In the absence of the Chairperson at a meeting of the Board, the Board shall appoint the chairperson from among the Directors present for such meeting of the Board.
- 5.7.3 In case of equality of votes, the Chairperson or any other person acting as chairperson at a meeting of the Board shall not have a second and casting vote.

5.8 **Resolution by Circulation**

- 5.8.1 Any resolution that is not required to be considered only at a Board meeting under applicable Law may be adopted by circulation by the Board, and such written resolution, if approved, shall be filed with the minutes of proceedings of the Board along with all the documents/information circulated with it ("Circular Resolution").
- 5.8.2 Subject to Article 10 (*Reserved Matters*), no Circular Resolution shall be deemed to have been duly passed by the Board, unless the resolution has been circulated in draft in accordance with the Act, together with the necessary papers required for considering the resolution, and approved in writing by a majority of the Directors as are entitled to vote on the resolution.

5.9 **Remote Participation**

Subject to the provisions of the Act:

- 5.9.1 the Directors may participate in a Board meeting by way of video conference or conference telephone or similar equipment ("Remote Participation") designed to allow the Directors to participate equally in the Board meeting; and
- 5.9.2 a Board meeting held by Remote Participation shall be valid so long as a quorum in accordance with Article 5.10 is achieved pursuant to the Directors being able to participate in such Board meeting through video conference, telephone conference or similar equipment.

5.10 Quorum

The quorum for a meeting of the Board, duly convened and held, including by Remote Participation, shall be one-third of the total number of Directors or two (2) Directors, whichever shall be higher. Provided however that, no quorum as aforesaid shall be validly constituted, and no business at any Board meeting shall be transacted, unless at least one (1) ICL Group Director and one (1) Vodafone Group Director are present at the commencement of such meeting and throughout its proceedings (unless this requirement has been expressly waived in writing by the relevant Promoter Group). In the absence of a valid quorum at a duly convened Board meeting, the Board meeting shall be automatically adjourned to the same day in the next week

at the same time. The quorum at such adjourned Board meeting shall, notwithstanding anything to the contrary contained hereinabove, be one-third of the total number of Directors or two (2) Directors, whichever shall be higher and all business transacted thereat shall be regarded as having been validly transacted, provided, however, that no Reserved Matters shall be discussed or transacted at any such adjourned Board meeting unless at least one (1) ICL Group Director and at least one (1) Vodafone Group Director are present at the commencement of such adjourned meeting and throughout its proceedings.

5.11 Voting

- 5.11.1 At any Board meeting, each Director may exercise one (1) vote.
- 5.11.2 Subject to Article 10 (*Reserved Matters*), all business arising at any Board meeting shall be approved by a resolution passed by a majority of the Directors present and voting at such meeting.
- 5.11.3 In case of equality of votes while voting on a resolution not pertaining to a Reserved Matter, the relevant resolution shall be referred to the chief executive officer of Vodafone Plc and the group chairperson of the ICL Group for their consideration and decision. In the event such representatives of the Promoter Groups are unable to resolve such matter, then *status quo* shall prevail.
- 5.11.4 Each Promoter Group shall use all reasonable endeavours to ensure that at least one (1) Director nominated by it shall attend each Board meeting.

5.12 Observers at the Board Meeting

The CEO and the CFO shall attend meetings of the Board as observers. In addition, the Board shall be entitled to invite any employees or advisors of the Company to attend meetings of the Board as observers or for such other purpose as it may deem fit.

5.13 Compliance

The Company shall, and each Promoter Group shall procure that the Company shall, comply with the Articles of Association, including Article 5. Each Promoter Group shall exercise its votes in relation to all the Equity Shares held by it and take all other actions necessary to ensure compliance with the Articles of Association, including Article 5.

6. SHAREHOLDERS MEETINGS

6.1 General Meetings of Shareholders

The Chairperson of the Board shall be the chairperson of the meeting of the Shareholders ("General Meeting"). In the absence of the Chairperson, the Directors present shall select the chairperson from among themselves for such General Meeting.

6.2 **Quorum**

Quorum at the General Meeting shall comprise of such number of Shareholders to be present in person as required under applicable Law, provided, however that, no quorum as aforesaid shall be validly constituted, and no business at any General Meeting shall be transacted, unless at least one (1) duly authorised representative of the ICL Group Shareholders and at least one (1) duly authorised representative of the Vodafone Group Shareholders are present at the commencement of such meeting and throughout its proceedings (unless this requirement has

been expressly waived in writing by the relevant Promoter Group). In the absence of a valid quorum at a duly convened General Meeting, the General Meeting shall be adjourned to the same day in the next week at the same time and place or such other date, time and place as the Board may determine. In the absence of a valid quorum at such adjourned General Meeting, the Shareholder(s) present in person thereat shall, notwithstanding anything to the contrary herein contained, constitute the quorum and all business transacted thereat shall be regarded as having being validly transacted, provided, however that, no Reserved Matters shall be discussed or transacted at any such adjourned General Meeting unless at least one (1) representative of the ICL Group Shareholders and at least one (1) representative of the Vodafone Group Shareholders are present at the commencement of such adjourned meeting and throughout its proceedings.

7. KEY EMPLOYEES

Subject to Article 17.3:

- 7.1 The appointment of the CEO and the COO shall require the approval of both Promoter Groups (and the ICL Group Directors and the Vodafone Group Directors, as applicable) in accordance with Article 10:
- 7.2 Either Promoter Group may at any time, by giving written notice to the other Promoter Group and the Company, require the dismissal from the Company of the CEO or the COO. Upon receipt of such notice, the Company shall effect such dismissal as soon as reasonably practicable and each Promoter Group shall take all steps necessary to effect such dismissal; and
- 7.3 The Vodafone Group Shareholders shall have the right to appoint or dismiss the CFO by giving written notice to the ICL Group Shareholders and the Company. Upon receipt of such notice, the Company shall effect such appointment or dismissal as soon as reasonably practicable and each Promoter Group shall take all steps necessary to effect such appointment or dismissal.

8. UNDERTAKINGS OF THE COMPANY

- 8.1 The Company hereby undertakes and covenants to the Promoter Groups as follows:
 - 8.1.1 the Company shall not recognise or register any Transfer of Equity Shares unless effected in accordance with the provisions of the Articles of Association;
 - 8.1.2 the Company shall maintain prudent insurance, including directors' and officers' liability insurance, with a well-established and reputable insurer(s) in accordance with current industry practice from time to time against all risks usually insured against by companies carrying on the same business as or a business similar to the Company;
 - 8.1.3 the Company and its Group, at all times, shall keep and maintain proper, complete and accurate proper Books and Records in accordance with Ind AS and applicable Law;
 - 8.1.4 the Company shall procure that its Group's Books and Records, as required, are duly audited by the auditors annually as soon as possible after the end of each Financial Year and as required from time to time pursuant to applicable Law;
 - 8.1.5 the Company shall use all reasonable endeavours to obtain and maintain in full force and effect all approvals, consents or licences necessary for the conduct of the Business and comply with all material applicable Law in the conduct of its business;
 - 8.1.6 subject to applicable Law, the Company shall provide such information to the Promoter Groups as may be required by any member of their Group for any statutory filings under applicable Law or any other general financial reporting of their Group;

- 8.1.7 the Company shall take all steps promptly to protect the Intellectual Property rights it or its Group owns or lawfully uses. The Company shall immediately notify the relevant Promoter Group upon becoming aware of any infringement of Intellectual Property rights of such Promoter Group;
- 8.1.8 the Company shall, and shall ensure that during the course of performance of their duties, the management of the Company shall, at all times, provide equal treatment to the Shareholders except as set forth in the Articles of Association;
- 8.1.9 no Shareholder, Director, officer, employee, agent or any of their respective delegates shall take any action purporting to commit the Company or a Subsidiary in relation to any of the Reserved Matters unless such Reserved Matter has been approved in accordance with Article 10;
- 8.1.10 the Company and its Group shall comply with such corporate policies and procedures, including in relation to anti-bribery and anti-corruption, insider dealing and data and privacy protection, as shall have been adopted in a form agreed, *inter alia*, among the Parties and effective as of the Effective Date; and
- 8.1.11 subject to Article 10 (*Reserved Matters*), if the Company or any member of its Group procures any products or services from any member(s) of a Promoter Group, the contract or arrangements entered into with respect to such products or services will be entered into on an arms' length basis and in accordance with applicable Law.

9. UNDERTAKINGS OF THE OTHER PARTIES

- 9.1 Each Promoter Group hereby undertakes and covenants to the other Promoter Group and the Company as follows:
 - 9.1.1 the Directors nominated by it shall:
 - (a) not wilfully or unreasonably fail to attend a Board meeting in order to prevent the transaction of business at that Board meeting; and
 - (b) exercise their rights to ensure compliance with the Articles of Association by the relevant Promoter Group and the Company;
 - 9.1.2 the members of the relevant Promoter Group shall, including through their duly authorised representatives, proxies or agents at General Meetings, exercise votes in respect of the Equity Shares held by them to ensure compliance with the Articles of Association by the relevant Promoter Group and the Company;
 - 9.1.3 if any shareholders' resolution contrary to the terms of the Articles of Association is proposed, the relevant Promoter Group shall vote against such resolution;
 - 9.1.4 if any shareholders' resolution is adopted or rejected otherwise than in accordance with the terms of the Articles of Association, the relevant Promoter Group shall cooperate with the other Promoter Group and the Company to convene another General Meeting or issue a fresh notice for a shareholders' vote;
 - 9.1.5 if any proposal that is a Reserved Matter is approved and/or implemented in contravention of the Articles of Association, it shall exercise all rights and powers available to it, including voting and causing the ICL Group Directors or the Vodafone Group Directors, as applicable, to vote in favour of, any subsequent resolutions of the

Board or the Shareholders, to procure that the position which prevailed prior to such proposal having been approved and/or implemented is restored;

- 9.1.6 it shall not Transfer, or cause to be Transferred, any Equity Shares held by such Promoter Group except in accordance with the Articles of Association; and
- 9.1.7 it shall not undertake any acquisition that results in any requirement to make a public announcement of an open offer with respect to the Company under the Takeover Code.

9.2 **ICL Group**

- 9.2.1 So long as any ICL Group Shareholder (or any Affiliate thereof) holds Equity Shares:
 - (a) KMB hereby undertakes and covenants to the Vodafone Group Shareholders and the Company that: (i) he shall, as a Shareholder, comply with the terms of the Articles of Association; (ii) he shall, and shall do everything within his power to cause the ICL Group Shareholders to (by way of his and his Affiliates' direct and indirect shareholding in the ICL Group Shareholders or otherwise), comply with the Articles of Association and vote the Equity Shares held by him and them to implement the provisions of the Articles of Association; (iii) he shall, directly or through his Affiliates, continue to be a promoter of each ICL Group Shareholder; and (iv) he shall own at least 26% of the share capital of each ICL Group Shareholder, either directly or through his Affiliates. If any of (iii) or (iv) is not satisfied in respect of any ICL Group Shareholder, the Shareholding of such ICL Group Shareholder shall be excluded for the purpose of determining whether the ICL Group Shareholders hold the Qualifying Threshold;
 - (b) the ICL Group Shareholders hereby undertake and covenant to the Vodafone Group Shareholders and the Company that Grasim Industries Limited ("GIL") and Aditya Birla Nuvo Limited (if not yet merged with GIL) shall remain ICL Group Shareholders;
 - (c) GIL undertakes that it shall, on an annual basis (within 60 (sixty) days of the end of its financial year), provide a confirmation to the Vodafone Group Shareholders that its Net Assets are equal to at least the Net Assets Threshold; and
 - (d) the ICL Group Shareholders undertake that if the Net Assets of GIL fall below the Net Assets Threshold, GIL's Shareholding shall immediately be transferred to an Affiliate that satisfies the Net Assets Threshold in accordance with Article 13.2.1.
- 9.2.2 Until 31 March 2020, the ICL Group Shareholders shall ensure that their Shareholding does not fall below 26% of the Share Capital.

9.3 **Vodafone Confirming Party**

The Vodafone Confirming Party hereby undertakes and covenants to the ICL Group Shareholders and the Company that so long as any Vodafone Group Shareholder holds Equity Shares:

9.3.1 it shall ensure that the Vodafone Group Shareholders shall comply with the Articles of Association and shall vote the Equity Shares held by them to implement the provisions of the Articles of Association;

- 9.3.2 it shall, on an annual basis (within 60 (sixty) days of the end of its financial year), provide a confirmation to the ICL Group Shareholders that its Net Assets are equal to at least the Net Assets Threshold; and
- 9.3.3 if at any time its Net Assets fall below the Net Assets Threshold, it shall procure that an Affiliate that satisfies the Net Assets Threshold will immediately replace it as the Vodafone Confirming Party by executing a deed of adherence that shall require compliance with its obligations under the Articles of Association.

For the avoidance of doubt, this Article 9.3 shall cease to apply in respect of an entity which ceases to be a Vodafone Group Shareholder pursuant to the Shareholders' Agreement and the Articles of Association.

9.4 Holding of Equity Shares

Save as permitted under Articles 13.2.2A, 13A and 13C, to the extent required by applicable Law, the Vodafone Group Shareholders shall hold Equity Shares that are subject to the Call Option 1 and the Call Option 2 for the prescribed time period, if any, for the exercise of such call options.

10. RESERVED MATTERS

- 10.1 No action shall be taken by the Company or any member of its Group in relation to any matter enumerated in Article 10.4 (each, a "Reserved Matter"): (i) without the affirmative vote of at least one (1) ICL Group Director and at least one (1) Vodafone Group Director present and voting if the matter is placed before a Board meeting and without the prior written approval of at least one (1) ICL Group Director and at least one (1) Vodafone Group Director if the matter is placed before the Board through a Circular Resolution; and (ii) if the matter is placed before the Shareholders at a General Meeting or otherwise, without the affirmative vote of all ICL Group Shareholders and all Vodafone Group Shareholders.
- In relation to any Reserved Matter that requires the approval of the Shareholders pursuant to the Act or the Articles of Association, such matter shall not be placed before the Shareholders until it has been approved by the Board in accordance with the Articles of Association. If a Reserved Matter has been approved by the Board pursuant to Article 10.1 and then placed before the Shareholders, each member of the Promoter Groups shall be required to vote in favour of it in their capacity as a Shareholder.
- 10.3 If a resolution for any matter that is a Reserved Matter is proposed directly by any Public Shareholder for the consideration of the Shareholders in a General Meeting pursuant to the Act, which matter has not previously been considered and approved by the Board then, unless both of the Promoter Groups agrees (in writing) to vote in favour prior to the General Meeting, each Promoter Group shall be required to vote against it at the General Meeting.
- 10.4 The following matters shall be the Reserved Matters under the Articles of Association:
 - 10.4.1 any amendment to the memorandum of association of the Company or the Articles of Association;
 - 10.4.2 any change to the rights attaching to any class of shares in the Company;
 - 10.4.3 any consolidation, sub-division, reclassification or cancellation of any Share Capital (or share premium or other reserve);

- 10.4.4 any redemption, reduction or buy-back of any Share Capital;
- 10.4.5 the issue or allotment of any Share Capital or the creation of any option or right to subscribe or acquire, or convert any security into, any Share Capital, including pursuant to employee stock option schemes, other than as permitted pursuant to Article 4;
- 10.4.6 liquidation or dissolution of the Company or the filing of a petition for winding up by the Company or the making of any arrangement with creditors generally or any application for an administration order or for the appointment of a receiver or administrator:
- 10.4.7 merger, amalgamation, demerger, reorganisation or restructuring of the Company, including pursuant to a scheme of arrangement under the Act;
- 10.4.8 any dividend policy which has effect during the Term and any change in the dividend policy or treasury policy of the Company;
- 10.4.9 declaration or payment of any dividend in any manner inconsistent with the dividend policy of the Company;
- 10.4.10 incurrence of any financial indebtedness in excess of Rs.70 billion or the variation or termination of any agreement for the raising of any such indebtedness (including early repayment) other than in accordance with the Company's treasury policy;
- 10.4.11 entering into any derivatives transactions, other than in accordance with the Company's treasury policy;
- 10.4.12 the adoption of any new Business Plan or any amendment to any current Business Plan, or the approval or ratification of any departure from the current Business Plan;
- 10.4.13 acquisition or disposal of any shares, assets (including receivables and financial assets), business, business organisation or division in any manner in excess of Rs.2 billion in a single transaction or series of related transactions (other than in accordance with the Company's treasury policy);
- 10.4.14 entry into (or the amendment, variation or termination of) any partnership, joint venture or profit-sharing agreement other than any arrangements entered into in the ordinary course of the Business;
- 10.4.15 entry into any agreement for the procurement of materials and/or services where the value of the contract over its term exceeds Rs.1 billion;
- 10.4.16 entry into (or the amendment or variation of) any related party transaction the value of which exceeds Rs.250 million in aggregate;
- 10.4.17 the appointment of the CEO and the COO;
- 10.4.18 any material change to the nature or scope of the Business;
- 10.4.19 any change to the name or key brands or branding strategy of the Business (including any decision to cease using the Idea or Vodafone brands), or any step to implement any such change;
- 10.4.20 any change in the size of the Board;
- 10.4.21 any change in statutory auditors or accounting policies;

- 10.4.22 authorising, or committing or agreeing to take, any of the foregoing actions; and
- 10.4.23 the effecting of any of the above matters by any member of the Company's Group (as if references to the Company were to such member).

11. BUSINESS PLAN

- 11.1 The Company shall procure that the executive management of the Company shall prepare a Business Plan which is submitted to the Board to replace the existing Business Plan (each, a "Draft Revised Business Plan") as follows:
 - 11.1.1 by no later than six (6) months prior to the end of the Financial Year commencing after the Effective Date, comprising a financial and strategic plan for a period of five (5) years from the commencement of the following Financial Year;
 - 11.1.2 by no later than 70 (seventy) days prior to the end of each Financial Year commencing after the Effective Date, an update of the plan prepared in accordance with Article 11.1.1 above and a detailed monthly operating budget for the 12 (twelve) months comprising the next Financial Year,

in the same format as the initial business plan in effect on or immediately after the Effective Date or in such other format as has been approved in accordance with Article 10 (*Reserved Matters*).

- 11.2 Each Draft Revised Business Plan submitted to the Board in accordance with Article 11.1 shall address, but not be limited to, the items and subject matter of the initial business plan in effect on or immediately after the Effective Date.
- 11.3 The Draft Revised Business Plan referenced in Article 11.1.2 shall be finalised by the executive management of the Company prior to the start of the period to which it relates. Promptly following such finalisation, such Draft Revised Business Plan shall be considered, and subject to Article 10 (*Reserved Matters*), adopted as the Business Plan, by the Board. The Board shall use all reasonable endeavours to approve the Business Plan referenced in Article 11.1.2 prior to the start of the last month of the Financial Year.
- In the event that a Draft Revised Business Plan is not approved and adopted as the Business Plan by the Board, the Company will continue to operate in accordance with the most recent approved Business Plan. In the event that the most recent approved Business Plan does not cover the next applicable period under Article 11.1.2, the Company shall be operated in accordance with the most recently approved Business Plan, adjusted to reflect the percentage change in the consumer price index (as published by the Government of India) for the relevant period.
- 11.5 The executive management of the Company shall present to the Board a comparison of the Company's actual operating performance with the Business Plan on a quarterly basis, in a format agreed with the Promoter Groups.

12. TERMS OF EQUALISATION

- 12.1 The provisions of this Article 12 shall apply until the earlier of: (i) the Equal Shareholding Date; and (ii) the expiration of nine (9) years and one (1) Business Day from the Effective Date, except as set forth in this Article 12.1:
 - 12.1.1 The Vodafone Group Shareholders shall promptly notify the ICL Group Shareholders and the Company (in writing) of any imposition and cessation of a Transfer Embargo

to which they are subject. If a Call Option 1 Notice or a Call Option 2 Notice is issued and, if one or more Vodafone Group Shareholders are subject to a Transfer Embargo as a result of which the Call Option 1 Shares or the Call Option 2 Shares, as the case may be, cannot be Transferred to the Call Option 1 Purchaser(s) or the Call Option 2 Purchaser(s) within the time periods set out in Article 12.3, the Promoter Groups shall complete the Transfer of such Equity Shares promptly upon cessation of the Transfer Embargo, provided that in the event that the Transfer Embargo is in force upon the expiration of twelve (12) months from the date of the Call Option 1 Notice or the Call Option 2 Notice, as applicable, the ICL Group Shareholders shall have the right to withdraw such notice. Following such withdrawal, the Vodafone Group Shareholders shall not have any obligation to Transfer the Call Option 1 Shares or the Call Option 2 Shares, as applicable, to the Call Option 1 Purchaser(s) or the Call Option 2 Purchaser(s), as applicable, and the Call Option 1 Shares or the Call Option 2 Shares, as applicable, shall be excluded from the calculation of Excess Equity Shares and, for the avoidance of doubt, shall not be subject to the voting restrictions under Article 12.2.

- 12.1.2 During the Step Down Option 1 Period or the Step Down Option 2 Period, if one or more Vodafone Group Shareholders are subject to a Transfer Embargo, the Step Down Option 1 Period and the Step Down Option 2 Period, as applicable, shall be deemed to be extended by the duration of the Transfer Embargo.
- 12.1.3 The provisions of Article 12.2 shall apply to any Excess Equity Shares: (a) in respect of which a Call Option 1 Notice or Call Option 2 Notice has been issued and that are subject to a Transfer Embargo until the earlier of (i) withdrawal of the Call Option 1 Notice or the Call Option 2 Notice, as applicable, by the ICL Group Shareholders and (ii) completion of the Transfer pursuant to the exercise of the Call Option 1 or the Call Option 2, as applicable, upon cessation of the Transfer Embargo as set out in Article 12.1.1; and (b) during any extension of the Step Down Option 1 Period and/or the Step Down Option 2 Period, as applicable, pursuant to Article 12.1.2.

12.2 Voting rights in Excess Equity Shares

Subject to Articles 15.1.3, 15.2.4 and 17.3:

12.2.1 ICL Opposition Notice

- (a) If the ICL Group Shareholders intend to oppose any resolution(s) at a General Meeting or through postal ballot, they shall, within five (5) Business Days of receipt of the notice for the General Meeting or postal ballot, send a written notice to the Vodafone Group Shareholders and the Company specifying the resolution(s) which they intend to oppose (the "ICL Opposition Notice").
- (b) Within five (5) Business Days of receipt of the ICL Opposition Notice, the Vodafone Group Shareholders shall inform the ICL Group Shareholders and the Company (in writing) of the number of Excess Equity Shares they hold at the time and whether they intend to oppose any resolution(s) specified in the ICL Opposition Notice ("Vodafone Confirmation Notice").
- (c) The Vodafone Group Shareholders shall waive, and shall not exercise, the voting rights attached to the Excess Equity Shares in relation to the resolution(s) specified in the ICL Opposition Notice unless the Vodafone Confirmation Notice specifies the intention of the Vodafone Group Shareholders to vote against any such resolution(s), in which case the Vodafone Group Shareholders shall exercise their voting rights attached to the Excess Equity Shares to vote against such resolution(s).

12.2.2 Vodafone Opposition Notice

- (a) If the Vodafone Group Shareholders intend to oppose any resolution(s) at a General Meeting or through postal ballot, they shall, within five (5) Business Days of receipt of the notice for the General Meeting or postal ballot, send a written notice to the ICL Group Shareholders and the Company specifying the resolution(s) they intend to oppose and the number of the Excess Equity Shares they hold at the time (the "Vodafone Opposition Notice").
- (b) Within five (5) Business Days of the receipt of the Vodafone Opposition Notice, the ICL Group Shareholders shall inform the Vodafone Group Shareholders and the Company (in writing) whether they intend to oppose any resolution(s) specified in the Vodafone Opposition Notice ("ICL Confirmation Notice").
- (c) The Vodafone Group Shareholders shall waive, and shall not exercise, the voting rights attached to the Excess Equity Shares in relation to the resolution(s) specified in the Vodafone Opposition Notice unless the ICL Confirmation Notice confirms the intention of the ICL Group Shareholders to vote against any such resolution(s), in which case the Vodafone Group Shareholders shall exercise their voting rights attached to the Excess Equity Shares to vote against such resolution(s).
- 12.2.3 If no ICL Opposition Notice or Vodafone Opposition Notice is received in respect of a shareholders' resolution, each Promoter Group shall exercise its vote in favour of such resolution at a General Meeting or through postal ballot in respect of all the Equity Shares held by it.

Any vote by the Vodafone Group Shareholders in respect of the Excess Equity Shares in violation of this Article 12.2 (*Voting rights in Excess Equity Shares*) shall be invalid, null and *void ab initio*, and the Company shall not recognise or give effect to such vote in respect of the resolution(s) to which the ICL Opposition Notice or the Vodafone Opposition Notice, as applicable, relates.

12.3 Equalisation Call Options

12.3.1 **Call Option 1**

- (a) During the Call Option 1 Period, the ICL Group Shareholders shall have the right to acquire from the Vodafone Group Shareholders, directly or through their Affiliates, such number of Equity Shares that is equal to or less than the Call Option Cap, in the manner set forth in this Article 12.3.1 ("Call Option 1").
- (b) Call Option 1 may be exercised a maximum of four (4) times during the Call Option 1 Period, each time in compliance with the provisions of this Article 12.3.1. For the avoidance of doubt, the number of Equity Shares that may be

- purchased by the ICL Group Shareholders pursuant to each exercise of the Call Option 1 shall not exceed the Call Option Cap at the time of such exercise.
- (c) The ICL Group Shareholders may exercise Call Option 1 by issuing a written notice to the Vodafone Group Shareholders (a "Call Option 1 Notice"), which shall specify: (i) the identity of the purchaser(s) (the "Call Option 1 Purchaser(s)"); (ii) certification that the Call Option 1 Purchaser(s) is an ICL Group Shareholder or an Affiliate of an ICL Group Shareholder (iii) the number of Equity Shares the Call Option 1 Purchaser(s) wishes to acquire (the "Call Option 1 Shares"); and (iv) the price payable for such Call Option 1 Shares, which shall be equal to the product of the Call Option 1 Equity Share Value and the number of Call Option 1 Shares (the "Call Option 1 Price").
- (d) The Vodafone Group Shareholders shall, at their sole discretion, determine the identity of the Vodafone Group Shareholder(s) that shall Transfer the Call Option 1 Shares to the Call Option 1 Purchaser(s) and shall, within five (5) Business Days of the receipt of the Call Option 1 Notice, notify the details thereof (in writing) to the Call Option 1 Purchaser(s) together with the number of Equity Shares that each such member will Transfer. Notwithstanding anything contained in the Articles of Association, no Vodafone Group Shareholder shall be required to Transfer any Call Option 1 Shares to the Call Option 1 Purchaser(s) at a price less than the Call Option 1 Price.
- (e) For the purposes of the sale and purchase of the Call Option 1 Shares, the Call Option 1 Purchaser(s) and the relevant Vodafone Group Shareholder(s) shall execute a share purchase agreement in the form set out in the Shareholders' Agreement. Such share purchase agreement, the Shareholders' Agreement and such other terms as may be mutually agreed shall be the sole terms which govern the sale and purchase of the Call Option 1 Shares.
- (f) Subject to Article 12.1, the consummation of the sale and purchase of any Call Option 1 Shares shall be completed within ten (10) Business Days of the date of receipt of a Call Option 1 Notice by the Vodafone Group Shareholders (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from, or make any necessary filing with, any Governmental Authority or seek any Vodafone Plc Shareholder Approval in accordance with Article 12.7, provided that such extended period shall be no longer than 12 (twelve) months from the date of the Call Option 1 Notice).
- (g) The ICL Group Shareholders shall ensure that the exercise of Call Option 1 does not result in any requirement to make a public announcement of an open offer with respect to the Company under the Takeover Code.

12.3.2 **Call Option 2**

- During a period of one (1) year following the expiration of the Call Option 1 Period ("Call Option 2 Period"), the ICL Group Shareholders shall have the right to acquire from the Vodafone Group Shareholders, directly or through their Affiliates, such number of Equity Shares that is equal to or less than the Call Option Cap, in the manner set forth in this Article 12.3.2 ("Call Option 2").
- (b) Within seven (7) Business Days of the commencement of the Call Option 2 Period, the ICL Group Shareholders shall notify the Vodafone Group Shareholders (in writing) of the precise number of Equity Shares, if any, that

the ICL Group Shareholders will acquire from the Vodafone Group Shareholders pursuant to the exercise of Call Option 2 during the Call Option 2 Period (the "Call Option 2 Shares"). If such notice is not given within such time period, Call Option 2 shall cease to be capable of exercise and shall lapse. If such notice is given within such time period, the ICL Group Shareholders must acquire the Call Option 2 Shares strictly in accordance with this Article 12.3.2.

- (c) Call Option 2 is required to be exercised only once during the Call Option 2 Period.
- (d) The ICL Group Shareholders may exercise Call Option 2 by issuing a written notice to the Vodafone Group Shareholders ("Call Option 2 Notice"), which shall specify: (i) the identity of the purchaser(s) (the "Call Option 2 Purchaser(s)"); and (ii) certification that the Call Option 2 Purchaser(s) is an ICL Group Shareholder or an Affiliate of an ICL Group Shareholder.
- (e) The Vodafone Group Shareholders shall, at their sole discretion, determine the identity of the Vodafone Group Shareholder(s) that shall Transfer the Call Option 2 Shares to the Call Option 2 Purchaser(s) and shall, within five (5) Business Days of the receipt of the Call Option 2 Notice, notify the details thereof (in writing) to the Call Option 2 Purchaser(s) together with the number of Equity Shares that each such member will Transfer. The price payable for the Call Option 2 Shares shall be equal to the product of the Vodafone Sale Price (determined on the date of the Transfer) and the number of Call Option 2 Shares (the "Call Option 2 Price"). Notwithstanding anything contained in the Articles of Association, no Vodafone Group Shareholder shall be required to Transfer any Call Option 2 Shares to the Call Option 2 Purchaser(s) at a price less than the Vodafone Sale Price.
- (f) For the purposes of the sale and purchase of the Call Option 2 Shares, the Call Option 2 Purchaser(s) and the relevant Vodafone Group Shareholder(s) shall execute a share purchase agreement in the form set out in the Shareholders' Agreement. Such share purchase agreement, the Shareholders' Agreement and such other terms as may be mutually agreed shall be the sole terms which govern the sale and purchase of the Call Option 2 Shares.
- (g) Subject to Article 12.1, the consummation of the sale and purchase of the Call Option 2 Shares shall be completed within 10 (ten) Business Days of the date of receipt of the Call Option 2 Notice by the Vodafone Group Shareholders (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority or seek any Vodafone Plc Shareholder Approval in accordance with Article 12.7, provided that such extended period shall be no longer than 12 (twelve) months from the date of the Call Option 2 Notice).
- (h) The ICL Group Shareholders shall ensure that the exercise of Call Option 2 does not result in any requirement to make a public announcement of an open offer with respect to the Company under the Takeover Code.
- (i) If, following the issue of a notice to the Vodafone Group Shareholders pursuant to Article 12.3.2(b), the ICL Group Shareholders fail to exercise the Call Option 2 for all of the Call Option 2 Shares within the Call Option 2 Period, the Call Option 2 Shares shall not be included in the Shareholding of the Vodafone Group Shareholders for purposes of calculation of the Excess Equity

Shares and, for the avoidance of doubt, shall not be subject to the voting restrictions under Article 12.2.

- 12.3.3 In respect of each exercise of Call Option 1 and the exercise of Call Option 2, the Vodafone Group Shareholders and the ICL Group Shareholders shall jointly appoint a Big Four Accounting Firm, or if no Big Four Accounting Firm is able or willing to act, another accounting firm of international standing, to provide:
 - (a) a certificate confirming (i) the Call Option 1 Price or the Call Option 2 Price, as applicable; and (ii) whether any Taxes are required to be withheld with respect to the sale and purchase of the Call Option 1 Shares or the Call Option 2 Shares, as applicable, if such certificate is required under applicable Law; and
 - (b) an opinion on computation of capital gains Taxes in connection with (a) above along with the necessary supporting documents in respect of cost of acquisition of the Call Option 1 Shares or the Call Option 2 Shares, as applicable,

(together, the "Withholding Computation"). The Vodafone Group Shareholders shall promptly provide any information required by the appointed accounting firm for purposes of issue of such certificate and shall confirm to the ICL Group Shareholders that such information is true and correct. All costs, fees and other expenses of the accounting firm appointed for the purposes of provision of the Withholding Computation shall be borne by the Call Option 1 Purchaser or the Call Option 2 Purchaser, as applicable.

- 12.3.4 The Call Option 1 Purchaser or the Call Option 2 Purchaser, as applicable, shall pay the Call Option 1 Price or the Call Option 2 Price, as applicable, without withholding or deduction of any Tax unless required by the Withholding Computation. If any such withholding or deduction is required pursuant to the Withholding Computation, the Call Option 1 Purchaser or the Call Option 2 Purchaser, as applicable, shall, at the time of payment of the Call Option 1 Price or the Call Option 2 Price, as applicable, pay to the Vodafone Group Shareholders such additional amount as will ensure that the Vodafone Group Shareholders receives the same total amount that they would have received if no such withholding or deduction had been required. If any sum payable by the Call Option 1 Purchaser or the Call Option 2 Purchaser, as applicable, to the Vodafone Group Shareholders pursuant to Articles 12.3.1 and 12.3.2 is required by applicable Law to be brought into charge to Tax in the hands of the Vodafone Group Shareholders, then the Call Option 1 Purchaser or the Call Option 2 Purchaser, as applicable, shall pay such additional amount as shall be required to ensure that the total amount paid, less the Tax chargeable on such amount (or which would be chargeable but for the use or set-off of any Tax relief of the recipient), is equal to the amount that would be payable if the sum payable by the Call Option 1 Purchaser or the Call Option 2 Purchaser, as applicable, were not required by applicable Law to be brought into charge to Tax in the hands of the Vodafone Group Shareholders.
- 12.3.5 It is clarified that any Tax benefits or refunds accruing to or received by the Vodafone Group Shareholders following completion of a Transfer of the Call Option 1 Shares and/ or the Call Option 2 Shares in respect of such Transfer shall promptly be transferred to the Call Option 1 Purchaser or the Call Option 2 Purchaser, as applicable, up to the maximum amount of the payment received by the Vodafone Group Shareholders from the Call Option 1 Purchaser or the Call Option 2 Purchaser, as applicable, under Article 12.3.4.

- 12.3.6 The Parties shall undertake all reasonable endeavours to ensure that the Transfers of Call Option 1 Shares and the Call Option 2 Shares, as applicable, is completed in a Tax efficient manner. The Vodafone Group Shareholders shall ensure that the Call Option 1 Shares and the Call Option 2 Shares will be Transferred by Vodafone Group Shareholders that are tax residents of Mauritius or India.
- 12.3.7 The ICL Group Shareholders and Vodafone Group Shareholders shall execute all such documents, take all such actions and shall render all such assistance to each other as may be reasonably required to complete the Transfer of the Call Option 1 Shares and the Call Option 2 Shares.

12.4 **Step Down Option 1**

If the Equal Shareholding Date has not occurred by the expiration of the Call Option 2 Period, then the Vodafone Group Shareholders shall, during a period of three (3) years of the expiration of the Call Option 2 Period (the "Step Down Option 1 Period"), sell and Transfer to such Persons as the Vodafone Group Shareholders may (subject to Article 13) choose and which Persons do not execute a Deed of Adherence in terms of Article 13, that part of their combined holding of Equity Shares as is in aggregate equal to the lower of:

- (a) the number of Excess Equity Shares as at the expiration of the Call Option 2 Period; and
- (b) 10% of the Share Capital as at the expiration of the Call Option 2 Period,

(as applicable, the "Step Down 1 Excess Shareholding"), together with any Equity Shares to the extent attributable to any split, reverse-split, bonus issue or any similar corporate action with respect to the Step Down 1 Excess Shareholding during the period starting at the commencement of the Step Down Option 1 Period and ending at the time of each relevant disposal pursuant to this Article 12.4 ("Step Down Option 1").

12.5 **Step Down Option 2**

If the Equal Shareholding Date has not occurred by the expiration of the Step Down Option 1 Period, then the Vodafone Group Shareholders shall, within a period of two (2) years of the expiration of the Step Down Option 1 Period (the "Step Down Option 2 Period"), sell and Transfer to such Persons as the Vodafone Group Shareholders may (subject to Article 13) choose and which Persons do not execute a Deed of Adherence in terms of Article 13, all remaining Excess Equity Shares at the expiration of Step Down Option 1 Period (together with any Equity Shares to the extent attributable to any split, reverse-split, bonus issue or any similar corporate action with respect to such Excess Equity Shares since the commencement of the Step Down Option 2 Period), provided that in the judgment of the Vodafone Group, acting reasonably, that:

- 12.5.1 market conditions are conducive for such sale;
- 12.5.2 the valuation that can be achieved for such sale is not lower than the product of the Step Down Share Value and the number of Equity Shares proposed to be Transferred; and
- 12.5.3 the ratio of (i) the proposed sale consideration per share multiplied by the number of Equity Shares of the Company (on a fully diluted basis) plus the Net Financial Debt of

the Company, in each case, on the date of the proposed Transfer to (ii) the LTM EBITDA as of the date of the Transfer, is higher than 6.5:1.

12.6 Standstill

- 12.6.1 Neither Promoter Group shall be permitted to Transfer any Equity Shares to any Person during the Call Option 1 Period except pursuant to Article 4.7 (Rights Recapitalisation Call Option prior to the Equal Shareholding Date), 12.3.1 (Call Option 1), 13.2.1 (Transfer to Affiliates), 13.2.2A, 13.6A (Transfer of Vodafone Surplus Shareholding), 13.6B (Transfer of ICL Surplus Shareholding), 13A (Vodafone Group Permitted Security), 13B (ICL Group Permitted Security), 13C (Permitted Sales) or 16 (Change in Control).
- 12.6.2 The Vodafone Group Shareholders shall not be permitted to Transfer any Call Option 2 Shares during the Call Option 2 Period except pursuant to (i) Article 12.3.2 (*Call Option 2*); (ii) 13.2.1 (*Transfer to Affiliates*), subject to Articles 9.4 and 12.3.6 and any subsequent Transfer pursuant to Call Option 2 being no less favourable to the Call Option 2 Purchaser than if made by the transferring Vodafone Group Shareholder; (iii) 13.2.2A; (iv) 13 A (Vodafone Group Permitted Security); (v) 13B (ICL Group Permitted Security); (vi) 13C (Permitted Sales); or (vii) Article 16.3.2.
- 12.6.3 Following the expiration of the Call Option 2 Period, any Promoter Group which holds lesser Equity Shares ("Non-Equal Shareholder") than the other Promoter Group ("Higher Shareholder") shall have the right to acquire such number of Equity Shares as would entitle the Non-Equal Shareholder to equalise its Shareholding with the Higher Shareholder in the manner set out in this Article 12.6.3:
 - (a) The Non-Equal Shareholder shall first make a written offer to the Higher Shareholder to purchase Equity Shares from the Higher Shareholder at a specified price ("Equal Offer Notice").
 - (b) If the Higher Shareholder declines, partially accepts or fails to respond to the Equal Offer Notice within ten (10) Business Days of receipt of the Equal Offer Notice ("Equal Offer Period"), then the Non-Equal Shareholder shall have the right to acquire all of or, if the Higher Shareholder partially accepts the Equal Offer Notice, of the remaining, Equity Shares as specified in the Equal Offer Notice from the market at or below the price specified in the Equal Offer Notice within 30 (thirty) days of the expiry of the Equal Offer Period.
 - (c) If the Higher Shareholder accepts the Equal Offer Notice (in writing), in full or part, within the Equal Offer Period, the Non-Equal Shareholder and the Higher Shareholder shall complete the Transfer of Equity Shares agreed to be Transferred within 10 (ten) days of the expiration of the Equal Offer Period (such 10-day period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the expiration of the Equal Offer Period).
 - (d) For giving effect to the Transfer contemplated in this Article 12.6.3, the Parties shall execute a share purchase agreement in the form set out in the Shareholders' Agreement and all such documents, take all such actions and shall render all such assistance to each other as may be reasonably required to complete the Transfer. Such share purchase agreement, the Shareholders' Agreement and such other terms as may be mutually agreed shall be the sole

terms which govern the sale and purchase of the Equity Shares contemplated in this Article 12.6.3.

12.7 **Cap**

12.7.1 Subject to Article 12.7.3, if a disposal of any Equity Shares by the Vodafone Group Shareholders pursuant to the Capped Options (an "Option Transfer") would, immediately following that Option Transfer, result in the total consideration received by the Vodafone Group Shareholders for all Option Transfers exceeding Rs.830 billion, the Vodafone Group Shareholders may (but shall not be obliged to) procure that Vodafone Plc shall, within a reasonable timeframe, seek Vodafone Plc Shareholder Approval for any such Option Transfer.

12.7.2 Subject to Article 12.7.3, if:

- (a) Vodafone Plc does not seek Vodafone Plc Shareholder Approval for such Option Transfer within 60 (sixty) days of exercise of the Capped Option; or
- (b) the ordinary resolution to approve any such Option Transfer is not passed by the shareholders of Vodafone Plc within 90 (ninety) days of the exercise of Capped Option,

the consideration for that Option Transfer shall be limited such that, immediately following that Option Transfer, the total consideration received by the Vodafone Group Shareholders for all Option Transfers shall not exceed Rs.830 billion.

- 12.7.3 The Vodafone Group Shareholders may disapply Article 12.7.1 and 12.7.2 by written notice to the ICL Group Shareholders within seven (7) days of the exercise of a Capped Option, in which case Article 12.7.1 and Article 12.7.2 shall cease to apply in respect of such Capped Option with immediate effect and the consummation of a Transfer of Equity Shares pursuant to the exercise of:
 - (a) Call Option 1 at the Call Option 1 Price, in accordance with Article 12.3; and/or
 - (b) Call Option 2 at the Call Option 2 Price, in accordance with Article 12.3, and/or
 - (c) the Rights Recapitalisation Call Option at the RCO Price, in accordance with Article 4.7,

(as applicable) or under Step Down Option 1, shall not require Vodafone Plc Shareholder Approval.

- 12.8 If either Promoter Group Transfers Equity Shares to a New Qualifying Shareholder, such New Qualifying Shareholder shall be liable to comply with this Article 12 instead of such Promoter Group, unless the Promoter Group continues to hold Equity Shares, in which case;
 - (a) both the New Qualifying Shareholder and such Promoter Group shall be liable to comply with this Article 12 as one block; but
 - (b) where any of the Vodafone Group Shareholders has become a New Qualifying Shareholder pursuant to Article 13.2.3A or has transferred Equity Shares to a New Qualifying Shareholder pursuant to Article 13.2.2A or an enforcement of Permitted Security, the Shareholding of the New Qualifying Shareholders and the Vodafone Group Shareholders shall be treated, for the purposes of this Article 12, as one block (but, for the avoidance of doubt, the rights, obligations,

covenants and undertakings of the Vodafone Group Shareholders, on the one hand, and the New Qualifying Shareholder, on the other hand, shall be several and not joint or joint and several) and the Shareholding of the Vodafone Group Shareholders shall be applied first (i) in determining the Excess Equity Shares and the Equity Shares the subject of the Equal Offer Notice and (ii) for the purposes of executing the Call Option 1, the Call Option 2, the Step Down Option 1 or the Step Down Option 2.

13. TRANSFER OF SHARES

13.1 The Equity Shares held by the Promoter Groups shall only be transferable in the manner provided in the Articles of Association. A member of a Promoter Group may Transfer its Equity Shares in a manner otherwise than in accordance with the Articles of Association with the prior written consent of the other Promoter Group and provided that such Transfer is completed within 30 (thirty) days of receipt of such written consent (such period to be extended pro tanto by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of receipt of the written consent), failing which, fresh written consent will be required. Any Transfer of Equity Shares by a member of a Promoter Group which is not in accordance with the Articles of Association shall be null and void ab initio and the Company shall not recognise or give effect to such Transfer or recognise any votes in respect of such Equity Shares until the Transfer is reversed (if already effected).

13.2 **Permitted Transfers**

13.2.1 Transfers to Affiliates

- (a) A member of a Promoter Group may Transfer all or some of the Equity Shares held by it to an Affiliate, subject to such Affiliate executing a Deed of Adherence and upon giving prior written notice to the other Promoter Group. The Affiliate must be under an obligation, given in favour of the Company and the other Promoter Group, to re-Transfer the Equity Shares to the original transferring Promoter Group member or another Affiliate of the original transferring Promoter Group member immediately, if it ceases to be an Affiliate of that original transferring Promoter Group member.
- (b) Following a Transfer of Equity Shares to an Affiliate: (i) the original transferring Promoter Group member (but not a subsequent transferor in a series of Transfers to Affiliates) shall, at the option of the non-transferring Promoter Group, remain a Party to the Shareholders' Agreement and shall be jointly and severally liable with the transferee and the other members of the relevant Promoter Group under the Shareholders' Agreement as a member of the relevant Promoter Group; and (ii) the transferee shall be included as a member of the relevant Promoter Group for the purposes of the Articles of Association. Without prejudice to Article 9.2.1(d) and other provisions of the Articles of Association, Article 13.2.1(b)(i) shall not apply with respect to an *inter se* Transfer of Equity Shares between members of a Promoter Group who are Parties on the date of the Shareholders' Agreement.

13.2.2 Transfers to Third Parties

Following the expiration of the Call Option 1 Period and subject to Article 12.6.2 in respect of the Vodafone Group Shareholders, any member of a Promoter Group shall be entitled to Transfer its Equity Shares to:

- (a) any Financial Investor, provided that such Transfer will not result in the transferee (together with its Affiliates) owning more than 10% (ten percent) of the Share Capital; and
- (b) subject to Articles 13.3 (*Right of First Refusal*) and 13.4 (*Tag-Along Right*):
 - (i) any Financial Investor which will result in the transferee (together with its Affiliates) owning more than 10% (ten percent) of the Share Capital; and
 - (ii) any Person other than a Financial Investor.
- 13.2.2A Notwithstanding anything to the contrary contained in the Articles of Association but subject to Article 13.5 (Prohibited Transfer), a Vodafone Group Shareholder shall be entitled to Transfer any of its Equity Shares to any Person where it is required to do so as a result of the occurrence of a Trigger Event or an Enforcement Event under and in accordance with the terms of the Dynamo Agreement. The relevant Vodafone Group Shareholder shall notify (in writing) the ICL Group Shareholders of any proposed Transfer of its Equity Shares under this Article 13.2.2A within five (5) Business Days of being notified under the Dynamo Agreement that it is required to undertake a Transfer of its Equity Shares as a result of the occurrence of a Trigger Event or an Enforcement Event. For the avoidance of doubt, neither Article 13.3 nor Article 13.4 shall apply to any Transfer of Equity Shares made in accordance with this Article 13.2.2A.
- 13.2.3 Notwithstanding anything to the contrary contained in the Articles of Association, any transferee of any Equity Shares that is not a member of a Promoter Group shall not execute a Deed of Adherence and shall not be entitled to any rights under Articles 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters) unless such transferee's Shareholding (together with its Affiliates) pursuant to a Transfer under Article 13.2.2(b), Article 13.2.2A or Article 13A will be equal to or more than the Qualifying Threshold (such transferee, a "New Qualifying Shareholder"). A New Qualifying Shareholder shall be required to execute a Deed of Adherence pursuant to which it shall become a party to the Shareholders' Agreement and be entitled to rights of the transferor Promoter Group under Articles 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters), provided that if the New Qualifying Shareholder's Shareholding (together with its Affiliates) will be equal to or more than the Qualifying Threshold and the transferor's Shareholding (together with its Affiliates) is also equal to or more than the Qualifying Threshold, the New Qualifying Shareholder (together with its Affiliates) shall not be entitled to any rights under Article 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) or 10 (Reserved Matters) and the transferor (together with its Affiliates) shall continue to be entitled to rights under Articles 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters). If a subsequent Transfer of Equity Shares results in the transferor's Shareholding (together with its Affiliates) being less than the Qualifying Threshold and the New Qualifying Shareholder's Shareholding (together with its Affiliates) being equal to or more than the Qualifying Threshold, the New Qualifying Shareholder shall, from the time of such Transfer, be entitled to rights under Articles 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters) in substitution for the transferor.
- 13.2.3A Notwithstanding anything to the contrary contained in the Articles of Association, if

any Vodafone Group Shareholder ceases to be a Vodafone Group Shareholder as a result of the enforcement of Permitted Security (an "Enforced Entity"), such Enforced Entity shall not execute a Deed of Adherence and shall not be entitled to any rights under Articles 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters) unless such Enforced Entity's Shareholding (together with its Affiliates) pursuant to such enforcement of Permitted Security is, immediately following such enforcement, equal to or more than the Qualifying Threshold (such Enforced Entity, a "New Qualifying Shareholder"). A New Qualifying Shareholder shall be required to execute a Deed of Adherence pursuant to which it shall become a party to the Shareholders' Agreement and be entitled to rights of the Vodafone Group Shareholders under Articles 5 (Board of Directors the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters), provided that if the New Qualifying Shareholder's Shareholding (together with its Affiliates) will be equal to or more than the Qualifying Threshold and the Shareholding of any remaining Vodafone Group Shareholders (together with their Affiliates), if any, is also equal to or more than the Qualifying Threshold, the New Qualifying Shareholder (together with its Affiliates) shall not be entitled to any rights under Article 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) or 10 (Reserved Matters) and the relevant Vodafone Group Shareholders (together with its Affiliates) shall continue to be entitled to rights under Articles 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters). If a subsequent Transfer of Equity Shares or enforcement of Permitted Security results in the Shareholding of the remaining Vodafone Group Shareholders (together with their Affiliates) being less than the Qualifying Threshold and the New Qualifying Shareholder's Shareholding (together with its Affiliates) being equal to or more than the Qualifying Threshold, the New Qualifying Shareholder shall, from the time of such Transfer, be entitled to rights under Articles 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters) in substitution for the remaining Vodafone Group Shareholders.

13.2.4 If any transferee that is not a member of a Promoter Group is required to make a public announcement of an open offer with respect to the Company under the Takeover Code following acquisition of Equity Shares from either Promoter Group, such transferee's Shareholding that has been acquired from the Public Shareholders pursuant to such open offer will be subject to voting restrictions specified in Article 12.2 and must be sold in the market to the public or to Financial Investor(s) pursuant to Article 13.2.2(a) within six (6) months of acquisition.

13.3 Right of First Refusal

- 13.3.1 Except as provided in Articles 13.2.1 and 13.2.2(a), 13.2.2A, 13A, 13C, 13.6A and 13.6B and subject to Article 12.6.2 in respect of the Vodafone Group Shareholders, in the event any member of a Promoter Group (a "**Transferring Shareholder**") receives a *bona fide* offer from any Person (a "**Proposed Transferee**") to Transfer any Equity Shares, it shall grant to the other Promoter Group a right of first refusal over any such Transfer of Equity Shares ("**First Refusal Right**") in the manner set forth in this Article 13.3.
- 13.3.2 If the Transferring Shareholder proposes to Transfer to the Proposed Transferee any of the Equity Shares which are subject to the First Refusal Right, the Transferring Shareholder shall first offer the Equity Shares to the other Promoter Group by serving a written notice ("Transfer Notice") on such other Promoter Group ("Nontransferring Shareholder") stating: (i) the number of Equity Shares proposed to be Transferred to the Proposed Transferee ("Offered Shares") and the maximum number of Equity Shares that the Proposed Transferee is willing to acquire; (ii) the consideration for the Transfer; (iii) the other terms and conditions of the Transfer, if

any, as may be reasonably necessary for the Non-transferring Shareholder to determine the price and other terms of such offer; and (iv) the identity of the Proposed Transferee and of its Ultimate Parent and beneficial owner(s). The Non-transferring Shareholder may also require the Transferring Shareholder to produce to the Non-transferring Shareholder such further information as may be reasonably required to enable the Non-transferring Shareholder to establish the *bona fides* of the offer of the Proposed Transferee.

- 13.3.3 Within 30 (thirty) days after the receipt of the Transfer Notice by the Non-transferring Shareholder ("Offer Period"), the Non-transferring Shareholder may deliver a written notice to the Transferring Shareholder: (a) requiring the Transferring Shareholder to Transfer all, but not some only, of the Offered Shares at the same price and on other terms no less favourable to the Non-transferring Shareholder than those stated in the Transfer Notice, to the Non-transferring Shareholder or its Affiliate ("Acceptance Notice"); or (b) stating that it declines to exercise its First Refusal Right on the Offered Shares or (c) stating that it is electing to exercise its Tag-Along Right under Article 13.4 ("Tag Exercise Notice"). An Acceptance Notice shall be irrevocable and shall constitute a binding agreement between the Transferring Shareholder and the Non-transferring Shareholder to purchase the Offered Shares. If the Non-transferring Shareholder fails to serve an Acceptance Notice within the Offer Period, it shall be deemed to have declined to exercise its First Refusal Right on the Offered Shares.
- 13.3.4 In the event an Acceptance Notice has been served pursuant to Article 13.3.3, the Transferring Shareholder shall be bound to Transfer the Offered Shares to the Nontransferring Shareholder.

13.3.5 If the Transferring Shareholder:

- (a) has not received an Acceptance Notice under Article 13.3.3 in respect of all of the Offered Shares or, having received the same, has not within 15 (fifteen) days thereafter (such 15-day period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority) received the consideration for the Offered Shares (provided the Transferring Shareholder is not in breach of this Article 13.3); or
- (b) has not received either an Acceptance Notice or a Tag Exercise Notice under Article 13.3.3,

it shall be entitled to Transfer all, but not some only, of the Offered Shares to the Proposed Transferee at the same price and on other terms no more favourable to the Proposed Transferee than those stated in the Transfer Notice, provided that, if such Transfer is not completed within 90 (ninety) days after the expiration of the Offer Period (such 90-day period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of expiration of the Offer Period), the right to Transfer the Offered Shares to the Proposed Transferee shall lapse.

13.4 **Tag-Along Right**

13.4.1 In the event that the Non-transferring Shareholder has not exercised its First Refusal Right upon receipt of the Transfer Notice in accordance with Article 13.3 (*Right of First Refusal*), the Non-transferring Shareholder shall have the *pro rata* right, but not

an obligation ("Tag-Along Right"), to require the Proposed Transferee to purchase from the Non-transferring Shareholder such number of Equity Shares as may be decided by such Non-transferring Shareholder in its sole discretion but not exceeding its *pro rata* entitlement, such that the number of Equity Shares sold by the Transferring Shareholder and the Non-transferring Shareholder, shall be proportionate to the respective *pro rata inter se* Shareholding of the Transferring Shareholder and the Non-transferring Shareholder, at not less than the price and on other terms no less favourable to the Non-transferring Shareholder than those stated in the Transfer Notice. To the extent the Non-transferring Shareholder exercises its Tag-Along Right, the number of Shares that the Transferring Shareholder may Transfer to the Proposed Transferee shall be correspondingly reduced. Notwithstanding anything contained in this Article 13.4.1, in the event that the proposed Transferee will result in a change in Control of the Company, the Tag-Along Right of the Non-transferring Shareholder shall extend to the entire Shareholding of the Non-transferring Shareholder.

- 13.4.2 In the event the Non-transferring Shareholder has served a Tag Exercise Notice within the Offer Period pursuant to Article 13.3.3, the Transfer of any Equity Shares to the Proposed Transferee shall be in the manner set forth in this Article 13.4. If the Non-transferring Shareholder fails to serve a Tag Exercise Notice within the Offer Period, it shall be deemed to have declined to exercise its Tag-Along Right.
- 13.4.3 The Transfer of the Non-transferring Shareholder's Equity Shares pursuant to the exercise of the Tag-Along Right ("**Tagged Shares**") shall be completed within 30 (thirty) days of the receipt of the Tag Exercise Notice (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of the Tag Exercise Notice).
- 13.4.4 The Transferring Shareholder shall not Transfer any of the Offered Shares to the Proposed Transferee unless and until, simultaneously with such Transfer, the Proposed Transferee purchases all the Tagged Shares at not less than the price, and on terms no less favourable, than those stated in the Transfer Notice.
- 13.4.5 For the avoidance of doubt, this Article 13.4 shall only apply when Article 13.3 is also applicable.

13.5 **Prohibited Transfer**

Notwithstanding anything contained in the Articles of Association, no member of a Promoter Group shall directly or indirectly Transfer any Equity Shares to an Indian Competitor.

13.6 Further Acquisitions

Except as provided in Articles 4 (Funding), 12 (Terms of Equalisation), 13 (Transfer of Shares), 13A (Vodafone Group Permitted Security), 13B (ICL Group Permitted Security) and 16 (Change in Control), no member of a Promoter Group or its Affiliates shall acquire any Equity Shares without the prior written consent of the other Promoter Group.

13.6A Transfer of Vodafone Surplus Shareholding

Notwithstanding anything contained in the Articles of Association but subject to Article 13.5, the Vodafone Group Shareholders shall be permitted to sell, transfer or otherwise dispose of the Vodafone Surplus Shareholding, in whole or in part, in one or more transactions, to any

Person at any time and in any manner as they may deem fit, including on a Recognised Stock Exchange.

13.6B Transfer of ICL Surplus Shareholding

Notwithstanding anything contained in the Articles of Association but subject to Article 13.5, the ICL Group Shareholders shall be permitted to sell, transfer or otherwise dispose of the ICL Surplus Shareholding, in whole or in part, in one or more transactions, to any Person at any time and in any manner as they may deem fit, including on a Recognised Stock Exchange.

13.7 For giving effect to (i) the Transfers contemplated in this Article 13 and (ii) the creation of enforcement of Permitted Security in accordance with Article 13A, the Parties shall execute all such documents, take all such actions and shall render all such assistance to each other as may be reasonably required to complete the Transfer.

13A. VODAFONE GROUP PERMITTED SECURITY

- 13A.1 Any:
- 13A.1.1 Vodafone Group Shareholder may create any bona fide Encumbrance on, over or affecting any Equity Shares held by them from time to time and/or any rights attaching to those Equity Shares; and
- 13A.1.2 direct or indirect holding company of a Vodafone Group Shareholder from time to time may create any bona fide Encumbrance on, over or affecting any shares in a subsidiary of such holding company held by such holding company from time to time and/or any rights attaching to those shares or other rights in respect of the relevant subsidiary,

in each case, in favour of any Financier and/ or any third party in order to secure any or all of their obligations or liabilities (or the obligations or liabilities of any of their holding companies) in respect of the Dynamo Agreement and/or the Facility Agreement ("Permitted Security", with such Equity Shares or shares being "Charged Shares" and such rights being "Charged Rights").

- 13A.2 Articles 9.4, 12.6, 13 (other than Articles 13.2.3, 13.2.3A and 13.5) and 16.1 shall not apply to:
- 13A.2.1 any transfer of Charged Shares (whether conditional or unconditional) to any Person or to the assignment of any Charged Rights (whether conditional or unconditional) to any Person, in each case where such transfer is pursuant to the creation and/or enforcement of Permitted Security; or
- 13A.2.2 any transfer of Charged Shares (whether conditional or unconditional) to any Person or to the assignment of any Charged Rights (whether conditional or unconditional) to any Person, in each case by or on behalf of any Financier following the enforcement of Permitted Security
- 13A.3 The Vodafone Group Shareholders shall notify (in writing) the ICL Group Shareholders of any proposed enforcement of Permitted Security within five (5) Business Days of being notified under the Dynamo Agreement of the proposed enforcement of Permitted Security.
- 13A.4 In the event of an enforcement of Permitted Security, the Vodafone Group Shareholders will use reasonable endeavours to facilitate a discussion between (i) the relevant Financier(s) or the relevant third party enforcing the Permitted Security and (ii) the ICL Group Shareholders, the purpose of which will be to discuss any proposals that might be made regarding a potential sale of Equity Shares held by the Vodafone Group Shareholders to the ICL Group Shareholders or a potential sale of Equity Shares held by the ICL Group Shareholders to the relevant Financier(s) or the relevant third

- party enforcing the Permitted Security.
- 13A.5 The Vodafone Group Shareholders agree and acknowledge that none of the provisions of this Article 13A constitutes a disclosure against any of the representations and warranties made by such Persons under the Implementation Agreement, and the inclusion of such provisions in the Shareholders' Agreement have no effect on, and do not operate as a waiver of, any right, power or remedy that any member of the ICL Group or any of their Representatives may have under the Implementation Agreement.

13B. ICL GROUP PERMITTED SECURITY

13B.1 Any:

- 13B.1.1 ICL Group Shareholder may create any bona fide Encumbrance on, over or affecting any Equity Shares held by them from time to time and/or any rights attaching to those Equity Shares; and
- 13B.1.2 direct or indirect holding company of a ICL Group Shareholder from time to time may create any bona fide Encumbrance on, over or affecting any shares in a subsidiary of such holding company held by such holding company from time to time and/or any rights attaching to those shares or other rights in respect of the relevant subsidiary,

in each case, in favour of any ICL Financier and/or any third party in order to secure any or all of their obligations or liabilities (or the obligations or liabilities of any of their holding companies) in respect of an ICL Financing Arrangement ("**Permitted ICL Security**"). Articles 13.6 and 16.1 shall not apply to the creation of any Permitted ICL Security.

- 13B.2 To the extent that any ICL Financing Arrangement requires the grant or creation of security pursuant to Article 13B.1, the Vodafone Group Shareholders will discuss in good faith with the ICL Group Shareholders the required amendments which might be made to the Shareholders' Agreement in order to facilitate the enforcement of such Permitted ICL Security, which amendments must be reasonable and substantially similar to those which have been made to facilitate the enforcement of Permitted Security and the Vodafone Group Shareholders and the Vodafone Confirming Party cannot unreasonably withhold, condition or delay their consent to such amendments.
- 13B.3 The ICL Group Shareholders agree and acknowledge that neither the definition of "ICL Financing Arrangement" nor any of the provisions of this Article 13B constitutes a disclosure against any of the representations and warranties made by such Persons under the Implementation Agreement, and the inclusion of such definition and provisions in the Shareholders' Agreement have no effect on, and do not operate as a waiver of, any right, power or remedy that any member of the Vodafone Group or any of their Representatives may have under the Implementation Agreement.

13C. PERMITTED SALES

13C.1 If, pursuant to the Rights Recapitalisation approved by the Board on 23 January 2019, the ICL Group Shareholders subscribe for Equity Shares for a total subscription amount of more than INR 72,500 million (the total number of Equity Shares subscribed for by the ICL Group Shareholders less the number of Equity Shares which have a total subscription price of INR 72,500 million being the "Excess Subscription Shares"), the ICL Group Shareholders have the right, but not the obligation, to sell and Transfer Excess Subscription Shares, through one Transfer or a series of Transfers, in any manner as they may deem fit, including on a Recognized Stock Exchange, subject to applicable Law.

14. DEADLOCK

14.1 For the purpose of this Article 14, a "Deadlock" shall be deemed to have occurred if:

- 14.1.1 a proposal is made in respect of any Reserved Matter but is not approved in accordance with Article 10 at two (2) consecutive duly convened meetings of the Board (or following the circulation of the relevant Circular Resolution in writing on two (2) separate occasions); or
- 14.1.2 a quorum is not present at two (2) consecutive duly convened meetings of the Board by reason of the absence of the Directors nominated and appointed upon request of the same Promoter Group.
- 14.2 In the event of a Deadlock, either Promoter Group may give written notice to the other and to the Company that it regards a Deadlock as having occurred ("Deadlock Notice") and immediately refer the Deadlock to the chief executive officer of Vodafone Plc and the group chairperson of the ICL Group (or nominees of such representatives) for resolution through mutual discussion (only one Deadlock Notice may be served in respect of any one proposal or series of related proposals).
- 14.3 If the Deadlock is not resolved within 30 (thirty) days of the Deadlock Notice, then *status quo* shall prevail, provided that if the Deadlock relates to a Draft Revised Business Plan, the provisions of Article 11.4 shall apply.
- 14.4 If the Deadlock is resolved within 30 (thirty) days of the Deadlock Notice, then the Promoter Groups shall procure that the Company gives effect to the relevant resolution(s).

15. DEFAULT

15.1 Failure to Comply with Voting

- 15.1.1 If any member of a Promoter Group fails to vote the Equity Shares held by it in accordance with the Articles of Association or votes the Equity Shares held by it contrary to the Articles of Association (such Group, the "Defaulting Promoter Group" and the failure to vote or voting contrary to the Articles of Association, a "Voting Default"), the Defaulting Promoter Group shall be deemed to be in material breach of the Articles of Association pursuant to which Article 15.1.3 shall apply unless:
 - (a) within 30 (thirty) days of the date of the Voting Default, the Promoter Groups or the Board agree to convene another General Meeting or issue a fresh notice for a shareholders' vote pursuant to Article 9.1.4, following which the Defaulting Promoter Group votes the Equity Shares held by it or refrains from voting the Equity Shares held by it in accordance with the Articles of Association and the relevant resolution is approved or rejected in accordance with the Articles of Association; or
 - (b) if the Defaulting Promoter Group demonstrates, within 30 (thirty) days of the date of the Voting Default, to the reasonable satisfaction of the non-defaulting Promoter Group, that the Voting Default was on account of an administrative error.
- 15.1.2 In the event the Defaulting Promoter Group is unable to demonstrate to the reasonable satisfaction of the other Promoter Group pursuant to Article 15.1.1(b) that the Voting Default was on account of an administrative error, either Promoter Group shall have the right to refer the matter to the chief executive officer of Vodafone Plc and the group chairperson of the ICL Group for their consideration and decision within 15 (fifteen) days of the expiration of the 30-day period referred to in Article 15.1.1(b). The chief

executive officer of Vodafone Plc and the group chairperson of the ICL Group shall decide the matter within 30 (thirty) days of the date of referral and such decision shall be final and binding on the Promoter Groups. In the event the chief executive officer of Vodafone Plc and the group chairperson of the ICL Group are unable to agree to a decision, either Promoter Group may refer the matter to an expedited arbitration procedure under the provisions of the Shareholders' Agreement to be completed within six (6) months of the date of referral.

- 15.1.3 If the Voting Default is not cured or resolved pursuant to Article 15.1.1(a) or 15.1.1(b) or 15.1.2:
 - (a) the rights of the Defaulting Promoter Group under Articles 5 (*Board of Directors of the Company*), 6 (*Shareholders Meetings*), 7 (*Key Employees*) and 10 (*Reserved Matters*) shall cease;
 - (b) the obligations of the Defaulting Promoter Group under the Articles of Association shall cease only if the Defaulting Promoter Group no longer holds any Equity Shares; and
 - (c) if the Defaulting Promoter Group comprises ICL Group Shareholders, Article 12.2 shall cease to apply.

15.2 Event of Default

- 15.2.1 An event of default ("Event of Default") shall occur or be deemed to have occurred in relation to a Promoter Group ("Defaulting Shareholder Group") if:
 - (a) a member of the Defaulting Shareholder Group commits a material breach of Article 4 (Funding), 5 (Board of Directors of the Company), 10 (Reserved Matters), 12 (Terms of Equalisation), 13 (Transfer of Shares) or 16 (Change in Control) and such breach is not cured within 30 (thirty) days of written notice by the non-defaulting Promoter Group, provided that any such breach that arises from non-receipt of any approval of a Governmental Authority in respect of Article 4 (Funding), 12 (Terms of Equalisation), 13 (Transfer of Shares) or 16 (Change in Control) for reasons beyond the control of the relevant Party shall not be regarded as a material breach; or
 - (b) any member of the Defaulting Shareholder Group has:
 - (i) an official manager, receiver, trustee, voluntary administrator, liquidator or provisional liquidator appointed for all or substantially all of its assets or undertaking and such appointment is not dismissed, reversed, vacated or stayed within 90 (ninety) days of such appointment; or
 - (ii) entered into or resolved to enter into winding up proceedings or an arrangement, composition or compromise with or assignment for the benefit of its creditors generally or any class of creditors, or proceedings are commenced by such Shareholder to sanction such an arrangement, composition or compromise, in each case, other than for the purposes of (A) a *bona fide* scheme of restructuring, reconstruction or amalgamation, or (B) a voluntary liquidation of entities that no longer hold Equity Shares and do not have substantial assets.

For the avoidance of doubt, the enforcement of Permitted Security in accordance with Article 13A shall not constitute an Event of Default.

- 15.2.2 The Defaulting Shareholder Group shall be entitled to demonstrate, within 30 (thirty) days of the date of notification of the Event of Default by the non-defaulting Promoter Group, to the reasonable satisfaction of the non-defaulting Promoter Group, that such Event of Default occurred on account of an administrative error.
- 15.2.3 In the event the Defaulting Shareholder Group is unable to demonstrate to the reasonable satisfaction of the non-defaulting Promoter Group pursuant to Article 15.2.2 that the Event of Default was on account of an administrative error, either Promoter Group shall have the right to refer the matter to the chief executive officer of Vodafone Plc and the group chairperson of the ICL Group for their consideration and decision within 30 (thirty) days of the expiration of the 30-day period referred to in Article 15.2.2. The chief executive officer of Vodafone Plc and the group chairperson of the ICL Group shall decide the matter within 30 (thirty) days of the date of referral and such decision shall be final and binding on the Promoter Groups. In the event the chief executive officer of Vodafone Plc and the group chairperson of the ICL Group are unable to agree to a decision, either Promoter Group may refer the matter to an expedited arbitration procedure under the provisions of the Shareholders' Agreement to be completed within six (6) months of the date of referral.
- 15.2.4 If an Event of Default is not cured or resolved pursuant to Article 15.2.1(a), 15.2.2 or 15.2.3:
 - (a) the rights of the Defaulting Shareholder Group under Articles 5 (*Board of Directors of the Company*), 6 (*Shareholders Meetings*), 7 (*Key Employees*) and 10 (*Reserved Matters*) shall cease;
 - (b) the obligations of the Defaulting Shareholder Group under the Articles of Association shall cease only if the Defaulting Shareholder Group no longer holds any Equity Shares; and
 - (c) if the Defaulting Shareholder Group comprises ICL Group Shareholders, Article 12.2 shall cease to apply.
- 15.3 Nothing in Article 15.1 or 15.2 shall affect the right of the non-defaulting Promoter Group to claim any losses, damages, costs and expenses, including legal fees and expenses, to the extent arising or resulting from a Voting Default or an Event of Default, regardless of whether such default has been cured.
- 15.4 Notwithstanding anything contained in the Articles of Association, if any member of either Promoter Group is unable to comply with any obligation under the Articles of Association pursuant to an order of a Governmental Authority issued in respect of such member, the rights of the relevant Promoter Group under the Articles of Association shall not cease provided that such Promoter Group uses all reasonable endeavours to procure that such order is vacated.

16. CHANGE IN CONTROL

16.1 Subject to Articles 13A, 13B and 16.8, each Promoter Group shall notify (in writing) the other Promoter Group of any proposed change in Control of any of its members (a "CoC Shareholder"), and the Vodafone Group Shareholders shall immediately notify (in writing) the ICL Group Shareholders of any proposed Vodafone Restricted Group Sale Disposal, in each case, within five (5) Business Days of the public announcement of such proposed transaction or the execution of binding documentation in respect of such proposed transaction, whichever is earlier, and such notice shall specify the manner in which such transaction will occur (the "CoC Notice").

- 16.2 Following issue of a CoC Notice in respect of a listed CoC Shareholder that is an ICL Group Shareholder:
 - 16.2.1 the other ICL Group Shareholders (the "Remaining ICL Shareholders") shall be entitled, within 30 (thirty) days of the CoC Notice (the "ICL CoC Period"), to require such CoC Shareholder to Transfer all of its Equity Shares ("CoC Shares") to them or their Affiliate(s) at the ICL CoC Price pursuant to a written notice ("CoC Exercise Notice") issued to the CoC Shareholder (with a copy to the Company and the Vodafone Group Shareholders), and the CoC Shares shall then be promptly Transferred to the Remaining ICL Shareholders or their Affiliate(s). The Transfer of the CoC Shares pursuant to this Article 16.2.1 shall be completed within 45 (forty five) days of the CoC Notice (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of the CoC Exercise Notice); and
 - 16.2.2 if the Remaining ICL Shareholders fail to issue a CoC Exercise Notice pursuant to Article 16.2.1 within the ICL CoC Period, the Vodafone Group Shareholders shall be entitled, within 30 (thirty) days of the expiration of the ICL CoC Period to require the CoC Shareholder to Transfer all of the CoC Shares to the Vodafone Group Shareholders or their Affiliate(s) at the Vodafone Purchase Price pursuant to a CoC Exercise Notice issued to the CoC Shareholder (with a copy to the Company and the ICL Group Shareholders), and the CoC Shares shall then be promptly Transferred to the Vodafone Group Shareholders or their Affiliate(s). The Transfer of the CoC Shares pursuant to this Article 16.2.2 shall be completed within 75 (seventy five) days of the CoC Notice (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of the CoC Exercise Notice).
- 16.3 Following issue of a CoC Notice in respect of an unlisted CoC Shareholder that is an ICL Group Shareholder, the Vodafone Group Shareholders shall be entitled to take either of the following actions within 30 (thirty) days of the CoC Notice:
 - 16.3.1 pursuant to a CoC Exercise Notice, require such CoC Shareholder to Transfer all of the CoC Shares to the Vodafone Group Shareholders or their Affiliate(s) at the Vodafone Purchase Price, and the CoC Shares shall then be promptly Transferred to the Vodafone Group Shareholders or their Affiliate(s). The Transfer of the CoC Shares pursuant to this Article 16.3.1 shall be completed within 45 (forty five) days of the CoC Notice (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of the CoC Exercise Notice); or
 - 16.3.2 pursuant to a CoC Exercise Notice, require the ICL Group Shareholders to purchase from the Vodafone Group Shareholders such number of Equity Shares as may be decided by the Vodafone Group Shareholders in their sole discretion but not exceeding their *pro rata* entitlement, such that the number of Equity Shares sold by the Vodafone Group Shareholders represents the same proportion of the Share Capital as the number of Equity Shares held by the CoC Shareholder, at the Vodafone Sale Price. The Transfer of Equity Shares pursuant to this Article 16.3.2 shall be completed within 45 (forty five) days of the CoC Notice issued by the Vodafone Group Shareholders (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental

Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of the CoC Exercise Notice).

- 16.4 Following issue of a CoC Notice in respect of a CoC Shareholder that is a Vodafone Group Shareholder or in case of a Vodafone Restricted Group Sale Disposal, the ICL Group Shareholders shall be entitled to take either of the following actions within 30 (thirty) days of the CoC Notice:
 - 16.4.1 pursuant to a CoC Exercise Notice, require such CoC Shareholder to Transfer all of the CoC Shares to the ICL Group Shareholders or their Affiliate(s) at the Vodafone Purchase Price, and the CoC Shares shall then be promptly Transferred to the ICL Group Shareholders or their Affiliate(s). The Transfer of the CoC Shares pursuant to this Article 16.4.1 shall be completed within 45 (forty five) days of the CoC Notice (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of the CoC Exercise Notice); or
 - 16.4.2 pursuant to a CoC Exercise Notice, require the Vodafone Group Shareholders to purchase from the ICL Group Shareholders such number of Equity Shares as may be decided by the ICL Group Shareholders in their sole discretion but not exceeding their *pro rata* entitlement, such that the number of Equity Shares sold by the ICL Group Shareholders represents the same proportion of the Share Capital as the number of Equity Shares held by the CoC Shareholder, at the Vodafone Sale Price. The Transfer of Equity Shares pursuant to this Article 16.4.2 shall be completed within 45 (forty five) days of the CoC Notice issued by the ICL Group Shareholders (such period to be extended *pro tanto* by the period required to obtain any necessary regulatory approval from or make any necessary filing with any Governmental Authority, provided that such extended period shall be no longer than 12 (twelve) months from the date of the CoC Exercise Notice).
- 16.5 Following any change in Control of a CoC Shareholder, the Shareholding of such CoC Shareholder shall not be counted towards the Shareholding of the Promoter Group to which it belonged (unless Article 16.2.1 or 16.6 is applicable).
- 16.6 Nothing in the Articles of Association shall prevent, constitute a breach, require the sale or Transfer of any Equity Shares or otherwise restrict in any manner, with respect to the Vodafone Group:
 - 16.6.1 any change in control of Vodafone Plc following (i) a successful public offer for shares (which, for the avoidance of doubt, may be implemented by a scheme of arrangement) or (ii) a transaction which involves the issue of shares in Vodafone Plc to one or more persons which would require a general offer for the shares of Vodafone Plc but for the requirement to make such offer having been waived, in each case in accordance with the UK City Code on Takeovers and Mergers;
 - 16.6.2 a Vodafone Direct Spin-off Disposal; or
 - 16.6.3 a Vodafone Permitted Group Sale Disposal;

(collectively, the "Vodafone Permitted Transactions") and, in each case, any Transfers pursuant to or in connection with any Vodafone Permitted Transaction. The Vodafone Group Shareholders shall notify (in writing) the ICL Group Shareholders of any proposed Vodafone Permitted Transaction within five (5) Business Days of the public announcement of such transaction or the execution of binding documentation in respect of such transaction, whichever

is earlier, and such notice shall specify the manner in which such transaction will occur. For the avoidance of doubt, none of the Vodafone Permitted Transactions shall be considered a breach of the Articles of Association or an Event of Default.

- 16.7 If any transferee that is not a member of a Promoter Group is required to make a public announcement of an open offer with respect to the Company under the Takeover Code following a change in Control of a member of either Promoter Group that is subject to the provisions of this Article 16, such transferee's Shareholding that has been acquired from the Public Shareholders pursuant to such open offer will be subject to voting restrictions specified in Article 12.2 and must be sold in the market to the public or to a Financial Investor pursuant to Article 13.2.2(a) within six (6) months of acquisition.
- 16.8 Article 16.1 shall not apply to any actual or proposed change in Control, direct or indirect, of a Vodafone Group Shareholder which results from the occurrence of a Trigger Event or an Enforcement Event under the Dynamo Agreement and is effected in accordance with the terms of the Dynamo Agreement. The relevant Vodafone Group Shareholder shall notify (in writing) the ICL Group Shareholders of any change in Control under this Article 16.8 within five (5) Business Days of such change in Control becoming effective.
- 16.9 Nothing in these Articles of Association shall prevent, constitute a breach, require the sale or Transfer of any Equity Shares or otherwise restrict in any manner, with respect to the Vodafone Group, entry by Vodafone Plc into the Dynamo Agreement.

17. TERMINATION; FALL AWAY OF RIGHTS

- 17.1 The Shareholders' Agreement shall automatically terminate:
 - 17.1.1 in respect of the rights and obligations of any Promoter Group, upon that Promoter Group ceasing to hold any Equity Shares, it being understood that the Shareholders' Agreement shall remain in force between the non-exiting Promoter Group and any New Qualifying Shareholder and, subject to the provisions of the Shareholders' Agreement, the non-exiting Promoter Group and the exiting Promoter Group shall not have any rights or obligations with respect to each other; and
 - 17.1.2 in respect of the rights and obligations of the Vodafone Group Shareholders in the event of a Vodafone Direct Spin-off Disposal.
- 17.2 The Shareholders' Agreement may be terminated by each Promoter Group with immediate effect upon the earlier of the following, the occurrence of which shall be promptly notified by the Company to each Promoter Group:
 - 17.2.1 any execution or other process of any Governmental Authority issued against or levied upon all or substantially all of the Company's assets, which is not discharged or withdrawn or stayed or vacated within 90 (ninety) days of the date of issue;
 - 17.2.2 an official manager, receiver, trustee, voluntary administrator, liquidator or provisional liquidator is appointed for all or substantially all of the Company's assets or undertaking and such appointment is not dismissed, reversed, vacated or stayed within 90 (ninety) days of such appointment; or
 - 17.2.3 the Company has entered into or resolved to enter into winding up proceedings, or an arrangement, composition or compromise with or assignment for the benefit of its creditors generally or any class of creditors, or proceedings are commenced by a Shareholder to sanction such an arrangement, composition or compromise, in each case, other than for the purposes of a bona fide scheme of restructuring, reconstruction or amalgamation.

- \$17.3 Notwithstanding anything contained in the Articles of Association:
 - 17.3.1 If the Shareholding of a Promoter Group falls below the Qualifying Threshold, the Parties to the Shareholders' Agreement shall engage in discussions in good faith to determine the consequences of such breach of the Qualifying Threshold, provided, however that, if no agreement is reached among such Parties or the breach has not been cured within a period of 90 days from the date on which the Qualifying Threshold of a Promoter Group is breached, the rights of such Promoter Group (the "Non-Qualifying Promoter Group") under Articles 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) and 10 (Reserved Matters) shall irrevocably cease and the following shall apply:
 - (a) the other Promoter Group (the "Qualifying Promoter Group") shall be entitled, by notice to the other Parties, to elect (in its sole discretion) to terminate the Shareholders' Agreement and this Part II of the Articles of Association, with immediate effect; and
 - (b) notwithstanding anything contained in the Articles of Association, including Article 3.2: (i) the Non- Qualifying Promoter Group shall be entitled to take necessary actions to declassify the persons / entities belonging to such Non-Qualifying Promoter Group, as promoter / promoter group of the Company under the applicable laws and regulations; and (ii) the Qualifying Promoter Group shall not have any obligation to facilitate any action described in (i) above in any manner whatsoever.
 - 17.3.2 Upon exercise by the Qualifying Promoter Group of its right under Article 17.3.1(a), the Shareholders' Agreement and this Part II of the Articles of Association, shall terminate and fall away with immediate effect.
 - 17.3.3 If the Qualifying Promoter Group does not exercise its right to terminate the Shareholders' Agreement and this Part II of the Articles of Association, under Article 17.3.1(a):
 - (a) the obligations of the Non-Qualifying Promoter Group under the Shareholders Agreement shall cease only if such Promoter Group no longer holds any Equity Shares; and
 - (b) if the Shareholding of the ICL Group Shareholders has fallen below the Qualifying Threshold, Article 12.2 shall cease to apply.
- If the rights of the ICL Group Shareholders under the Articles of Association fall away pursuant to Article 17.3.1 as a result of non-participation or partial participation in a Rights Recapitalisation under Article 4, such rights shall be restored if the ICL Group Shareholders increase their Shareholding to at least the Qualifying Threshold in accordance with Article 4.7 (Rights Recapitalisation Call Option prior to the Equal Shareholding Date) within the time period specified therein (the "Rights Cure Period"), in which case:
 - (a) the ICL Group Shareholders shall not be entitled to any rights under Article 5 (Board of Directors of the Company), 6 (Shareholders Meetings), 7 (Key Employees) or 10 (Reserved Matters) during the Rights Cure Period;

^{\$} Article 17.3 has been altered pursuant to Special Resolution passed by the members of the Company at the Extra-ordinary General Meeting held on 26th March 2022

^{\$\$} Article 17.4 has been inserted pursuant to Special Resolution passed by the members of the Company at the Extra-ordinary General Meeting held on 26th March 2022

- (b) before, during and after the Rights Cure Period, the obligations of the ICL Group Shareholders under the Shareholders Agreement shall cease only if no ICL Group Shareholder holds any Equity Shares; and
- (c) Article 12.2 shall not be applicable during the Rights Cure Period and, if the Shareholding of the ICL Group Shareholders is not increased to the Qualifying Threshold within the Rights Cure Period, Article 12.2 shall cease to apply thereafter.

18. JOINT AND SEVERAL LIABILITY

- 18.1 Notwithstanding any provisions to the contrary in the Articles of Association, all ICL Group Shareholders shall be treated as a single shareholder for the purpose of the Articles of Association and their rights, obligations, covenants and undertakings hereunder shall be joint and several, and a breach by any one ICL Group Shareholder of its rights, obligations, covenants or undertakings hereunder shall be deemed as a collective breach by the other ICL Group Shareholders of their respective rights, obligations, covenants and undertakings hereunder.
- 18.2 Notwithstanding any provisions to the contrary in the Articles of Association, all Vodafone Group Shareholders shall be treated as a single shareholder for the purpose of the Articles of Association and their rights, obligations, covenants and undertakings hereunder shall be joint and several, and a breach by any one Vodafone Group Shareholder of its rights, obligations, covenants or undertakings hereunder shall be deemed as a collective breach by the other Vodafone Group Shareholders of their respective rights, obligations, covenants and undertakings hereunder.

19. CONSENTS; NOTICES

References to consents or notices by the Parties may be satisfied by GIL on behalf of the ICL Group Shareholders or Euro Pacific Securities Ltd. on behalf of the Vodafone Group Shareholders.

20. ANTI-CORRUPTION LAWS

The Parties shall not, and shall procure that the members of their respective Groups shall not, directly or indirectly through their Representatives or any Person authorised to act on their behalf (a) offer, promise, pay, authorise or give money or anything of value to any Person for the purposes of (i) influencing any act or decision of any governmental official, (ii) inducing any government official to do or omit to do an act in violation of a lawful duty, (iii) securing any improper advantage or (iv) inducing any government official to influence the act or decision of a Governmental Authority or (b) engage in any other activity, practice or conduct which would give rise to an offence under, or non-compliance with, any applicable anti-bribery and anti-corruption Laws.

21. FURTHER ASSURANCES

Each Party shall, upon being required to do so by any other Party, execute such documents and perform such acts and things as such other Party may reasonably consider necessary for giving effect to the provisions of the Articles of Association.

SCHEDULE 1 INVESTMENT BANKS

- Morgan Stanley
 UBS
- 3. Goldman Sachs
- 4. Deutsche Bank
- 5. Bank of America Merrill Lynch
- 6. J.P. Morgan7. HSBC
- 8. Credit Suisse
- 9. Citibank

We, the several persons whose names, address and occupation are subscribed hereunder are desirous of being formed into a Company in pursuance of these Articles of Association and we respectively agree to take the number of shares in the capital of the Company set opposite our respective names:

Name, Address, Description and occupation of each Subscriber	Number of Equity Shares taken by each Subscriber	Signature of Subscriber	Signature of Witness and his name, address, description and occupation
Shri Mahesh Chandra Bagrodia S/o Shri K D Bagrodia B-2 15th Floor, Prithvi Apartments, Altamount Road, Bombay 400 001	10 (Ten)	Sd/- M C Bagrodia	^
Service			
Shri Raghuvir Bhandari S/o Dr Shree Krishna Raj Bhandari 1 Tahiti Co-op Housing Society 23 Juhu Versova Link Road Bombay 400 053	10 (Ten)	Sd/- R Bhandari	
Service			
Shri Sushil Kumar Saboo S/o Madan Lal Saboo Satnam Apartments 4th Floor Colaba	10 (Ten)	Sd/- S K Saboo	
Bombay 400 005 Service			Witness to All Sd/-
Shri Gopi Krishna Tulsian	10	Sd/-	Shri Narayan Rathi
S/o Laxmi Narayan Tulsian 301 Pradeep Worli Hill Estate Bombay 400 018	(Ten)	G K Tulsian	S/o Shri Tulsidas Rathi R1/15 Goverdhangiri Society Bangur Nagar Goregaon (W)
Service			Bombay 400 090
Shri Raghuram Raju S/o Shri K V Rama Raju E-GE DDA Flats Mayapuri New Delhi 110 064	10 (Ten)	Sd/- R Raju	Company Secretary
New Deini 110 064			
Advocate			
Shri Deepak Adalkha S/o Shri L D Adalkha E-25 A East of Kailash New Delhi 110 065	10 (Ten)	Sd/- D Adlakha	
Advocate			
Ms Jyoti Pande	10	Sd/-	
D/o Shri R C Pande	(Ten)	J Pande	
104 Friends Colony			
New Delhi 110 065			
Advocate			
Total	70		
	(Seventy)		

Place: Bombay Date: March 01, 1995